# **Leiden University**

# **Master Thesis**

#### 2017/08/08

#### **Aleksandar Taskov**

"An exploration of Russia's military approach during the Ukrainian crisis (2013-2015)"



**Aleksandar Taskov** 

Student number: 1896857

Thesis supervisor Stef Wittendorp

Word count: 33373

MSc. Crisis and Security Management Faculty of Governance and Global Affairs

## **Abstract**

In the end of 2013, and the beginning of 2014, Russia launched an unprecedented, unconventional military campaign in Ukraine. In less than a month Russia annexed the Crimean Peninsula and sparked a bloody civil war in the Eastern regions of Ukraine. The new Russian military approach shocked the whole World, yet there was no term that can describe it. The media soon filled this gap with the "label" "hybrid warfare", an almost forgotten concept developed in the early 2000s in the United States professional military literature. This appellation quickly gained popularity and started to appear not only in news articles but also in official governmental and intergovernmental documents. In academia, however, a debate that is still continuing arose. On one side, there are scholars who argue that the new Russian military approach can be examined as an instance of hybrid warfare but on the other, many disagree. The main problems are that there is no generally accepted definition of hybrid warfare, the literature is fragmented and, often researchers use the term simply because is "catchy". This master thesis engaged in the above-mentioned discussion. First an appropriate theoretical framework which was developed before the debate had started was selected, then it was applied to the Ukrainian case, with the prime goal to test if the Russian actions in the Ukrainian crisis (2013-2015) can be conceptualized as an exemplification of a hybrid war. In doing so a knowledge gap was filled, because prior to this work, there was no extensive exploration of the Ukrainian crisis through the lens of a hybrid war theory. The analysis proved that Russian military approach, exposed during the Ukrainian crisis, can be conceptualized as an instance of a hybrid war, yet it is not certain if the selected theory is the most appropriate analytical tool.

Key words: hybrid warfare, Ukrainian crisis, Russia, Ukraine, military approach, physical domains of operations, non-physical domain of operations.

**Foreword** 

A number of factors contributed to the decision to write my master thesis on Russia and

particularly its involvement in the Ukrainian crisis. First and foremost, I speak Russian and I was

capable to follow the crisis while it was developing. There was a significant discrepancy in the way

the Russian and Western media represented the events. This ambiguity has ever since puzzled me

and I always wanted to explore in detail what actually happened in Ukraine. Second, I am a

Bulgarian and my country lays in an important "geopolitical crossroad", where a number of foreign

interest meet. The expansionist Russian actions have strong implications not only for my country's

security but also, for Europe's in general.

I hope that my thesis makes a small step towards the attainment of a better understanding

of the new Russian military approach. The analysis is organized in a way where information is

derived from a variety of different sources. In doing so this work may be used as a foundation for

future research not only on the Ukrainian crisis but also on the new Russian warfare.

Herewith, I would like to thank my supervisor Stef Wittendorp for the demonstrated respect,

help and support during the conduction of this research. I also would like to thank Mr. Sergei

Boeke, for his advice and helpful contributions.

Aleksandar Taskov

Sofia, August 8, 2017

4

# **Table of Contents**

List of figures	7
List of acronyms and abbreviations	8
1.Introduction	9
1.1. Russia's military approach in the 21st century	9
1.2. Research question	11
1.3 Sub research questions	11
1.4. Societal and Academic relevance	12
1.5. Initial remarks on hybrid warfare	14
1.6. Structure of this thesis	16
2. Theoretical framework	18
2.1. Evolution of the theories of warfare - hybrid conflict origins	18
2.2. McCulloh's theory	20
2.3. Theoretical framework – conclusions	24
3. Research Design	25
3.1. Research Design	25
3.2. Methodology	28
3.3. Theory-testing Process Tracing	29
3.4. Limitations	33
3.5. Research Design – conclusions	34
4. Expressions of hybrid warfare in the physical domains of operations during the Ukrainian crisis	35
4.1. Ukrainian crisis - overview	35
4.2. Exploitation of the unique conflict context	39
4.3. Capability overmatch	45
4.4. Combination of conventional and unconventional elements	50
4.5. Defensive type operations	52
4.6. Attritional tactics	55
4.7 Conclusions	58
5. Expressions of hybrid warfare in the non-physical domain of operations during the Ukrainian crisis	59
5.1. Exploitation of the unique conflict context	59
5.2. Ideology of the hybrid actor	63
5.3. Existential threat to the hybrid actor	67
5.4. Attritional tactics	70
5.5. Conclusions	73
6. General conclusions	74

	6.1. Answers to sub research questions	. 74
	6.1. Answers to central research question	. 81
	6.3. Limitations of McCulloh's theoretical framework	. 82
	6.4. Limitations of the thesis and avenues for further research	. 82
В	ibliography	. 84

# List of figures

Figure 1: Single-case Design	27
Table 1 Ukrainian conflict main battle tanks	48
Table 2 Ukrainian conflict self-propelled artillery	48-49

# List of acronyms and abbreviations

BBC British Broadcasting Corporation

CIA Central Intelligence Agency

EU European Union

GDP Gross domestic product

HCNM High Commissioner on National Minorities

MVnR Bulgarian Ministry of Foreign Affairs

MZS Ukrainian Ministry of Foreign Affairs

NATO North Atlantic Treaty Organization

NGO Non-governmental Organization

OHCHR Office of the United Nations High Commissioner for Human Rights

OSCE Organization for Security and Co-operation in Europe

OUN Organization of Ukrainian Nationalists

PT Process Tracing

RT Russia Today

TASS Russian News Agency

UN United Nations

UNIAN Ukrainian Independent Information Agency of News

US(A) United States (of America)

USSR Union of Soviet Socialist Republics

WW2 World War Two

# 1.Introduction

## 1.1. Russia's military approach in the 21st century

"[The] collapse of the Soviet Union was the largest geopolitical catastrophe of the [20<sup>th</sup>] century" Vladimir Putin, 2005 (Putin, 2005)

Since the beginning of the 21<sup>st</sup> century, the Russian Federation is being increasingly proactive on the international arena (Melkonyan, 2017). Under the presidency of Vladimir Putin, the country's economic and demographic indicators have been gradually improving (Bank, 2015). However, during the last four years, Russia's relations with the West have deteriorated to a point where authors and policy makers have expressed opinions that the world is entering into a second Cold War (Johnston, 2014), (Monaghan, 2015). One of the key catalysts for these developments was the Russian involvement in the Ukrainian crisis (2013-present), particularly the annexation of Crimea in March 2014 and the support to the separatist in the eastern regions of the country (Ray, Ukraine crisis, 2016). The crisis served as a showcase of the modern Russian military approach but, there was a lack of a term that described the new way Russia conducted its operations (Bērziņš, 2015, p. 42). This gap was soon filled in the media, by the appellation "hybrid war", a concept developed in the early 2000s in the United States professional military literature (Kendall, 2014), (Boffey, 2016), (Nemeth, 2002). Quickly the term gained popularity.

Governmental officials started using it. In November 2015, the Bulgarian President, Rosen Plevneliev, asserted that Russia plans a hybrid war on the Balkans, aiming to destabilize Europe (Independent, 2015). It also began to appear in official documents. For example, the United Kingdom's 2015 National Security Strategy, examines the new Russian "hybrid tactics" as a security threat (Government, 2015, p. 18). Even intergovernmental organizations acknowledged the term. The European Union (EU) and the North Atlantic Treaty Organization (NATO) are developing a "Joint Framework on countering hybrid threats a European Union response" (Commission, 2016). In a recent European Parliament briefing paper which examines the achievements of EU-NATO cooperation, it is noted: "The concept of hybrid threat has gained

traction in relation to Russia's actions in Ukraine" (Pawlak, 2017). It can be seen that in the public sphere hybrid wars are becoming officially accepted.

Yet, in academia there is an ongoing debate regarding the appropriateness and novelty of the hybrid war theory, to describe the new Russian military approach. There are two dominant lines of discussion. The first one examines the uniqueness of hybrid wars as a new form of conflict. Bettina Renz argues that hybrid wars are nothing new because "indirect approaches and unconventional tactics" have been part of the toolkit "of most states for many years" (Renz, 2016, p. 284). She continues by stating that the term, "does not adequately reflect the direction of ongoing Russian military modernisation" and that the effectiveness of the Russian operation in Crimea was a result not of the application of a "new war-winning formula" but of "extremely favorable circumstances" (Renz, 2016, p. 284). Oscar Jonsson and Robert Seely, express a similar opinion. They argue that events in Crimea "reflected a failure of imagination, rather than novel Russian military capabilities" (Seely, 2015, p. 4). They describe the Russian military approach as "old wine in new bottles" (Seely, 2015, p. 4). Nicu Popescu follows the same line of thought. He argues that hybrid tactics are not only not new but also, they are not exclusively Russian, "they are as old as war itself, and Western states have often used elements of it quite effectively" (Popescu, 2015).

The second dominant line of discussion examines the problems related to the usage of hybrid wars as an analytical tool. There are a number of works which study the Ukrainian crisis from the perspective of hybrid wars (Korybko, 2015), (Stoyanov, 2016) (Vlasiuk, 2015). However, the fact that there is no generally accepted definition and theoretical framework of hybrid warfare creates ambiguity. Many authors have used the concept in their studies, but have focused on specific aspects of the crisis. Vitalii Vlasiuk and Outi Korhonen have observed only the legal challenges that the hybrid war in Ukraine has created, Maria Snegovaya has studied only the information warfare as part of the hybrid war, Liisa Past has examined only cyber warfare as part of the hybrid conflict, Maciej Bartkowski has investigated only non-violent civil defense as a means to counter the Russian hybrid warfare. (Vlasiuk, 2015), (Korhonen, 2015), (Snegovaya, 2015), (Geers, 2015, pp. 95-103), (Bartkowski, 2015). Mark Galeotti notes the problem: "the word hybrid is catchy, since it may represent a mix of anything" (Galeotti, 2016). Many scholars have used the term hybrid warfare to present their research on the Ukrainian crisis, but the literature is fragmented. As consequence, the theory of hybrid wars has been widely criticized. In relation to the Ukrainian crisis, authors have suggested other conceptualizations of the conflict. Robert

Heinsch, examines the conflict as a proxy war, Janis Berzins, argues that it is an instance of a new generation warfare, Timothy Thomas notes that it is indirect and asymmetric conflict, Oscar Jonsson and Robert Seely, on the other hand study it as a full-spectrum conflict (Heinsch, 2015), (Bērziņš, 2015), (Thomas, 2015), (Seely, 2015).

It is evident that there is a discrepancy between the public acceptance of hybrid warfare as the new Russian military approach and the academic debate regarding the appropriateness of such a classification. The lack of a comprehensive work which examines all aspects of the Ukrainian crisis through the lens of the hybrid war theory has increased the need for such an analysis. Is the Ukrainian crisis a hybrid conflict or not? This work will address that knowledge gap. It will present the key academic achievements in the field of hybrid warfare, argue why the theoretical framework developed by Timothy McCulloch is an appropriate analytical tool (Johnson, 2013). Then his theory will be applied to the Ukrainian case, which will show if the Russian actions during the crisis can be conceptualized as an instance of a hybrid war or not. Thus, the findings of the thesis will present valuable knowledge, which will contribute to the debate regarding the appropriateness and novelty of the hybrid war theory as a conceptualizing tool for the assessment of the new Russian military approach manifested during the Ukrainian crisis.

# 1.2. Research question

To what extent can the Russian actions in the Ukrainian crisis (2013-2015) be conceptualized as an exemplification of a hybrid war according to the theory of McCulloh?

# 1.3 Sub research questions

- 1. What are hybrid wars?
- 2. How can the Ukrainian crisis (2013-2015) be studied through the lens of the hybrid war theory?
- 3. Which expressions of hybrid warfare in the physical domains of operations can be identified in the Ukrainian crisis?
- 4. Which expressions of hybrid warfare in the non-physical domain of operations can be identified in the Ukrainian crisis?

#### 1.4. Societal and Academic relevance

The Russo-Georgian War from August 2008, exposed some of the problems in the Russian Army (BBC, 2008). Although psychological and cyber operations were conducted, the conflict resembled a classic invasion, that relied mostly on regular land forces with heavy weaponry, supported by the air forces and the Black Sea fleet (Minasyan, 2015, p. 5), (Geers, 2015, pp. 70-77). As a consequence, Russia had suffered high equipment losses (McDermott, 2009, pp. 70-73). One month after the end of the five-day war, during a session of the collegium of the Russian Defense Ministry, the Minister Anatoliy Serdyukov, declared the beginning of a major military reform (Gazmanov, 2011, p. 35). It was announced that the reform will be divided into three stages, during which the number of military personnel will be decreased, a new command and control structure will be developed, the military education system will be modernized, militaries will be trained for the execution of new tasks and the military equipment will be updated to the most modern weapons systems by at least 30% by 2015 and by 100% by 2020 (Gazmanov, 2011, p. 36). In 2013, the Russian magazine "Voenno-promishlenii Kurier" published an overview of a report titled: "Main trends and developments of forms and methods of using the Armed Forces, current tasks of military science and its improvement" which was prepared by General Valery Gerasimov, Chief of the Russian General Staff (Gerasimov, 2013). In the overview, it is noted that "in the 21st century the line between state of war and peace is blurred" and that "rules of war have changed significantly" (Gerasimov, 2013). Gerasimov argues that "asymmetric actions, allow to level the superiority of the enemy in an armed struggle" (Gerasimov, 2013). He continues by stating that the importance of "information confrontation" (information warfare) has risen. Overall, it can be seen that the Russian military is undergoing significant transformations.

During the Ukrainian crisis, Russia managed to annex Crimea in less than a month with no casualties (OHCHR, 2014). Compared to the actions in Georgia, it is evident that something has changed. In May 2016, the President of Estonia, Toomas Hendrik, stated: "We need a credible deterrence policy in the Baltic region to influence the Russian calculus to make the costs of interference too high" (Solovjova, 2016). With the increased concerns in NATO Eastern European countries regarding potential Russian aggression in this region, it is essential for policymakers and

<sup>&</sup>lt;sup>1</sup> It needs to be noted that according to news reports, two Ukrainian soldiers were killed, however in the United Nations (UN) report there in no such data (Reuters, 2014).

strategy planners to have a better understanding of the new Russian military approach (Independent, 2015), (Solovjova, 2016). However, Richard Sokolsky notes another problem. He argues that an exaggeration of "the new Russian threat" may "increases the likelihood of reaching incorrect conclusions that could work to Russia's advantage" (Sokolsky, 2017). Therefore, the examination of the appropriateness of the conceptualization of the Russian actions in Ukrainian as hybrid warfare will serve a clear societal purpose. Because once a unified understanding is obtained, then suitable defense measures can be developed.

In crisis management literature, an extensively examined notion is the "resilient society" concept. Edwin Bakker and Beatrice de Graaf present a broad definition of resilience: "the capacity of materials, persons or biotopes to resist sudden changes or stress, as well as the capacity to recover and return to the situation as before" (Graaf, 2014, p. 6). Rudolph Giuliani, a former mayor of New York City, argues that "a resilient society depends on active, engaged citizens" (Giuliani, 2008). In that respect, academia (especially in the field of security studies) should have an important role for the overall security of a state. Because the more a specific phenomenon is studied, the better control a society would have over it.

As is was presented in section 1.1. there is an ongoing academic debate regarding the representation of the new Russian military approach, revealed during the Ukrainian crisis, as hybrid warfare. On one side, there are scholars who argue that this is the most appropriate way to describe the new Russian military approach in the 21<sup>st</sup> century and on the other side, there are researchers who claim that the hybrid war theory is not suitable. This lack of cohesion in academia creates ambiguity. The fact that there is no exhaustive study that applies the hybrid war theory to the Ukrainian crisis, increases the academic relevance of this work. This research will shed light on the debate. It will apply the theory of hybrid warfare to the Ukrainian case, thus demonstrating if the Russian actions during the crisis can be described as hybrid warfare or not. As consequence, this work will make one step towards the establishment of a common framework for the analysis of the 21<sup>st</sup> century Russian military approach. Once this is achieved, the overall resilience of the societies that feel threatened by Russia will be improved.

#### 1.5. Initial remarks on hybrid warfare

Before structure of the thesis is presented it is important that the reader is acquainted with the differences between conventional wars and hybrid wars. What follows is a short description of the academic developments on this topic.

#### 1.5.1. Body of knowledge

The new hybrid form of conflict differs from conventional wars in a number of aspects. If in the past wars were an official and opened confrontation between states that use their national armies for the fulfillment of specific goals, today there are certain constraints such as nuclear parity, military alliances, and international law, which deter countries from declaring war as the costs (including: economic, human, political, social, developmental and, strategical) of such a decision would exceed the benefits (economic, strategic and, geostrategic) (Korybko, 2015, p. 9). Instead, military and strategic planners are developing new unconventional and indirect strategies that overcome the aforementioned "barriers" (Korybko, 2015, p. 9). Hybrid wars resemble this new approach in military planning.

As it was noted in section 1.1., hybrid warfare emerged as a term in the early 2000s in the United States professional military literature (Nemeth, 2002). However, it was theorized in 2007 by Frank Hoffman in his work "Conflict in the 21st century: The rise of hybrid wars" (Hoffman, 2007). By examining the 2006 Israel–Hezbollah War, he argues that "Hybrid Wars incorporate a range of different models of warfare, including conventional capabilities, irregular tactics and formation, terrorist acts including indiscriminate violence and coercion, and criminal disorder" (Hoffman, 2007, p. 14). His insights are criticized by Timothy McCulloch for being too narrow and connected only to the Israel–Hezbollah case (Johnson, 2013, pp. 9-10). McCulloch's understandings will be presented more extensively in chapter two.

Since the publication of Hoffman's work, one of the two dominant lines of research connected to this form of warfare was the inability of the North Atlantic Treaty Organization to respond to such conflicts. The main argument is that the capabilities of the alliance are predominantly conventional (military personnel, conventional weapons), whereas the new

potential adversaries will use unconventional tactics and irregular means to avoid direct confrontation and target the alliance's weaknesses (Blum, 2015), (Bachmann, 2015), (Michael Aaronson, 2011), (Gunneriusson, 2012). Take for instance the 2007 cyber-attacks on Estonia. The country, a member state of NATO, suffered from a series of cyber-attacks, that blocked both public and private websites (Traynor, 2007). Although this was an act of aggression, NATO was incapable of initiating actions according to Article 5 (NATO, 2016) (the key section of the Washington Treaty that presents the principle for Collective defense) since there was no evidence of the perpetrators of the attack. This clearly illustrates that there are specific situations where an adversary may initiate offensive actions and cover the traces leading to him/her.

However, after the annexation of Crimea in March 2014, the alliance acknowledged the risks connected to hybrid warfare (NATO, 2014). In 2014, after the September Wales Summit, the Heads of State signed a declaration in which it was stated: "We will ensure that NATO is able to effectively address the specific challenges posed by hybrid warfare threats" (NATO, 2014). In December 2015 NATO Ministers of Foreign Affairs approved a new strategy on hybrid warfare threats (NATO, 2015). Also, NATO and the EU are working together on a "Joint Framework on countering hybrid threats a European Union response" (Commission, 2016). Yet, there is no officially accepted classification of hybrid warfare but the threats that this form of conflict poses are defined as "a wide range of overt and covert military, paramilitary, and civilian measures employed in a highly integrated design." (NATO, Wales Summit Declaration, 2014).

The second dominant line of research in hybrid warfare literature is the classification of the Ukrainian crisis as an example of a hybrid war. <sup>2</sup> As it was noted in section 1.1., the literature is fragmented and there is a variety of different focuses (legal challenges, information warfare, cyber warfare, civil defense etc.) (Vlasiuk, 2015), (Korhonen, 2015), (Snegovaya, 2015), (Geers, 2015, pp. 95-103), (Bartkowski, 2015). A different approach can be observed in the work of the Russian scholar Andrew Korybko (Korybko, 2015). The author examines the role of "Color Revolutions", which in other words is a civil unrest against the government of a specific country, and the first step in an offensive hybrid war strategy. By exploring contemporary protest movements in Europe (Ukraine, Hungary, Serbia, Macedonia, Greece, Turkey), Asia (Armenia, Iran, Turkmenistan,

<sup>&</sup>lt;sup>2</sup> It has to be noted that not all works focus only on Ukraine as a case study. See (Lasconjarias, 2015), (Hoffman, 2007), (Moreland, 2016)

Uzbekistan, Kyrgyzstan), and the Middle East (the Arab spring in general and focusing on Syria), he argues that specific patterns can be found and that such crises are deliberately triggered (Korybko, 2015, pp. 11-115). Korybko notes that these protest movements, including the one in Ukraine, serve as a foreign policy instrument of the United States (US) (Korybko, 2015, pp. 25-32). Although the pro-Russian perspective of Korybko's works, his insights would be useful for the thesis because the author explains how citizens can be influenced by propaganda and organized through social media in order to carry out a hybrid attack. His work also shows the Russians understanding of hybrid warfare (Kasapoglu, 2015, p. 2).

From this section, it can be seen that hybrid war literature is not cohesive. Although this form of conflict can easily be distinguished from conventional wars, and NATO has recognized it, there is no generally accepted theoretical framework.

#### 1.6. Structure of this thesis

The thesis consists of six chapters. The following part will present key aspects in the academic understandings of the evolution of warfare and then will outline the theoretical framework through which the Ukrainian crisis will be studied. This chapter will give an answer to the first sub research question "What are hybrid wars?". The third chapter will introduce the research design of the thesis. In it the methods and units of analysis will be presented, following an operationalization of the theory of hybrid warfare in clear indicators, answering the question "How can the Ukrainian crisis (2013-2015) be studied through the lens of the hybrid war theory?". The qualitative exploratory case study of the Ukrainian crisis will be divided into two chapters. The fourth chapter will focus on elements of hybrid warfare in the physical domains while in the fifth, the non-physical (cognitive) domain will be discussed. Thus, the questions "Which expressions of hybrid warfare in the physical domains of operations can be identified in the Ukrainian crisis?" and "Which expressions of hybrid warfare in the non-physical domain of operations can be identified in the Ukrainian crisis?" will be answered in these chapters. The final sixth chapter will present the conclusions of the thesis. In it the central research question "To what extent can the Russian actions in the Ukrainian crisis (2013-2015) be conceptualized as an

exemplification of a hybrid war?" will be answered. Also, the limitations of the study will be highlighted and suggestions for future research will be presented.

# 2. Theoretical framework

"Everything is very simple in war, but the simplest thing is difficult"

Carl von Clausewitz, 1873 (Clausewitz, 1873, p. 40)

In this chapter, the theoretical framework on which the study will be built will be presented and also an answer to the first sub research question "What are hybrid wars?" will be suggested.

# 2.1. Evolution of the theories of warfare - hybrid conflict origins

The theory of hybrid wars represented a step in the evolution of warfare. However, it is not especially new, as it is built upon other warfare theories (Johnson, 2013, p. 6). The rationale behind the term hybrid<sup>3</sup> warfare is the combination of different forms of conflict. First Frank Hoffman, then Timothy McCulloh and Richard Johnson base their works on previous studies which illustrate the changing nature of war (Johnson, 2013, p. 3), (Hoffman, 2007, p. 30). To a significant degree, the authors to whom they refer overlap. Therefore, it is important that the reader is acquainted with them. According to McCulloh, the origins of hybrid warfare can be found in two well-acknowledged theories - fourth generation warfare and compound wars.

#### 2.1.1. Fourth Generation Warfare

The generation theory of modern warfare, proposed by Thomas Hammes in his book "The Sling and the Stone: On War in the 21st Century", examines the transformations in military technology and its effects on the tactical, strategic and social levels (Hammes, 2004). The starting point of Hammes's research is 1648 (a key year in international relations – the end of the Thirty Years' war and the establishment of Westphalian sovereignty (Kissinger, 1994, pp. 65-66) ). He argues that during the first three generations of modern warfare (the period between the 17<sup>th</sup> and the late 20<sup>th</sup> centuries), war was an activity strongly linked to state actors, and the differences between these generations came from to the application of new military technologies and tactics (Hammes, 2004, p. 7). However, since the mid to late 20<sup>th</sup> century Hammes reasons, that a fourth generation of warfare has emerged (Hammes, 2004, p. 7). The key difference between this

<sup>&</sup>lt;sup>3</sup> The word hybrid can be defined as a "a thing made by combining two [or more] different elements" (Oxford, 2017).

generation and the other three is the rise of non-state actors that call into question the legitimacy of the state (Hammes, 2004, p. 7). The fourth generation is characterized by the use of conventional and unconventional means, including guerrilla warfare, insurgency, influencing tactics which are applied in order to neutralize the conventional military capabilities of an enemy (Hammes, 2004, p. 7). From this theory, McCulloh adopts the idea that in modern military conflicts, by the combination of conventional and unconventional means, non-state actors are capable to effectively fight against a regular military forces (Johnson, 2013, pp. 7-8).

#### 2.1.2. Compound Wars

In his book "Compound Warfare: That Fatal Knot" Thomas Huber expresses similar observations as Hammes (Huber, 2002). The author describes this form of conflict as the use of both regular and irregular forces that are coordinated under the same command (Huber, 2002, p. 8). The effectiveness of this form of conflict emanates from the exploitation of the advantages of both types of forces which reinforce each other (Huber, 2002, p. 8). The regular forces benefit from the speed, intelligence, and counterintelligence of the irregular militaries, while they on the other hand benefit from the operational pressure that the conventional forces put on the adversary and also their military logistics and strategic intelligence assets (Huber, 2002, p. 8). From the theory of compound wars, McCulloh accepts the complementary effect of the combination of both conventional and unconventional forces which operate under a unified command (Johnson, 2013).

#### 2.1.3. Hybrid Wars

The monograph "Hybrid Warfare" by Major Timothy McCulloch and Major Richard Johnson, from the US Armed Forces, resembles an exhaustive research in the field of hybrid wars. The work is divided into two sections. The first part "The Inadequacy of Definition and the Utility of a Theory of Hybrid Conflict: Is the 'Hybrid Threat' New?", written by McCulloch, presents a complete theoretical framework through which this form of conflict can be studied. The second section "Operational Approaches to Hybrid Warfare" by Richard Johnson suggests measures on the operational and strategic level on how the US Army can respond to this form of conflict.

In the first part, after presenting the predating theoretical achievements which express the transformative nature of modern conflicts (see section 2.1.1. and 2.1.2.), McCulloch continues his analysis by focusing on the "newest" concept – hybrid warfare. He describes Frank Hoffman's "Conflict in the 21st century: The rise of hybrid wars" as a "benchmark" that has influenced the majority of works in hybrid warfare (Hoffman, 2013, p. 9). McCulloch argues that Hoffman's

postulates build upon the above-mentioned theories<sup>4</sup> by suggesting that not only the new wars will be waged by state and non-state, by the use of both conventional and nonconventional means, but also that social disruptive actions (including terrorism and criminal disorder) will be applied in order to generate a synergetic effect (Hoffman, 2013, p. 9).

The author also explores other works that discuss hybrid warfare. He refers to Colonel Jeffrey Cowan, who argues that globalization and the proliferation of information and technology create a situation where a non-state actor can gain access to knowledge and means, that were previously only under the control of state actors (Johnson, 2013, p. 11). Moreover, McCulloch discusses Lieutenant Colonel Daniel Lasica's insights on hybrid warfare, who considers that actors that wage hybrid wars will deliberately exploit the cognitive and moral domains, by taking advantage of the enemy's mistakes (Johnson, 2013, p. 12). Lasica deliberates that hybrid warfare is both strategy and tactic and that the hybrid actor can be much faster than a conventional one (Johnson, 2013). Israeli military specialists advance this idea and suggest that the hybrid actor will have a network organizational structure, which compared to hierarchies is much more flexible and prompt in terms of actions on the battlefield (Johnson, 2013, p. 10). Not only McCulloch's understandings of hybrid warfare are influenced by the described above literature, but also, he borrows ideas from the discussed authors and specialist. However, his approach is different.

# 2.2. McCulloh's theory

Taking into account the works described in the previous section, McCulloch argues that hybrid warfare literature is predominantly descriptive and fails to grasp the underlying logic of a hybrid actor. He also criticizes Frank Hoffman's widely accepted work, for being connected only to the Israel–Hezbollah case (Johnson, 2013, pp. 9-14). He states that Hoffman's paper "does not capture a concise form, function, and logic to explain a hybrid organization that conducts hybrid warfare" (Johnson, 2013, p. 9). The author argues that a theory of hybrid warfare should be composed of principles that will facilitate "a broad understanding or rationale for hybrid organizations' existence", thus enabling prediction (Johnson, 2013, p. 9).

\_

<sup>&</sup>lt;sup>4</sup> In his work, Hoffman also uses the concept of "Unrestricted Warfare" or "war beyond limits" developed by two Colonels from the Chinese People's Liberation Army, which is not mentioned by McCulloh; see in (Hoffman, 2007, pp. 22-25)

McCulloch asserts that countries with developed armies are trained for offensive, defensive and stabilizing operations yet, their forces are "not prepared for a specific employment context, but rather optimizes to best meet a broad array of scenarios for employment<sup>5</sup>" (Johnson, 2013, p. 15). Meaning that the means and strategies of conventional armed forces are developed to serve specific pre-planned military scenarios. However, these plans often fail to acknowledge the unique characteristics of the specific conflict situation. The author reasons that such "optimization", creates situations where the effectiveness of a military operation can be altered by unforeseen developments, resulting from the unique context. On the other hand, McCulloch affirms that actors with lower capacities will increase their effectiveness against a stronger, conventional force by utilizing unconventional means and by taking full advantage of the unique operational environment. (Johnson, 2013, p. 15). Thus, a hybrid organization can effectively and efficiently fight a stronger conventional opponent. McCulloch defines hybrid wars as an: "optimized form of warfare that allows a combatant to attempt to utilize all available resources—both conventional and unconventional—in a unique cultural context to produce specific effects against a conventional opponent" (Johnson, 2013, p. 16). By conducting a historical analysis of the Israel-Hezbollah War (2006) and the Second World War (1941-1945), particularly the Eastern Front and the Soviet Partisan Network, he validates his theory and proves that both Hezbollah and the Soviet Partisan can be examined as hybrid entities that have successfully fought stronger adversaries (Johnson, 2013, pp. 19-39).

His theory comprises seven principles which enable a broader understanding of this form of conflict. He describes them as follows:

The first principle of hybrid war [...] is that a hybrid force's composition, capabilities, and effects are unique to the force's own specific context. This context relates to the temporal, geographic, socio-cultural, and historical setting in which the given conflict takes place.

The second principle is that there exists a specific ideology within the hybrid force that creates an internal narrative to the organization. This ideology is inherently linked to the strategic context and is grounded within the socio-cultural, religious identity of the

<sup>&</sup>lt;sup>5</sup> To employ (US military term) - integrating ends, ways, and means (Gortney, 2010, p. 175)

hybrid force. The resulting narrative serves to redefine the extant rules within the strategic context.

The third principle is that a hybrid force perceives an existential threat by a potential adversary. This perceived threat drives the hybrid force to abandon conventional military wisdom to achieve long-term survival.

The fourth principle is that a capability overmatch between the hybrid force and a potential adversary exists. The hybrid force contains less conventional military capability in comparison to its adversary and therefore must seek a way to offset this apparent advantage in military capability.

The fifth principle is that a hybrid force contains both conventional and unconventional elements. These elements often comprise "accepted" military technology and nonmilitary, guerrilla type technology. The elements may also include the use of terrorist or other criminal tactics. These combined capabilities create an asymmetric advantage for the hybrid force.

The sixth principle proposes that hybrid organizations rely on inherently defensive type operations. The hybrid force seeks to defend its existence and employs an overall strategy of defensive operations. These operations will often include offensive components, but the overarching intent is still one of defense.

The seventh principle is that hybrid organizations use attritional tactics in the employment of the hybrid force. These tactics manifest in both the physical and the cognitive domains in order to continually whittle away the adversary's forces and his will to use them (Johnson, 2013, pp. 16-17).

McCulloch argues that these principles can serve as a "predictive template that can be used to baseline the analysis of a hybrid threat", and defines his hybrid warfare theory as a "form of warfare in which one of the combatants bases its optimized force structure on the combination of all available resources—both conventional and unconventional—in a unique cultural context to produce specific, synergistic effects against a conventionally-based opponent" (Johnson, 2013, p. 37), (Johnson, 2013, p. 17).

It can be seen that McCulloch's theoretical framework is exhaustive. It moves beyond the predominantly descriptive hybrid warfare literature. He not only acknowledges the importance of

the cognitive domain, but also enables the exploration of the rationale of the hybrid actor. Compared to other works, McCulloch's framework is more specific and systematic. The seven principles of hybrid warfare will be easily transformed into observable indicators, which will facilitate the exploration of the new Russian military approach, exposed during the crisis in Ukraine. Therefore, McCulloch's theory is an appropriate guideline for this particular study.

#### 2.2.1. Justification of the choice

Although Frank Hoffman's pioneering theoretical framework has been applied in a number of studies on hybrid warfare and it is widely acknowledged, this research will use the less popular framework developed by Timothy McCulloh. This decision can be justified by three main reasons.

Firstly, Hoffman does not explicitly suggest that an actor that wages a hybrid war will exploit the cognitive domain. He argues that "terrorist acts including indiscriminate violence and coercion, and criminal disorder" will be used as a "shock and awe" tools that will distort the will to fight of the conventional opponent, however he does not argue that the hybrid actor can exploit the cognitive domain, through other means (Hoffman, 2007, pp. 29-30). On the other hand, McCulloh has explored this matter in his seventh principle (Johnson, 2013, p. 17).

Secondly, Hoffman's framework does not examine the underlying logic of a hybrid actor (why the hybrid actor has emerged, why he fights, what are their objectives), in other words as McCulloh states, the "rationale for hybrid organizations" existence", which he addresses in the first tree principles (Johnson, 2013, p. 9).

Thirdly, and maybe most importantly Hoffman's work is largely descriptive and does not provide concrete guidance for the application of the theory. In contrast, McCulloh's framework is future orientated and it is intentionally developed to serve as an analytical tool, that enables an indepth understanding of the characteristics of the hybrid adversary (Johnson, 2013, p. 35).

Therefore, for the exploration of the Ukrainian crisis, McCulloh's framework will be used. The suggested seven principles will allow an exhaustive investigation of the logical basis of the hybrid actor also, they will facilitate the analysis, by serving as a template.

#### 2.3. Theoretical framework – conclusions

From the previous sections, it can be concluded that hybrid wars are not only a form of conflict, but also a strategy and a tactic. Such conflicts can be initiated by both state and non-state actors. Their effectiveness and efficiency derive from the exploitation of the unique context of the conflict and the optimization of the hybrid entity's force structure by the utilization of both conventional and unconventional resources, that are simultaneously applied in both the physical and cognitive domains, thus producing a synergistic effect. This allows the hybrid entity to overcome a more powerful (in terms of resources and equipment) conventional actor.

The theoretical framework developed by Timothy McCulloh is an appropriate tool for the exploration of the Ukrainian crisis, which will enable to investigate if the new Russian military approach can be conceptualized as an instance of a hybrid war.

# 3. Research Design

"It is a capital mistake to theorize before one has data. Insensibly one begins to twist facts to suit theories, instead of theories to suit facts"

Arthur Conan Doyle, 1892 (Doyle, 1892, p. 7)

In this chapter, the research design of the thesis will be presented, along with the methodology that is going to be used and an operationalization of McCulloh's hybrid warfare theory. In doing so, the second sub-research question: "How can the Ukrainian crisis (2013-2015) be studied through the lens of the hybrid war theory?" will be answered.

#### 3.1. Research Design

The investigation of the central research question (see section 1.2.) needs to be carefully organized in order to allow an in-depth exploration of the new Russian military approach, expressed in the Ukrainian crisis.

In his book "Case study research: Design and methods", Robert Yin argues that when a researcher is trying to understand a complex real life social phenomenon, a suitable strategy would be the development of a case study (Yin, 2003, pp. 1-2). He defines it as "an empirical inquiry that investigates a contemporary phenomenon within its real-life context; when the boundaries between phenomenon and context are not clearly evident; and in which multiple sources of evidence are used" (Yin, 2003, p. 13).

The Ukrainian crisis is a present-day problem, which exposes the multiplex nature of the modern Russian military approach. Therefore, an appropriate way to study it would be the conduction of a case study.

Yin notes that case studies can be divided into three groups: explanatory, exploratory and descriptive, depending on the objective of the research (Yin, 2003, p. 1). Since the primary focus of this thesis is to investigate if the new Russian military approach, revealed during the Ukrainian crisis, can be conceptualized as an example of a hybrid war, it can be reasoned that the case study that is going to be developed should be exploratory. However, the three categories of case studies are not mutually exclusive, and the thesis will also have an explanatory focus.

As already presented in chapter two, hybrid warfare resembles a multi-aspect form of conflict that incorporates a number of different elements including different actors (state and non-state), means (conventional unconventional), tactics (direct and indirect) that are applied in both the physical and cognitive domains. In order to facilitate the analysis, the aspects of hybrid warfare need to be operationalized. This will be achieved by the suggestion of two units of analysis (in section 3.3.2. justification of this choice is presented). The first will focus on expressions of hybrid warfare in the physical domains of operations and the second, on expressions of hybrid warfare in the non-physical domain of operations. Also, the Ukrainian conflict is an ongoing crisis (since 2013), therefore a time boundary needs to be set. The case study will cover the period from November 2013 to February 2015 because this was the active phase of the conflict. The design of the case study is presented in figure one.

<sup>.</sup> 

<sup>&</sup>lt;sup>6</sup> On the 12<sup>th</sup> of February, a truce was signed between representatives from Ukraine, Russia, OSCE, Donetsk People's Republic and Lugansk People's Republic (OSCE, 2015). Although almost every day since the signing of the ceasefire there are reports, marking violations from both sides, February 2015 is an appropriate time boundary because since then, there have not been any heavy fights in Donbass (OSCE, 2017). It has to be noted that in some occasions, the analysis may go beyond the time boundary, yet the dominant focus will be on the period November 2013 to February 2015.



Figure 1: Single-case Design

# **Embedded**

(multiple units of analysis)

## 3.2. Methodology

In order to increase the validity of the of the study, two analytical approaches will be used. The main method of analysis - process tracing (PT) will be supplemented by the conduction of discourse analysis.

#### 3.2.1. Process Tracing

This research method is defined by Alexander George and Andrew Bennett as:

the use of histories, archival documents, interview transcripts, and other sources to see whether the causal process a theory hypothesizes or implies in a case is in fact evident in the sequence and values of the intervening variables in that case (Bennett, 2012, p. 8).

Derek Beach and Rasmus Brun argue that "in qualitative in-depth single case studies using PT enables the researcher to make strong within-case inferences" (Brun, 2011, p. 4). They continue by noting that process tracing incorporates three distinct variants of analysis which are theory-testing, theory-building, and explaining outcomes (Brun, 2011, p. 6). In relation to the first one (theory-testing), the authors reason that it "deduces a theory from the existing literature and then tests whether there is evidence that a hypothesized causal mechanism is actually present in a given case" (Brun, 2011, p. 6). They suggest a three-step model which encompasses the analysis (it will be extensively examined in section 3.3.).

Since the key goal of the thesis is to explore if the hybrid warfare theory is a suitable analytical tool for the conceptualization of the new Russian military approach (exposed during the Ukrainian crisis), process tracing will serve as an appropriate technique for the analysis of the two units suggested in section 3.2.

#### 3.2.2. Discourse Analysis

Nelson Phillips and Cynthia Hardy define discourse as: "an interrelated set of texts, and the practices of their production, dissemination, and reception, that brings an object into being" (Hardy, 2002, p. 4). They continue by noting that discourse analysis explores the relation between discourse and reality and how they affect each other (Hardy, 2002, p. 4). By referring to Jacques Derrida, Lene Hansen argues that "meaning is established not by the essence of a thing itself but through a series of juxtapositions, where one element is valued over its opposite" (Derrida, 2006, p. 19). These ideas also correspond to Rein and Schön's policy issue's framing approach, where the focus

is on the language used in policy discourse, which involves specific metaphors, stories, and categories that are used to frame a specific phenomenon (Rein, 2013, p. 229).

This method will be used as a tool for the exploration of expressions of hybrid warfare in the non-physical domain of operations. Because some of the aspects of Russia's information warfare in Ukraine are connected to the unique setting of the conflict. Discourse analysis will allow these nuanced characteristics to be studied.

# 3.3. Theory-testing Process Tracing

In section 3.2.1., it was noted that three different variants of process tracing analysis exist. The first one – theory testing PT, fits the objective of this particular work. Derek Beach and Rasmus Brun argue that for its execution a researcher needs to follow three distinct steps (Brun, 2011, p. 10).

#### 3.3.1. Conceptualization of a causal mechanism

At the core of theory-testing PT is the exploration of an assumed causal mechanism<sup>7</sup>, or how X causes Y (Brun, 2011, p. 7). In other words, the scholar searches for actual evidence that will support the predictions of a theory within a specific case (Brun, 2011, pp. 5-7). Beach and Brun state that often, within the theory that is being examined, X and Y are present, yet the causal mechanism between them is not explicitly conceptualized (Brun, 2011, pp. 7-8). Therefore, the first step in theory-testing PT is to is to "conceptualize a causal mechanism between X and Y based upon existing theorization" (Brun, 2011, p. 9).

In Timothy McCulloh's hybrid war theory, X can be identified as the actor (that "bases its optimized force structure on the combination of all available resources—both conventional and unconventional—in a unique cultural context to produce specific, synergistic effects against a conventionally-based opponent" (Johnson, 2013, p. 17)) and Y is the effect – hybrid warfare. Yet, his theory does not clearly express causality. This can be surmounted by a reformulation of McCulloh's definition of the hybrid war theory (see original definition in section 2.2.). For example, as: If during a conflict one of the combatants – X bases its optimized force structure on the combination of all available resources - both conventional and unconventional - in a unique

<sup>&</sup>lt;sup>7</sup> One of the definitions of a causal mechanism is "a complex system, which produces an outcome by the interaction of a number of parts" (Brun, 2011, p. 4)

cultural context to produce specific, synergistic effects against a conventionally-based opponent, then this entity is waging a hybrid war - Y. When this causal mechanism is applied to the particular focus of this case study, then X becomes the Russian Federation, Y - hybrid warfare, and the characteristics of this form of conflict described by Timothy McCulloh, resemble the relation between X and Y. To clarify, if during the Ukrainian crisis, Russia's actions meet the characterization of hybrid warfare (as presented by Timothy McCulloh) then their new military approach can be conceptualized as an exemplification of a hybrid war.

#### 3.3.2. Operationalization

According to Beach and Brun, the second step in theory-testing PT is to transform the theoretical expectations into observable indicators of the causal mechanism (Brun, 2011, pp. 8-10).

The theoretical framework, which was extensively examined in chapter two, offers a number of conditions that need to be met in order to classify a conflict as a hybrid one. Particularly the seven principles of hybrid wars and the definitions of both this form of conflict and the theory describing it, which are suggested by Timothy McCulloh, will be especially useful.

However, if all of the characteristics of hybrid warfare are simultaneously used for the research of the Russian military approach in Ukraine, the analysis will become too broad and confusing. Therefore, in order to avoid an "information overload", two units of analysis were suggested. The features of hybrid warfare will be divided into two groups in consonance with the units. It has to be noted that two of McCulloh's seven principles (the first and seventh) of hybrid warfare affect both the physical and non-physical domains, thereof they will be used twice.

The first unit is expressions of hybrid warfare in the physical domains of operations. Which include actions taking place on land, also in air, sea, and space (Thibault, 2005). In other words, where all matter interacts (Veritas, 2005). Five of the principles fall into this group.

According to the first principle developed by McCulloh, an actor that is initiating a hybrid war will build one's forces composition and capabilities in relation to the unique context of the conflict. Hence, the hybrid actor takes full advantage of the conflict situation. The author describes the context as the "temporal, geographic, socio-cultural, and historical setting" (Johnson, 2013, p. 16). The exploitation of the context characteristics, deriving from the unique temporal, geographic, socio-cultural, and historical setting, that affect the physical domains will be used as indicators for the analysis.

The fourth principle entails that the conventional opponent has more power in terms of military capabilities, therefore the hybrid actor seeks ways to offset this disadvantage. A relevant indicator for this principle can be the use of light weapons<sup>8</sup> by the hybrid actor, for the neutralization of heavy weaponry systems<sup>9</sup> of the conventional opponent.

The fifth principle is also connected to the physical domains. McCulloh states that the hybrid actor holds both conventional and unconventional elements which create an asymmetrical advantage. In this instance, appropriate indicators will be the use of conventional military technology along with unconventional tactics and means, including irregular militaries, terrorism, sabotage and criminal disorder.

McCulloh's sixth principle states that the hybrid actor will rely mostly on a defensive strategy and operations. The author however, does not exclude offensive action. The indicator for this principle is self-evident – predominantly defensive strategy and actions.

The last principle developed by McCulloh points that an actor that is waging a hybrid war will use attritional tactics in both the physical and cognitive domains in order to not only undermine the morale of the conventional forces, but also to diminish their capabilities. A stalemate situation of the conflict is a suitable indicator for this principle.

The second unit of analysis within the Ukraine case study will explore expressions of hybrid warfare in the non-physical (cognitive<sup>10</sup>) domain of operations. Four of McCulloh's principles can be related to this group.

According to the first principle, the effectiveness of a hybrid actor is connected to the exploitation of the unique context of the conflict (the characteristics of which are described above). The exploitation of the context characteristics (in the non-physical domain) will be used as indicators for the analysis.

The second principle entails that a hybrid actor will develop a specific ideology which supports and justify one's actions. It is linked to the unique context of the conflict (including the

31

<sup>&</sup>lt;sup>8</sup> According to the UN classification light weapons are: "weapons designed for use by two or three persons serving as a crew, although some may be carried and used by a single person. They include, inter alia, heavy machine guns, handheld under-barrel and mounted grenade launchers, portable anti-aircraft guns, portable anti-tank guns, recoilless rifles, portable launchers of anti-tank missile and rocket systems, portable launchers of anti-aircraft missile systems, and mortars of a calibre of less than 100 millimetres" (UN, 2008, p. 19)

<sup>&</sup>lt;sup>9</sup> According to the UN classification major conventional arms include: battle tanks, armored combat vehicles, large-calibre artillery systems, combat aircraft and unmanned combat aerial vehicles (UCAV), attack helicopters, warships, missiles and missile launchers (UN, 2017)

 $<sup>^{10}</sup>$  The cognitive domain can be defined as: "the mental action or process of acquiring knowledge and understanding through thought, experience, and the senses" (Dictionaries, 2017)

socio-cultural setting and the religious identity of the hybrid force). The presence of a specific set of beliefs, developed by the hybrid actor will serve as an indicator for this principle.

McCulloh's third principle suggests that the hybrid actor has abandoned "conventional military wisdom" as a result of a perceived "existential threat by a potential adversary" and the will to ensure their long-term survival. The indicator for this principle will be the presence of an existential threat for the hybrid actor.

The seventh principle states that the hybrid actor will use both the physical and cognitive domains in order to prolong the conflict, weaken the conventional forces' morale and, undermine their capabilities. Relevant indicators for this principle would be the use of propaganda and disinformation (by the hybrid actor), as tools that trigger an emotional response of the targeted audiences.

McCulloh notes that the effectiveness of a hybrid war depends on the simultaneous application of all of the principles described above, which produces a synergistic effect.

#### 3.3.3. Collection of evidence

The third step in theory-testing PT is to collect evidence that will either support or disprove the proposed causal mechanism (Brun, 2011, p. 10).

Two main sources for collecting information will be used. First, primary sources: official governmental documents (both Ukrainian and Russian), official reports by the Organization for Security and Co-operation in Europe (OSCE) Special Monitoring Mission to Ukraine and the UN's Reports on Human Rights Situation in Ukraine. Also, a variety of secondary sources of information will be used including: news media reports (during the crisis both sides used disinformation tactics therefore, news reports will be retrieved from a number of news agencies: "The Ukrainian News Agency", "Russian News Agency TASS", "Reuters", "Associated Press", as well as online newspapers in Russian, Ukrainian, English, and Bulgarian), academic works (there is a significant number of researches of the Ukrainian crisis that will facilitate the analysis), reports from non-governmental organizations (NGO's), videos and photographs may also be used. The diversity of data sources will result in an in-depth investigation of the two units of analysis.

#### 3.4. Limitations

#### 3.4.1. External validity

Zaidah Zainal, argues that the key disadvantage of using a single-case design, is "its inability to provide a generalising conclusion, in particular when the events are rare" (Zainal, 2007, p. 2). However, she notes that this disadvantage can be surmounted by the combination of different methods of analysis (Zainal, 2007, p. 2). Therefore, the composition of theory-testing process tracing, along with discourse analysis will increase the external validity of the work. Also, the research is will explore only the Ukrainian crisis, and universal conclusions regarding the new Russian military approach will not be suggested. Yet, this narrow focus will allow the work to engage in the debate regarding the conceptualization of the Russian actions during the Ukraine crisis. Therefore, it can be argued that although only one case will be explored, the thesis will have a moderately high external validity.

#### 3.4.2. Internal validity

The combination of two methods of analysis will result in an exhaustive investigation of the suggested case. However, the methods have specific disadvantages that need to be noted.

Derek Beach and Rasmus Brun argue that theory-testing PT "cannot be used to make cross-case inferences beyond the single case" (Brun, 2011, p. 14). However, this disadvantage of the method would not affect the thesis, because as presented in the previous section, the focus is limited only to the Russian actions during the Ukrainian crisis.

One of the key critiques of discourse analysis is that it "does not provide definite answers", it is "not a hard science" and "meaning is never fixed […] everything is always open to interpretation and negotiation" (Mogashoa, 2014, pp. 111-112). These weaknesses of the method will be mitigated through its use only as a supplement analytical tool. Also, extensive justifications, regarding the findings obtained by it will be presented.

However, another potential pitfall, which may affect the whole study, has to be noted. It is connected to the reliability of the data sources. One of the key aspects of the Ukrainian crisis are

the information-psychological operations. In Russia, a significant part of the media is either state-owned or it is in the possession of "Kremlin-friendly businessmen" (Seely, 2015, p. 12). Also, the UN Human Rights Committee has expressed strong concerns regarding the freedom of expression in the Russian Federation (Committee, 2009). Therefore, the authenticity of Russia media materials is questionable. This disadvantage will be surmounted by the provision of verification (when possible) from other sources.

Despite the described above, the internal validity of the work will be moderately high because first, the suggested two research methods reinforce each other second, data will be derived from a number of sources and third, the potential shortcomings are acknowledged.

# 3.5. Research Design – conclusions

In this chapter, a suggestion of how the Ukrainian crisis (2013-2015) can be studied through the lens of the hybrid war theory was presented. The investigation will be organized as an exploratory case study with two units of analysis. Within the case study, two research methods will be used - theory testing PT and discourse analysis. The first two steps in theory testing PT, were carried out. As a result, Timothy McCulloh's hybrid war theory was reformulated in order to express causality and the seven principles of hybrid warfare were transformed into observable indicators which will direct the research. Evidence will be collected through analysis of both the discourse and a variety of documents. This would allow an in-depth investigation of the crisis. Although this research design has specific limitations, the overall validity of the work will be moderately high.

# 4. Expressions of hybrid warfare in the physical domains of operations during the Ukrainian crisis

"If I want to, I can take Kiev in two weeks"

Vladimir Putin, 2014 (Squires, 2014)

This chapter will address the first unit of analysis of the case study, which was presented in chapter three. The theoretical framework of hybrid warfare, developed by Timothy McCulloch, will be applied to the Ukrainian case. The analysis will be focused on the Russian actions in the physical domains of operations. In doing so, the third sub research question "Which expressions of hybrid warfare in the physical domains of operations can be identified in the Ukrainian crisis?" will be answered. Also, the findings from this chapter will make a step towards the explanation of the central research question. However, before that, the reader needs to be acquainted with the key developments during the Ukrainian crisis, as they are important for the whole case study.

#### 4.1. Ukrainian crisis - overview

#### 4.1.1. Inception

The Ukraine crisis (2013-present) begun as a civilian demonstration against the pro-Russian Ukrainian President Viktor Yanukovych (BBC, 2015). In November 2013, the President decided to cancel the signing of an Association Agreement with the European Union, and turned the country towards a closer relationship with the Russian Federation (BBC, 2013). Soon after the announcement of his decision, in the capital Kiev, thousands of Ukrainians took to the streets in a peaceful protest. The government responded by sending riot police in order to disperse the protesters. During the following months, violence in Kiev started escalating. In the front lines of the demonstrations, a group called Right Sector "a motley confederation of football hooligans and far-right nationalists", begun intense fights with the police forces (Economist, 2014). On the 20<sup>th</sup> of February 2014, more than fifty people were killed, in Kiev's Maidan square (Gatehouse, 2015). The next day, under the mediation of the European Union and Russia, President Yanukovych and representatives from the opposition met and signed an "Agreement on the Settlement of Crisis in Ukraine" (Guardian, 2014). However, the agreement did not last because, on the 22<sup>nd</sup> of February

2014, the Rada (Ukraine's Parliament) voted to dismiss President Viktor Yanukovych from his powers (Sindelar, 2014). A new provisional government took the power and Oleksandr Turchynov, was appointed as an interim president (Kyivpost, 2014), (BBC, 2014). The next day (the 23<sup>rd</sup> of February 2014), the Parliament canceled the law "On State Language Policy" which gave an official status of the Russian, Bulgarian, Romanian, Tatarian and Hungarian languages (MVnR, 2014), (Tchuzakov, 2014). In response, ethnic Russians which comprise the biggest minority group in Ukraine, 17.3% of the total population (CIA, 2017) started opposing the new government in Kiev. From that point, the crisis started to transform from a political upheaval against corruption, with demands for a new constitution and government, to a low-intensity ethnic conflict. As powerful international actors backed both opposing sides in Ukraine (Russia supported the Russian minority in Ukraine while, the US, the EU, and other the new Ukrainian government) the crisis took the form of a civil war with two main geographical areas of action (Secretary, 2014) (BBC, 2014) (Europe, 2014).

#### **4.1.2.** Crimea

The first one is the Crimean Peninsula, which is a historically contested area. Once part of the Russian Empire, then an autonomous republic within the Russian Soviet Federated Socialist Republic, in 1954 it was transferred to Ukraine but it remained in the Soviet Union (USSR) (Britannica, 2017). After 1991, the peninsula became a part of the newly independent Ukrainian state (Britannica, 2017). For the Russian Federation Crimea has a high geopolitical importance. During Soviet times in the peninsula was situated a significant part of the Soviet Black Sea Fleet. After the dissolution of the Soviet Union the faith of the fleet was uncertain. In 1997 Russia and Ukraine signed three agreements which governed the status of the former Soviet Black Sea Fleet (Kimball, 2014). Russia was allowed to station no more than 25.000 troops, 132 armored combat vehicles and 24 pieces of artillery at its military facilities in Crimea, until 2017 (Kimball, 2014). In 2010 under the presidency of Viktor Yanukovych, agreements were renewed with an expiry date in 2042 (Kimball, 2014). As part of the deal, Russia offered a 30% discount on the price of natural gas exported to Ukraine (Pan, 2014).

The population of the peninsula is predominantly Russian. According to a census conducted in 2001, 58.3 percent of the people are ethnic Russian (RT, 2014). The same day Oleksandr

<sup>&</sup>lt;sup>11</sup> An act which is not specified in Ukraine's constitution (Sindelar, 2014)

Turchynov became Ukraine's new president and the Parliament cancelled the law "On State Language Policy", in the city of Sevastopol (the biggest city on the peninsula), thousands of people went on the streets in order to protest against the new regime in Kiev and to express their willingness to join the Russian Federation (Guardian, 2014). The following days masked and armed men, carrying the latest Russian military equipment occupied key governmental buildings on the peninsula (Shevchenko, 2014). The pro-Russian protests continued and on the 6<sup>th</sup> of March, the Crimea Parliament voted to join the Russian Federation (Jazeera, 2014). On the 16<sup>th</sup> of March, a referendum was held and 97 percent of the Crimean people voted for separate from Ukraine and to join Russia (BBC, 2014). Two days later the Russian president signed laws according to which the Republic of Crimea and the city of Sevastopol became part of the Russian Federation (TASS, 2014).

The second geographical area of action is the Donbass<sup>12</sup> region.

## **4.1.3. Donbass**

In the beginning of March 2014 pro-Russian protest also broke out in other South-Eastern regions of Ukraine. Including Odessaka, Zaporizhska, Kharkivska, Mykolaivska, Dnipropetrovska, Khersonska oblasts (regions) (RT, 2014), (Roth, 2014). However, the strongest unrests took place in the Donetsk and Lugansk regions, where the majority of people are native Russian speakers (Krasnozhon, 2014).

On the first of March, in the city of Donetsk thousands of people went on a protest against the new government in Kiev (RT, 2014). They demanded for federalization of Ukraine, more power for their local government and closer ties with Russia (RT, 2014). The following days the protesters occupied governmental buildings and the demonstrations continued. After the Crimean Parliament voted to join Russia the manifestations intensified. On the 13<sup>th</sup> of March, a pro-Russian and a pro-Ukrainian protest collided leaving fifty people injured and two dead (Valeiskaq, 2014) (Donbasa, 2014). Three days later people started erecting picket lines on bridges and key junctures in the city in order to prevent Ukrainian authorities to send military personnel to the town (Duboyaq, 2014).

<sup>&</sup>lt;sup>12</sup> From Russian Donetsky Basseyn, a historical, cultural, and economic region in eastern Ukraine (Ray, 2014)

In the city of Lugansk, the events unfolded in a similar manner. The first big anti-governmental demonstration was held on the first of March (Ray, 2016). The following days people started opposing the local authorities and occupied governmental buildings (Ray, 2016).

April 2014, can be regarded as a turning point of the crisis. The separatists in both Donetsk and Lugansk started taking control over the regions. On April, the 7<sup>th</sup> the rebels occupied the local government building and proclaimed the establishment of the Donetsk People's Republic (Ray, 2016). In Lugansk, the protesters continued to oppose the legitimacy of the local authorities and called for a people's government. Fearing that the country will disintegrate, on the 15<sup>th</sup> of April Ukraine's acting President, Olexander Turchynov, announces the start of an "anti-terrorist operation" against pro-Russian separatists (BBC, 2014). On April, the 28<sup>th</sup> the rebels declared the establishment of the Lugansk People's Republic (Lenta, 2014). On the 11<sup>th</sup> of May referendums for independence were held both in Donetsk and Lugansk, where the majority of people voted to live Ukraine (Ray, 2016).

During the following months, the sporadic clashes between the Ukrainian forces and the separatists in the South-Eastern regions of Ukraine started escalating. After heavy military weapons were sent to the regions the crisis was transformed into a full-scale war, with the hardest fights, in Mariupol and the Donetsk airport (BBC, 2015).

On July, the 17<sup>th</sup> the international community was shocked after the Malaysia Airlines flight MH17 traveling from Amsterdam to Kuala Lumpur, was shot down in rebel-held territory (BBC, 2015). Everybody on board was killed (298 people), including 196 Dutch people (BBC, 2015).

## 4.1.4. Aftermath

The first attempt to establish a ceasefire in Donbass was on September the 5<sup>th</sup> 2014 (Walker S., 2014). The protocol (known as Minsk One) was signed by representatives from the Russian Federation, Ukraine, the Organization for Security and Co-operation in Europe, the Donetsk and Lugansk regions and was followed by a second document which clarified the contracting parties commitments (MZS, 2014), (OSCE, 2014). The truce didn't hold up and in the beginning of 2015 heavy fights was renewed (BBC, 2015). The second attempt to deescalate the conflict was made in February 2015, with the signing of the second protocol (OSCE, 2015). However, two years after

the truce, sporadic clashes still occur and the situation in Ukraine is being described as froze conflict (Tudoroiu, 2016, p. 378).

In Crimea, according to the to the "Report of the United Nations High Commissioner for Human Rights on the situation of human rights in Ukraine", from 21 November 2013 to 5 September 2014, human rights were violated however, there is no data for casualties during the annexation of the peninsula (OHCHR, 2014). In Donetsk and Lugansk, the picture is completely different. The UN "Report on the human rights situation in Ukraine 16 November 2016 to 15 February 2017" notes that from April 2014 till March 2017 approximately 9.940 people were killed (including approximately 2000 civilian casualties) and at least 23.455 were injured (OHCHR, 2017).

# 4.2. Exploitation of the unique conflict context

Ukraine is of high strategic importance to Russia for a number of reasons. First, from a geostrategic perspective, Ukraine acts as "as the last buffer" against the expansion of NATO (Waslander, 2014). Second, from an economic perspective, the country is the "gateway" for the Russian economic expansion towards Europe. Most importantly, the Russian gas and oil pipelines run through the country (Waslander, 2014). Third, (as it was pointed in section 4.1.2.) from a military perspective, the Crimean Peninsula is quite important for Russia, as a significant part of the Russian Black Sea Fleet is situated there (Bastion, 2017). Therefore, in February 2014, when the pro-West, anti-corruption Maidan protests escalated and the new, Western-oriented provisional government took the power in Kiev, Russia started taking decisive actions to keep Ukraine in its orbit of influence.

According to McCulloch's first principle of hybrid warfare, an actor that is initiating such a form of war will build one's forces composition and capabilities in relation to the unique context of the conflict. Which include the historical, socio-cultural, geographic and, temporal setting. In that sense, a number of circumstances enabled Russia to initiate hybrid actions against the Ukrainian state. As the conflict conditions are highly important for the entire analysis of Russia's actions in Ukraine, they will be exhaustively examined.

## 4.2.1. Historical setting

To a significant degree, Russia and Ukraine share their history. Their relation dates back to the 9th century AD, when the first Russian state - Kievan Rus was formed (Enciklopediq, 2017). This country existed until the Mongol-Tatar invasions of 1240 and its capital was Kiev (Alekseyevich, 2017). The following centuries the territory of today's Ukraine was under the rule of Galicia-Volhynia, a country successor of the Kievan Rus (Alekseyevich, 2017). In the end of the 16<sup>th</sup> century Ukraine fell under the rule of Poland, then in the middle of the 17<sup>th</sup> century, it became part of a Cossack state (Alekseyevich, 2017). In 1686, the country was divided in two (Alekseyevich, 2017). The territories west to the bank of the Dnieper River became Polish and the eastern territories, including Kiev, Russian (Alekseyevich, 2017). After the October Revolution of 1917, when the Russian Empire started crumbling, Ukraine began its struggle for independence (Alekseyevich, 2017). However, the fights were chaotic and between 1917 – 1918 several Ukrainian states came into existence (Alekseyevich, 2017). In 1921, the Ukrainian Soviet Socialist Republic emerged, yet it soon became part of Soviet Russia (Alekseyevich, 2017). After the fall of the Soviet Union in December 1991, Ukraine gained independence (Alekseyevich, 2017).

The history of the Crimean Peninsula is also connected to Russia. In the 10<sup>th</sup> century, it became part of Kievan Rus (Britannica, 2017). Yet the following centuries the territory fell under the rule of the Kipchaks, later the Tatars, and then in 1475, it became a part of the Ottoman Empire (Britannica, 2017). Since the 15<sup>th</sup> century, Russia and the Ottoman Empire fought a series of wars for the control of the territory (Britannica, 2017). In 1783 Catherine the Great annexed the peninsula (Britannica, 2017). During the Russian Civil War (1917–1921), as in Ukraine, several fractions were fighting for the control over the peninsula (Britannica, 2017). In the end of the war (1921), the Crimean Autonomous Soviet Socialist Republic was created and it became part of Soviet Russia (Britannica, 2017). In the end of the Second World War, the peninsula was "downgraded" to a province (oblast) of the Russian Soviet Federated Socialist Republic (Britannica, 2017). In 1954, the province was transferred to Ukraine (Britannica, 2017). After the dissolution of the Soviet Union, as the majority of the population were ethnic Russians, relations with Ukraine became tens (Britannica, 2017). The region had an autonomous status but, after struggles for independence, in 1994 it was abolished (Britannica, 2017).

From this brief historical overview, it is evident that Russia had strong presence in the territories of today's Ukraine, especially the lands east to the bank of the Dnieper River and the Crimean Peninsula. From the perspective of Russia's military and strategic planners, the fact that for more than a millennium Russians inhabited Ukraine is an important advantage, which has a number of implications. They are examined in the following sections.

## 4.2.2. Socio-cultural setting

The long standing historical connections with Russia, to a significant degree, have affected the socio-cultural setting in Ukraine. In his book "The Clash of Civilizations and the Remaking of World Order", Samuel Huntington divides the "post-1990s" world into nine "major civilizations" (Huntington, 1996, p. 26). He examines Russia and Ukraine as part of the Orthodox civilization. 13 However, he states that "Ukraine is divided between the Uniate nationalist Ukrainian-speaking west and the Orthodox Russian-speaking east" (Huntington, 1996, p. 138). As it was pointed in section 4.1.1., ethnic Russians comprise the biggest minority group in Ukraine, 17.3% of the total population (CIA, 2017). According to a 2001 census, in the nine South-Eastern regions of Ukraine, the people who use Russian as their first language, vary between 29.3% in Kherson oblast and 76.6% in Crimea (in Donetsk and Lugansk, the percentages are 74.9% and 68.8%). (Ukraina, 2001) (Europe, 2017). In the same regions, the percentage of people who perceive themselves as ethnic Russians is also high. It ranges between 14.1% in Kherson and Mykolaiv oblasts, and 58.3% in Crimea (Ukraina, 2001), (Europe, 2017). The ethnic Russians in Donetsk and Lugansk are 38.2% and 39%. This strong presence of Russian speakers and ethnic Russians is an important advantage for the Russian military and strategic planners. From a political perspective, they could use it as a source of justification and legitimization of their actions. As Vladimir Putin has stated, "The Russian Federation will always protect the out compatriots' abroad" (TASS, 2015). From a military perspective, the Russian minority (in some regions majority) in Ukraine can facilitate the operational effectiveness. This population can be used as a fifth column, which can provide manpower, intelligence, and shelter.

-

<sup>&</sup>lt;sup>13</sup> Including the countries from the former Soviet Union, the countries from the former Yugoslavia (except Croatia and Slovenia), Bulgaria, Cyprus, Greece and Romania (Huntington, 1996, pp. 26-27)

## 4.2.3. Geographic setting

The Geographic setting is another advantage that the Russian state has in terms of their relations with Ukraine. First, as noted in section 4.1.2., the two countries have signed a contract allowing Russia to station up to 25.000 troops, 132 armored combat vehicles and 24 pieces of artillery at its military facilities in Crimea (Kimball, 2014). As this is classified information, one can only guess what were the exact number of personnel and heavy weaponry that Russia had at its disposal by 2014 in Ukraine. However, it can be argued that the Russian military capabilities in Crime may have overmatched the whole Ukrainian fleet. Because by 2014, before the annexation of the peninsula, Ukraine had a total of 15.470 navy troops and 113 combat vehicles (Sukhov, 2015). Russia can exploit this advantage in a number of ways. Not only their forces could be used openly or as a deterrence measure but also, the naval bases in Crimea could serve as logistic a hub that will facilitate a variety of different types of operations (including covert).

The second advantage that Russia has in relation to the geographic setting in Ukraine is that the two countries share a 2063.04 km border (including 1974.04 km by land and 321 km by sea) (Ukraini, 2007). This is a key factor in terms of military planning because Russian operations in Ukraine can be facilitated by an unlimited amount of supplies which can be easily transferred through the common border.

## **4.2.4.** Temporal setting

By the time before the (Ukrainian) crisis had escalated (the end of 2013 and the beginning of 2014), the Ukrainian state was in a disadvantageous position. According to a 2013 research, the country was one of the least stable states in Europe (Nate Haken, 2013). This means that if a potential adversary wanted to intervene in Ukraine, the timing was right. Because the opponent would not have meet strong resistance. It can be argued that Ukraine's instability to a significant degree, was a result of the powerful influence (both political and economic) that Russia had in the country's domestic affairs (Simao, 2016, pp. 498-502). According to a study carried by Transparency International, in 2013 Ukraine had the most corrupt public sector in Europe (International, 2013). The elite of the country including the President (in that time Viktor Yanukovych), his political party (The Party of Regions), and most of the oligarchs had strong connections with Russia (Matuszak, 2012, pp. 50-71). Russia also had (and still has) a solid economic presence in Ukraine because they are the country's main trade partner both in terms of

import and export (Solutions, 2017). Therefore, when Ukraine started shifting towards the West, Russia had to act promptly, while they still had an advantageous position. Because hypothetically, the more they waited, the stronger Ukraine's ties with the West would have become. Meaning that the costs (including: economic, human, political, social, developmental and, strategical) for a Russia intervention would have risen in time.<sup>14</sup>

## 4.2.5. Analysis of the exploitation of the unique conflict context

From the previous sections, it can be seen that in the end of 2013 and the beginning of 2014 Russia had strong influence in Ukraine. However, when the people of Ukraine decided that they prefer closer relations with the EU (and the West in general), and started protesting in Kiev, Russia's interests were under threat. Until the middle of February, there was still hope that the pro-Russian President Viktor Yanukovych will restore order in the country. Yet, when the Maidan protests escalated, leaving more than fifty killed in a single day, it was evident that Yanukovych was losing the grip over the country (Gatehouse, 2015). When on the 23rd of February, the Rada declared that Yanukovych is no longer president and that the position is taken by the Westernorientated Oleksandr Turchynov, Russia had to act (Kyivpost, 2014), (BBC, 2014). But, a number of factors (which will be extensively examined in the following chapter) deterred the Federation to openly take actions against the will of the Ukrainian people and the new provisional government. Still, Russian military and strategic planners had a number of advantages, deriving from the historical, socio-cultural, geographic and, temporal settings which they were able to exploit, in order to keep Ukraine under their influence. One of the first actions that the new Ukrainian government made was a mistake, which provided Russia with a strong expedience. The Rada canceled the law "On State Language Policy" which gave an official status of the Russian, Bulgarian, Romanian, Tatarian and Hungarian languages (MVnR, 2014), (Tchuzakov, 2014). Now Russia could use its strongest leverage - the "threatened" Russian-speaking minority. The following weeks anti-Maidan protests broke out in all nine South-Eastern regions of Ukraine, which have big Russian minorities (RT, 2014), (Roth, 2014). This was the start of Russia's hybrid operation in the physical domain. It is hard to determine to what extent did the people participating in the protest decided to so on their own or they had other incentives. Yet, there are a number of

<sup>&</sup>lt;sup>14</sup> In November 2014, the new government in Ukraine announced that membership in NATO is one of its main goals (Europe, 2014).

reports stating that some of the protesters received money for their participation (Miller, 2014), (Ukrainskaq Pravda, 2014). In a number of cities, the pro-Russian protesters started occupying governmental buildings which was the second step in Russia's hybrid operation (Ray, Ukraine crisis, 2016). This served as a legitimization of the actions because it was presented as a bottom-up initiative (the will of the people). After Ukraine initiated the "anti-terrorist operation" against the pro-Russian separatists and violence started escalating in Donetsk and Lugansk, the locals formed voluntary combat forces, that fought the official Ukrainian Army (Telegraph, 2014).

Russia also exploited the advantage deriving from the common border with Ukraine and its military presence in Crimea in a number of ways. In Eastern Ukraine, before the conflict had erupted, the OSCE Human Rights Assessment Mission to Ukraine reported that in many of the pro-Russian protests, locals have observed the presence of unknown Russian people, to whom they refer as "titushky", which is defined as, "as mercenary support agents of various groups, often posing as street hooligans, and ready to use violence for money" (HCNM, 2014). When the conflict in Donbass escalated, Russia used its common border with Ukraine, to supply the separatist with weapons, which was proven by the General Court of the European Union in January 2017 (Union, 2017). The close proximity to Ukraine also enabled Russia to send military personnel to the conflict zone. Daniel Baer the Ambassador of the United States to the OSCE, stated that the observer mission has documented, "more than 30,000 individuals in military-style dress", crossing only at two border checkpoints (OSCE, 2016).

During the crisis, Russia had repeatedly denied any involvement in Ukraine. Including that the masked and armed men in Crimea (often referred to as "little green man" or "polite people"), carrying the latest Russian military equipment, that occupied the Simferopol International Airport, the Ukrainian military bases situated on the peninsula and other key governmental buildings, along with the Parliament in Simferopol, were actually Russian forces (Shevchenko, 2014). However, a month after the annexation of the peninsula, Vladimir Putin personally admitted that "Russian servicemen did back the Crimean self-defence forces" (Putin, 2014). Seven months later President Putin stated, that in fact Russian forces, occupied buildings in Crimea and blocked the Ukrainian forces situated there (Novosti, 2014). One could not argue from where did the "little green man" came to Crimea, yet the fact that Vladimir Putin himself had admitted that they were Russian forces, point in the direction of the Russian military bases, situated on the peninsula.

In terms of the temporal setting, it can be argued that Russia, took full advantage of it. In the end of February 2014, Ukraine was in a state that can be described as a revolution. More than fifty people were killed in the Maidan square, and the protests were continuing to escalate (Gatehouse, 2015). In that time, the Ukrainian Rada dismissed President Yanukovych and appointed the Parliament speaker Oleksandr Turchynov at his place (BBC, 2014). The new Western-orientated political circle in Kiev, had the difficult task to consolidate their positions and restore order in the country. Russia took advantage of the political instability and chaos in Ukraine and launched its hybrid campaign. The timing was right, and the anti-Maidan protests overwhelmed Ukraine's officials. The promptness of Russia's response allowed them not only to achieved most of their goals<sup>15</sup> in Ukraine but also, they accomplished this before Ukraine had receive substantial help its new allays. Therefore, the costs for Russia's campaign were less, because they did not waste time in initiating it.

Overall, it can be seen that the Russian military and strategic planners had taken full advantage of the complex context of the Ukrainian crisis.

# 4.3. Capability overmatch

According to McCulloch's fourth principle, the conventional opponent, for this particular analysis Ukraine, has more power in terms of military capabilities, therefore the hybrid actor - in Crimea the little green men and in Donbass the separatists, seeks ways to offset this disadvantage. In chapter three as a relevant indicator for this principle, was suggested the use of light weapons by the little green men and the separatists, for the neutralization of heavy weaponry systems of the official Ukrainian forces. Since the conflict in Ukraine developed in two distinct lines, the analysis will focus first on Crimea and then on Donbass. It has to be noted that, only the forces that were actually used will be examined. Thus, the Russian Black Sea Fleet situated in Crimea, is not included in the analysis.

<sup>-</sup>

<sup>&</sup>lt;sup>15</sup> The goal in Crimea was annexation. In Donbass, however it was not clear what the objective was. A probable explanation is that Russia wanted to prolong the conflict, which has a double effect. Both exhaust the Ukrainian economy and increasing their dependence on Russia, and also spark political instability in the country which also would serve Russia's interests.

## 4.3.1. Capability overmatch in Crimea

In Crimea, as presented in the previous sections, there were no open fights between the Ukrainian security forces and the so called the little green men. By the beginning of March 2014, all of Ukraine's military bases situated in the peninsula were besieged by the lightly armed masked men (Igor Gujava, 2014). Their arsenal included: 7.62 mm PKP machine guns, 5,45 mm AK74M assault rifles, RPG-30 anti-tank grenade launchers, GAZ Tiger multipurpose, all-terrain infantry mobility vehicles and military trucks Kamaz models 4326, 4350, 6350 (Pulkki, 2014), (Shuster, 2014), (Meduza, 2014). Data on what weapons did the Ukrainian soldiers situated in Crimea had is unavailable as this is classified information. However, there is information on some of the military bases the Ukrainian Army had in Crimea by 2014, which include: Belbek military airport, Baherove air base, Yevpratoria airbase, Bzanchoi airbase Southern Naval Base, Feodosia naval base, Sevastopol naval base, Balaklava coast guard base, Bakchisary military base, Perevalne military base, Simferopol military base (KyivPost, 2014), (Reuters, 2014), (Oliphant, 2014), (BBC, 2014). It is hard for one not to suggest that in eleven military bases where approximately 20.000 soldiers were situated, there were no major conventional arms (Loiko, 2014), (Baczynska, 2014).

The question why the Ukrainian Army did not fight back, the lightly armed men, remains unanswered. A potential explanation can be that the Russian Black Sea Fleet situated in Crimea, deterred the Ukrainian government to order an offensive (Bastion, 2017). However, in that time, (March 2014) Russia had not admitted any involvement in the crisis and the only forces that were actually using for the annexation were the anti-Maidan protesters and the little green men. Therefore, it can be reasoned that there was a capability overmatch between the pro-Russian protesters supported by the little green men and the official Ukrainian military personnel, situated in Crimea.

In the end of March 2014, the Government in Kiev announced that it is evacuating its soldiers from the peninsula (Loiko, 2014), (Baczynska, 2014).

## 4.3.2. Capability overmatch in Donbass

Compared to Crimea, the situation in Donetsk and Lugansk regions was more complex because the violence in both areas escalated to a point where the conflict could be described as a civil war. In the beginning of the crisis, the separatists were armed with light Soviet and Russian weapons. When Ukraine announces on the 15<sup>th</sup> of April, the start of an "anti-terrorist operation" against the pro-Russian separatists (BBC, 2014). The regular Ukrainian Army was deployed to

retake control in the cities occupied by the separatist. When on the following day (16<sup>th</sup> of April) two Ukrainian military convoys were sent to the cities of Kramatorsk and Slavyansk, hundreds of locals surrounded them and captured their armored vehicles (Grove, 2014). According to a Russian organization, which monitors the destroyed and captured armored fighting vehicles during the conflict, between April and November 2014, the separatists captured more than 225 armored vehicles (Armor, 2016). A report developed by the private intelligence consultancy company Armament Research Services, shows what weapons did the separatist had in their possession by November 2014. The light weapons included: self-made small arms, pistols (Makarov, Stechkin), shotguns (Saiga 12, Armscor Model 30, Mosin M9), machineguns (AKS-74, PM, RGD-5, RPG-26, RPG-26, AKMS, RPG-26, AK-74, AKS), anti-materiel rifle (PTRS-41 anti-tank rifle, PTRD, ASVK anti-materiel rifle), heavy machineguns (DShKM, NSV, NSVT, KPV, KPVT), grenade launchers (GP-25, 7P17 VOG-25, 7P24 VOG-25P, AGS-17 AGL), portable antitank systems (PG-7, PG-7L, PG-7M, PG-7S, OG-7), mortars (PG-9 type HEAT rockets, OG-9 HE-FRAG, O-832D HE-FRAG, OF-842 HE-FRAG and S-842), man-portable air defence systems and anti-aircraft guns (9K38 Igla, e SA-18 Grouse, 9K38 system, 9K32 Strela-2, ZPU-1, ZPU-2, ZPU-4) (Jones, 2014, pp. 18-63). It can be seen that the separatists were well equipped in terms of light weapons. However, their origin is unknown. In an interview in August 2014, when asked if Russia is supplying the separatist with arms, the Russian deputy Defense Minister Anatoli Antonov replied that "they have taken it from Ukrainian military storages or captured it from the Ukrainian security forces (Nabrejnov, 2014). Yet, as presented in section 4.2.5., in January 2017 the General Court of the European Union proved that Russia was sending arms to the separatist in Ukraine (Union, 2017).

In terms of heavy weaponry, the Armament Research Services report, presents a comparison between the major conventional arms of Ukrainian government forces and the separatist's forces. Yet, the report does not give information on the numbers of the weapons used by both sides (Jones, 2014, p. 65).

\_

<sup>&</sup>lt;sup>16</sup> Arms used against military equipment, instead against combatants

Ukrainian government forces	Pro-Russian separatist forces	
Main battle tanks		
T-64A	IS-3	
T-64B	T-34-85	
T-64B1M	T-64A	
T-64BM	T-64B	
T-64BV	T-64BM	
	T-64BV	
	T-72B	
	T-72B3	
	T-72BA	
	T-72B Mod	

Table 1 Ukrainian conflict main battle tanks, source: (Jones, 2014, p. 65)

Pro-Russian separatist forces			
Self-propelled artillery			
2S1			
2S19			
2S3			
2S5			
2S9			
2S6			
9K33			

	9K35	9K35
		9K373
		Pantsir-S13
Self-propelled rocket artillery	9K51	9K51
	9K57	
	9K58	
	9К79	

Table 2 Ukrainian conflict self-propelled artillery, source: (Jones, 2014, p. 70)

From the tables, it is evident that half a year after the conflict had started there was no capability overmatch between the Ukrainian government forces and the separatist forces in terms of the types of weapons they were using.<sup>17</sup> The only advantage that the Ukrainian Army had was its air forces. Yet the separatist had anti-aircraft systems and managed to shoot down five battle helicopters and ten military airplanes (Jones, 2014, p. 77).

<sup>&</sup>lt;sup>17</sup> As already was noted there is no information on the numbers of heavy weapons used by both sides. According to the Russia monitoring organization Lost Armour, by February 2015, Ukraine had lost 827 pieces of armored vehicles, compared to 185 lost by the separatists. However, this data does not suggest how many major conventional arms, did both sides possessed during the given period (until February 2015).

# 4.4. Combination of conventional and unconventional elements

McCulloh's fifth principle of hybrid warfare, states that a hybrid actor needs to have both conventional and unconventional elements which create an asymmetrical advantage when fighting a conventionally-based opponent.

In a 2006 speech, President Vladimir Putin, addressed the problems in the Russian military, arguing that it needs a reform (which was initiated in 2008 see section1.4.), he stated: "Our responses should be based on intellectual superiority. They will be asymmetric [and] less costly" (Buckley, 2006). The Ukrainian crisis showed that these ideas had been implemented in Russia's military strategy. The most apparent unconventional element during of the Ukrainian crisis were the irregular fighters. The Russian military and strategic planners exploited the advantages diving from the unique conflict setting described in section 4.2. and used such forces in two distinct ways.

In Crimea, the conventional component in Russia's operation were the Russian Navy troops, situated on the Peninsula. Although they did not take any actions, these militaries were used as a deterrent measure. The most visible unconventional element used by Russia were the unmarked, masked and well equipped (with light weaponry), troopers deployed in Crimea. Some reports suggest that they were Spetsnaz (Russia's Special Forces) (Haines, 2014). Their main task was to safeguard the annexation of the peninsula, or as Vladimir Putin describes it: "to ensure the open, honest and decent way for people to express their opinion" (RT, 2014). They operated along with the well-organized and often armed demonstrators, which also can be examined as an unconventional force (Sindelar, 2014). The actions of the little green men and the demonstrators were well synchronized and they reinforce each other. The masked armed men provided protection, while the demonstrators were used as a "justification tool" that allowed the operation to be framed as an act of self-determination. Their swift actions caught by surprise Ukraine's officials. They fell into a situation where if they had ordered the Ukrainian militaries, station in the peninsula to engage in fights with the little green men and the protesters, not only they would have lost their international support (most notably the EU and US), because they would have to fight civilians (as Yanukovych's regime did in Kiev), but also this would have allowed Russia to openly engage in the crisis, and order its troops (stationed in Crimea) to actively protect the Russian population. In Crimea, the Russian officials were always one step ahead. The forces that Russia coordinated and used enabled them, to promptly execute their main objective – the annexation of the peninsula, at very low costs, without taking any casualties. Such effective and efficient results cannot be achieved by the use of a conventional army.

In Donbass, until the middle of April 2014, the events unfolded in a similar manner (BBC, 2014). The pro-Russian protests escalated and governmental buildings were occupied. However, the demonstrators were not supported by armed Russian Special Forces as in Crimea. Civil units erected picket lines on bridges and key roads in order to prevent Ukrainian forces from coming to the regions of Donetsk and Lugansk (Dubovaq, 2014). After the Crimean Peninsula was annexed, in March 2014, Ukraine had ground to justify actions against the civil units in Donbass. However, when on the 16<sup>th</sup> of April 2014 (one day after the Ukrainian government had announced the start of start of an "anti-terrorist operation" against the pro-Russian separatists in the Eastern regions of the country) the first Ukrainian military convoys were sent to the cities of Kramatorsk and Slavyansk, unarmed local residents captured their armored vehicles (Grove, 2014), (BBC, 2014). As reported by Reuters "A Ukrainian officer said his men were not prepared to fire on fellow Ukrainians" (Grove, 2014). Until this point, the Russian presence was not as visible as in Crimea, although there were reports of "titushky", participating in the civil units (Minorities, 2014). Yet, the following weeks violence escalated and "voluntary groups" or in other words pro-Russian insurgents, started appearing in the eastern regions of Ukraine. The main combat groups included: Donbass People's Militia (commanded by Igor Girkin, a Russian citizen, among their ranks there were a number of veterans from Afghanistan), Army of the South-East (which included defectors from the disbanded Ukrainian Special Police Forces – Berkut), Russian Orthodox Army (commanded by Igor Strelkov, a Russian citizen and a veteran, later became the Minister of Defence of the Donetsk People's Republic), Vostok Battalion (commanded by Alexander Khodakovsky the former commander of the Alpha Anti-terrorism Special Unit of the Security Service of Ukraine) (Tsvetkova, 2014), (Arwa Damon, 2014), (Luhn, 2014), (Pravda, 2014), (Ukraina G. P., 2014), (Zverev, 2014). In September 2014, these fractions combined and formed the "United Armed Forces of Novorossiya" (Post, 2014).

All of the described above combat groups and the pro-Russian protestors can be regarded as irregular forces or guerrilla forces.<sup>18</sup> Also, according to the Ukrainian government they can be classified as terrorists (BBC, 2014). It is hard to determine to what extent did they operate

\_

<sup>&</sup>lt;sup>18</sup> guerrilla force definition – "A group of irregular, predominantly indigenous personnel organized along military lines to conduct military and paramilitary operations in enemy-held, hostile, or denied territory" (Gortney, 2010, p. 100)

autonomously or they were controlled by Russia. However, there are a number of evidence that show clear connections with Russia. They include: OSCE reports, that more than 30.000 armed men have entered Ukraine from Russia, the conclusion of the General Court of the European Union, that Russia had supplied the insurgents with military equipment, the fact that two of the key pro-Russian combat groups were commanded by Russian citizens, and most importantly the fact that in December 2015, Vladimir Putin admitted that Russia had sent military intelligence officers to help the separatists in Eastern Ukraine (OSCE, 2016), (Union, 2017), (Walker S. , 2015), (Ukraina G. P., 2014), (Zverev, 2014). Therefore, the protesters and the pro-Russian paramilitary groups in Donbass can be examined as the unconventional element in Russia's strategy.

During the first months of fights (between April and September 2014), Russia also used a conventional component in its Donbass strategy - military exercises. Two of which (in April and August) along its' border with Ukraine (MacAskill, 2014), (Luhn, 2014). Exercises were also carried in June, in Kaliningrad (Russia's Baltic enclave) and in September, in the Russian Far East, which was "the biggest-ever post-Soviet drills" (Williams, 2014), (RT, 2014). This demonstration of military superiority was used as a deterrent measure. As it was pointed in the beginning of this chapter Vladimir Putin had declared that if he wants, he can take Kiev in two weeks (Squires, 2014).

Overall, both in Crimea and in Donbass, Russia used a combination of both conventional and unconventional elements. In Crimea, the objective was fulfilled. In Donbass however, it is not clear what the main goal was. A potential explanation is that Russia wanted to prolong the conflict, and exhaust the Ukrainian economy thus, increasing their dependence on Russia (the main trade partner of Ukraine both in import and export is Russia) or that Russia could not cope with the international pressure, yet this are only hypothesis (Solutions, 2017).

# 4.5. Defensive type operations

According to McCulloh's sixth principle, an actor that is waging a hybrid war will try to hold one's ground by developing a defensive strategy and relying mostly on defensive operations. Since in Crimea there were no direct battles between the Russian Spetsnaz (the so-called little green men) and the Ukrainian Army, the focus will be primarily on the events in Eastern Ukraine. However, it has to be noted that Russia's operations for the annexation of the peninsula, were

offensive because they involved an armed besiegement of official military forces which is a clear violation of Ukraine's sovereignty.<sup>19</sup>

In Eastern Ukraine, by the beginning of April 2014, the pro-Russian rebels had taken control over most of the Donetsk and Lugansk regions (BBC, 2015). As already noted, their strategy included the occupation of key governmental buildings and erection of picket lines (Dubovaq, 2014). In this regard from the rebel's perspective, their actions during the first weeks of the conflict were not offensive, because they did not try to push further into Ukrainian territory and they only wanted to exercise their right of self-determination.<sup>20</sup> Their goal was to establish and hold their control over the two regions.

In the middle of April (2014), after the Ukrainian military was sent to retake control over the two regions, and the first military convoys were captured by civilians (see section 4.3.1.), three key theaters of action were opened (Grove, 2014), (BBC, 2014). The first fights broke out near the city of Slavyansk, which is situated in the North-Western part of the Donetsk region (Whewell, 2014). By May 2014 two more battle lines were opened – in the city of Mariupol, which is situated in the South of the Donetsk region on the Azov Sea, and in the Donetsk Airport, which is in the Western outskirts of the city of Donetsk (Walker P., 2014), (Tavernise, 2014). In the end of May and the beginning of July, governmental forces went on the offensive, and recaptured the cities of Slavyansk, Kramatorsk and Mariupol and the Donetsk Airport (all in the borders of previously rebel-held territory) (Stern, 2014), (Stern, 2014). By that time, the pro-Russian separatist had started withdrawing to the center of the territory they claimed. It can be argued that this was a defensive type strategy because they concentrated most of their capabilities in the capitals of the two regions – Donetsk and Lugansk and their main goal was to hold these cities. By the end of July and the beginning of August, the Ukrainian forces had besieged the two cities and were launching heavy artillery attacks on them (BBC, 2014), (BBC, 2014). The separatists were almost cut from the Eastern border with Russia and had lost a number of their strongholds in the North (near the Azov Sea) and in the West of Donetsk and Lugansk (Herszenhorn, 2014). However, in the end of

<sup>&</sup>lt;sup>19</sup> From the Russian perspective, the campaign for the annexation of Crime was a defensive operation because the Russian special forces intervened in order to protect the threatened Russian-speaking population. However, the approach that they used, compared to the one in Donbass was offensive because it included lightly armed, but well-trained soldiers, while in the Donetsk and Lugansk regions, the initial phase of the operation was carried by civilians, the majority of which were unarmed (Kushch, 2014).

<sup>&</sup>lt;sup>20</sup> However, from the Ukrainian perspective, these acts were clear violation of their sovereignty.

August, there were reports that a column of armored Russian vehicles has crossed into Ukraine from Russia (Hewitt, 2014). Soon after, the pro-Russian forces initiated their first major attack operation. The counter-offensive was relatively successful and the separatists managed to regain control over some of the territories that they had lost in the East and South (Farquhar, 2014). On the 5<sup>th</sup> of September, under international mediation, a ceasefire was established (Walker S., 2014) which was followed by a memorandum signed on the 19th of September which outlined the geographical parameters of the truce and the establishment of a buffer zone (OSCE, 2014). The following months the Minsk Protocol was violated a number of times and low-intensity fights along the buffer zone were continuing (OHCHR, 2014). In January 2015, the separatist launched a second offensive (OSCE, 2015). On the 20th of January the Ukrainian President Petro Poroshenko, accused Russia of sending "more than 9.000 troops and 500 tanks, heavy artillery and armored personnel carriers in eastern Ukraine" (BBC, 2015). By the end of the month, the separatists had recaptured the Donetsk Airport and had started fights along the buffer zone in the North, West, and South (Oliphant, 2015). On the 12th of February, a second Minsk Protocol was signed between representatives from Ukraine, Russia, OSCE, Donetsk People's Republic and Lugansk People's Republic (OSCE, 2015). The key points of the document included: a full ceasefire by the 15th of February 2015, withdraw of heavy weapons systems from the conflict zone, reestablishment of the buffer zone (set in the Memorandum of 19th of September 2014), constitutional reform in Ukraine, and exchange of all hostages and prisoners (OSCE, 2015). Yet, heavy fights continued in the strategic city Debaltseve, which the Ukrainian Army fled on the 18th of February 2015 (Grytsenko, 2015). More than two years after the second Minsk Protocol was signed, the truce is still holding although, the OSCE Special Monitoring Mission to Ukraine is reporting violations almost every day (OSCE, 2017).

Overall, the strategy used by Russia during the annexation of the Crimea Peninsula was offensive, while the actions of the Russian-backed separatist in Eastern Ukraine can be described as defensive. In the East, the separatist's main goal was to hold the capitals of the two regions Donetsk and Lugansk and gain legitimacy over as much territory they as they can (within the original borders of both regions). Although they did not manage to hold the lands which they claimed on the 11<sup>th</sup> of May 2014 (the date of their self-proclaimed independence) the separatists initiated two major offenses which were reportedly backed by the Russian military (BBC, 2014). According to McCulloh, the hybrid actor's strategy and operations "will often include offensive

components, but the overarching intent is still one of defense" (Johnson, 2013, p. 17). Both of attacks were not prolonged and were followed by peace talks and the signing of peace protocols. Therefore, the pro-Russian separatists' operations in Eastern Ukraine fit the description of McCulloh's sixth principle, while the Russian actions in Crimea, does not.

# 4.6. Attritional tactics

McCulloh's seventh principle holds that a hybrid actor will use attritional tactics both in the physical and cognitive domains in order to undermine the morale of the conventional opponent and to diminish their capabilities. As this chapter explores the first unit of analysis of the case study, the focus will be only on the physical domains of operations. Attritional tactics in the cognitive domain will be explored in the following chapter.

In the first phase of the Ukrainian crisis (before April 2014), the key attritional tactic that the Russian forces in Crimea and the pro-Russian forces in Eastern Ukraine used was the occupation of governmental buildings. The participation of civilians during these operations, allowed Russia to frame these acts as a bottom-up initiative which corresponds to the principle of self-determination (RT, 2014). It can be argued that the Russian military and strategic planners deliberately used this approach because it did not involve escalation of violence which allowed the pro-Russian forces to consolidate their positions. If the anti-Maidan protesters (in Crimea supported by the little green men), had started heavy fights with Ukraine's security forces, this would have allowed the Ukrainian government to order the anti-terrorist operation, not in the middle of April 2014, but in the end of February (BBC, 2014). If this had happened, the crisis would have developed in a completely different manner and probably Russia would have involved openly. This, on the other hand, would not have served Russia's interest because the costs of a conventional intervention would have been much higher, compared to the unconventional one that they actually undertook (see section 5.3.3). Therefore, the occupation tactic can be seen as attritional, because on one hand, it allowed Russia to prolong the crisis and strengthen its irregular forces in Ukraine, and on the other it did not allow Ukraine to intervene because they would have lost the support of the US and the EU (see section 4.4). This also had a negative effect on the morale of the Ukrainian soldiers because they felt that they were let down by the new Ukrainian government (this matter will be examined in the following chapter). In Crimea, this approach was successful and in the end of March 2014, the besieged (in their military bases) approximately 20.000 Ukrainian soldiers were ordered by their government to evacuate from the peninsula (Loiko, 2014), (Baczynska, 2014). In Donbass, the "occupation tactic" was effective until the first weeks after the Ukrainian government had announced the start of the anti-terrorist operation against the pro-Russian separatists (BBC, 2014).<sup>21</sup> By the time the first fights broke out, the pro-Russian separatist had managed to capture military equipment from the Ukrainian Army and had fortified their positions (Grove, 2014), (Whewell, 2014).

The second attritional tactic used by the separatists became apparent after violence escalated and the Ukrainian forces launched their first major attack. Instead of holding their ground, the separatist gradually retreated to their strongholds in the cities of Donetsk and Lugansk, thus averting a major battle which could have decided the outcome of the conflict (BBC, 2014).

The two main offenses that the pro-Russian insurgents initiated in August 2014 and in January 2015 exposed the third attritional tactic (Hewitt, 2014), (Telegraph, 2015). In both attacks, a specific pattern can be found. In August and January, the separatists reportedly received military support from Russia (Hewitt, 2014), (BBC, 2015). This caught by surprise the Ukrainian forces and the separatists quickly gained control over the battlefield. In August 2014, approximately 1.400 Ukrainian soldiers were encircled in the city of Ilovaisk (Peleschuk, 2014). During the second attack in January 2015, a similar scenario unfolded in Debaltseve, where nearly 6.000 Ukrainian troops were besieged (Vlasova, 2015). The swift successes of the separatist allowed them to enter into peace negotiations. Even though the Minsk Protocols were tripartite contractual agreements (between official representatives from OSCE, Ukraine, and Russia) both documents were also signed by representatives from Donetsk and Lugansk Peoples Republics (MZS, 2014), (OSCE, 2015). To a significant degree, the Minsk Protocols enabled the pro-Russian insurgents to consolidate their positions in Donetsk and Lugansk regions. The first Minsk Protocol envisaged only a law that Ukraine needs to adopt in order to ensure "Special Status" of the two regions, while the second protocol obliged Ukraine to carry out a constitutional reform<sup>22</sup> which will ensure political decentralization and allow more autonomy to Donetsk and Lugansk (see point three of the

<sup>&</sup>lt;sup>21</sup> The annexation of Crime, allowed the Ukrainian government to justify the operation

<sup>&</sup>lt;sup>22</sup> The constitutional reform remains blocked in the Ukrainian Parliament (Carnegie, 2017)

first Minsk Protocol and points four and eleven of the second Minsk Protocol) (MZS, 2014), (OSCE, 2015). Therefore, it can be argued that to some extent these documents legitimized the self–governing structures of Donetsk and Lugansk Peoples Republics. The other important factor which derives from the Minsk agreements is that both documents obligated the adversaries to immediately implement a fill ceasefire (see points one of both Minsk Protocols) (MZS, 2014), (OSCE, 2015). Although the first truce lasted only four months before it was broken, the second one is still holding. As a result, the conflict in Ukraine is at a stalemate situation.

To summarize, the use of attritional tactics is one of the key components in Russia's military approach exposed during the Ukrainian crisis. In Crimea, the "occupation tactic" was effective and allowed Russia to annex the peninsula in less than a month (OHCHR, 2014). In Donbass, the operational environment was significantly different and more complex. This forced the pro-Russian separatist to use additional attritional tactics that reinforce each other. The most effective however, were the two swift attacks launched by the separatists in August 2014 and January 2015 (Telegraph, 2015). To a significant degree the peace talks that followed, especially the second Minsk agreement, enabled the pro-Russian insurgents to secure their positions in Donetsk and Lugansk regions. The deadlock situation in Eastern Ukraine serves the Russia's interests because as Jokull Johannesson argues, "Russia is to continue its hybrid warfare with the objective to slowly drain Ukraine of resources and the will to fight resulting in the collapse of the pro-western governments" (Johannesson, 2017, p. 70).

# **4.7 Conclusions**

In this chapter, the first unit of analysis of the case study was explored. Russia's acts in the physical domains of operations during the Ukrainian crisis were analyzed through Timothy McCulloch's hybrid warfare theoretical framework. From the study, it is evident that Russia used different strategies and tactics in Crimea and Donbass. Although on two occasions discrepancies between McCulloch's principles and the Russian actions were found, the overall military approach exposed during the crisis, to a significant degree fits the characteristics of hybrid warfare. A more detailed account of the findings will be presented in the final chapter.

# 5. Expressions of hybrid warfare in the non-physical domain of operations during the Ukrainian crisis

"If you know the enemy and know yourself, you need not fear the result of a hundred battles"

Sun Tzu 5th century BC (Tzu, 1910)

In this chapter, the second unit of analysis of the case study will be examined. McCulloch's theoretical framework will be applied to the Ukrainian case. The focus of the exploration will be on Russia's actions in the non-physical domain of operations. This would allow to suggest an answer to the fourth sub research question: "Which expressions of hybrid warfare in the non-physical domain of operations can be identified in the Ukrainian crisis?".

However, before the analysis begins an important remark needs to be made. There is a substantial body of literature that examines Russia's information warfare approach see: (Snegovaya, 2015), (Vladimir Sazonov, 2016), (Darczewska, 2014), (Thomas, 2004), (Pursiainen, 2017). Yet, the main objective of this work is to explore what extent can the Russian actions in the Ukrainian crisis be conceptualized as a hybrid war according to a specific theoretical framework. Therefore, some aspects of Russia's information warfare, may not be examined.

# **5.1.** Exploitation of the unique conflict context

According to McCulloch's first principle of hybrid warfare, a hybrid actor will exploit all advantages deriving from the unique context of the conflict (the historical, socio-cultural, geographic and, temporal setting) and build one's forces composition and capabilities in relation to it. In the previous chapter, the conflict setting was extensively examined and it was proven that in the physical domains of operations Russia took full advantage of it (see sections 4.2. - 4.2.5.). One of the most important factors for the success of Russia's operations in Crimea and Donbass

resulted from the favoring socio-cultural setting.<sup>23</sup> The Russian-speaking minority living in these regions of Ukraine was mobilized and used in a number of different ways (see chapter four). However, the process of how these people were influenced and utilized to serve Russia's interests was not presented. This section will help to elucidate this matter.

One of the key characteristics of the Ukrainian crisis, especially during the period between its inception and the signing of the second Minsk agreement, were the information operations carried by both sides. Russian military and strategic planners examine such operations as one of the main components of modern-day conflicts. According to the Russian military doctrine from December 2014:

An early conduction of information confrontation activities will allow the achievement of political goals without the use of military force, subsequently, this will facilitate the formation of a favorable response of the world community for the use of military force [if needed] (Rossii, 2014).

The Chief of the Russian General Staff General Valery Gerasimov, argues that "The focus of applied methods of conflict has altered in the direction of the broad use of political, economic, informational, humanitarian, and other nonmilitary measures – applied in coordination with the protest potential of the population" (Gerasimov, 2013). One of the main tools that Russia used in its information campaigns both in Crimea and Donbass was its the strongly centralized media (Seely, 2015, p. 12). The main target was the Russian-speaking minority.

In 2014, the global consulting company Gallup reported that television (TV) is the prime source for news information among 96.8% of Ukrainians (Gallup, 2014). As presented in section 4.2.4., Russia has a strong presence in Ukraine's economy. This includes the broadcasting industry. According to a 2014 article by Joanna Szostek, a significant portion of Ukraine's media market is owned by Russian companies. The top news providers which had Russian stakeholders or partners include: Pervyy Kanal, Komsomolskaya Pravda v Ukraine, Izvestiya v Ukraine, Argumenty i Fakty v Ukraine, Kommersant-Ukraina (Szostek, 2014, p. 468). Also, by the time of the crisis, Russian

60

<sup>&</sup>lt;sup>23</sup> It has to be noted that all conflict characteristics (historical, socio-cultural, geographic and, temporal setting) are interconnected and affect each other. Yet, the historical and socio-cultural settings were the characteristics that Russia exploited the most, in relation to the cognitive domains of operations. The historical setting was used in a manner, that is connected with McCulloch's second principle. Therefore, prime focus in this section will be on the exploitation of the socio-cultural setting.

TV channels were more popular than Ukrainian among the Russian-speakers in Ukraine (Bērziņš, 2015, p. 7).

When the crisis in Ukraine escalated in February 2014, Russia used its media to launch a massive information attack (BBC, 2014). In his book "Hybrid Wars: The Indirect Adaptive Approach to Regime Change" Andrew Korybko presents a guideline for indirect regime change operations which he describes as "Color Revolutions" (Korybko, 2015). According to him, one of the most important factors for the success of such operations is the creation of a "hive mind" of anti-government activists (Korybko, 2015, pp. 43-45). He states that:

Effective information outreach campaigns and the construction of social networks over a period of time can lead to the development of a hive mind of anti-government activists. These individuals can then be guided [...] into strategic swarms that work to overwhelm the authorities and initiate a soft coup (Korybko, 2015, p. 76).

It can be argued that the key goal of Russia's information campaign was to consolidate the Russianspeaking minority, trigger an emotional response and set these people against the Ukrainian government.

The Russian strategy in the cognitive domain was centered around the formation of fear among the Russian-speakers in Ukraine. The overall approach was multi-aspect and complex, yet two main lines of action can be identified. The first one is connected to the pro-European Maidan protest movement. The Russian media depicted the Maidan activist as "radicals" (Ishchenko, 2016, p. 466). The dominant focus of Russian news reports was on confrontation and violence during the protests (Ishchenko, 2016, p. 460). Even though among the Maidan protesters there was a variety of different groups, including "Kiev's intellectual elite and students", the most frequently mentioned organization were the far-right nationalists from Right Sector (Strovsky, 2015, p. 104), (Ishchenko, 2016, p. 460). Thus, Russia's media not only managed to "combine these people into one mass" but also, they succeeded in framing them as violent fascists (Strovsky, 2015, p. 104). When on the 22<sup>nd</sup> of February 2014 the Maidan protesters seized Ukraine president's office, the Ukrainian Parliament voted to dismiss the pro-Russian President Viktor Yanukovych, and a new provisional government took the power, fears of repressions amongst ethnic Russians in Ukraine

<sup>&</sup>lt;sup>24</sup> Korybko describes it as a "simplifying the mass thinking of the society during the onset of the Color Revolution coup attempt" (Korybko, 2015, p. 34)

rose (Kyivpost, 2014), (Coalson, 2014). These concerns were justified on the 23<sup>rd</sup> of February, when the Rada canceled the law "On State Language Policy" which gave an official status of the Russian, Bulgarian, Romanian, Tatarian and Hungarian languages (MVnR, 2014), (Tchuzakov, 2014). A turning point was reached. The improvident governmental act divided Ukraine's population. Russian-speaking groups in the South-Eastern regions of Ukraine united and anti-Maidan protests intensified (RT, 2014), (Roth, 2014).

The second line of action carried by the Russian state was to deepen the fears of the Russian speakers in Ukraine and impel them towards actions against the Ukrainian government. This was achieved again by framing. The official Russian position was that the new government in Kiev is illegitimate because the overthrow of President Yanukovych was unconstitutional and it had violated the "Agreement on the Settlement of Crisis in Ukraine" sighed in February 21st, 2014 (Guardian, 2014), (Kelley, 2014). In Russian media, this matter was taken a step further and the new Ukrainian government was often referred to as "junta" or "fascist junta" (Vladimir Sazonov, 2016, p. 69), (BBC, 2014). By the end of February 2014, Russia had managed to create a "hive mind" of anti-government and pro-Russian activists. What followed was the occupation of governmental buildings in the South-Eastern regions of Ukraine. In Crimea, (as presented in chapter four) the "occupation tactic", which was backed by the Russian special forces, was successful and the result, according to Andrew Korybko's classification, was a soft coup (Korybko, 2015, p. 76). In Donbass however, the situation was different and the "occupation tactic" was not enough, to separate the Donetsk and Lugansk regions from Ukraine. There the pro-Russian protesters took arms, formed voluntary combat forces and started fighting the official Ukrainian Army (Telegraph, 2014). Korybko notes that:

Should the Color Revolution [soft coup] fail in overthrowing the government, then the transition to Unconventional Warfare occurs, whereby the social infrastructure of the Color Revolution becomes the foundation of the violent campaign being waged by the antigovernment movement. It is at this point that the hard coup commences and all elements of the state are thrown into strategically engineered chaos (Korybko, 2015, p. 76).

What happened in Donbass according to Korybko's classification was a "hard coup".

Overall, Russia exploited the advantages deriving from the unique context of the conflict. They built their forces composition and capabilities in relation to it. In this section, the process of

how Russia used the favorable socio-cultural setting in Ukraine was presented. Russia utilized its centralized media and initiated a massive information attack, targeting the Russia-speaking population in Ukraine. This cognitive attack was successful in the creation of an environment of fear and discontent amongst the Russian minority in Ukraine. Once the hive mind was formed, these people were manipulated into serving Russia's own interest. It is important to note that a similar cognitive approach was described in the Russian military doctrine from December 2014. Also, Russia's information operation in Ukraine, to a significant degree, followed the guideline for indirect regime change operations created by the Russian scholar Andrew Korybko.

# 5.2. Ideology of the hybrid actor

In his second principle of hybrid warfare, McCulloch argues that a hybrid actor will develop a specific ideology which supports and justify one's actions. This ideology will be linked to the unique context of the conflict and create an internal narrative of the hybrid organization. The analysis will be focused on Russia's operation in Donbass because the one in Crimea lasted less than a month, and there was not enough time for the formation of a specific ideology there.

The Encyclopaedia Britannica defines ideology as: "a form of social or political philosophy in which practical elements are as prominent as theoretical ones. It is a system of ideas that aspires both to explain the world and to change it" (Cranston, 2014). In that sense, it can be argued that the leaders of separatists in Donbass (which received support from Russian intelligence officers, see section 4.4.), along with Russian officials and the Russian media used a specific discourse, that created a set of normative beliefs (ideology), amongst the insurgents. This ideology was centralized around two ideas. First, that the pro-Russian separatists are "heroic defenders of the motherland" and second, that they are the indigenous population within the lands they claimed. In this section, four terms will be extensively examined. The analysis of the discourse will show how the separatists identify themselves and how they perceive Ukrainian officials.

The separatist ideology was created through the use of historical analogies and myths. Andrew Wilson argues that elites (the potential leaders of ethno-nationalist movements and their ideologues) "often find that potential members of their target group are politically passive, isolated from one another, and/or more interested in private goals, and therefore difficult to mobilize politically" (Wilson, 1995, p. 265). He continues by stating that historical myths are "an extremely effective means of firming up a target group's collective identity, encouraging group coalescence, and stimulating political mobilization" (Wilson, 1995, p. 265).

The first term that is going to be examined is the Russian word "opolcheniye". During the first year of the crisis, this word was widely used not only by the separatists but also in Russia's public discourse, for the portrayal of the pro-Russian forces fighting in Donbass (Krechetnikov, 2014). In English, opolcheniye can be translated as militia, or local defense volunteers. In Russian however, opolcheniye has a strong historical symbolism and it is much more than militia. According to the "Great Russian Encyclopedia" the term "opolcheniye" means an armed (military) formation, composed of citizens (mostly volunteers) that are generally not subject to conscription or mobilization; one of the forms of participation of the masses in the armed defense of the state (Verhovod, 2014). It is also described as a method and form of organization of an armed resistance against an external enemy (Verhovod, 2014). The first opolcheniye groups can be traced back to ancient times, and according to the Encyclopedia, different forms of opolcheniye can be found in the history of many Western European countries. However, "in Russia, opolcheniye had a much greater role than in the West countries" (Verhovod, 2014). In the 9-11 centuries opolcheniye formations stave off attacks of the hordes of nomads (Khazars, Pechenegs, Polovtsy), in 12-15 centuries the invasion of Mongol-Tatars, and the German Knights, opolcheniye was also formed during "The Great Patriotic War" - World War Two (WW2) (Verhovod, 2014). Opolcheniye describes a civilian participation in the defense of the country against the regular armies of external enemies. Therefore, in Russian, when a group of people are being referred to as opolcheniye this means that they are civilians that have united in order not to attack, but to defend their land against the regular forces of an external enemy. The word was also used in the official Russian public discourse, which reinforced the political mobilization effect amongst Russian-speakers in Donbass (by The President Vladimir Putin, in The State Duma (the Lower House of Russia's Parliament) and by Russia's international news agency RIA Novosti) (Novosti R., 2014), (Putin, 2014), (Duma, 2015). This was a deliberate choice because in Russian, there are other words that could have been used. For example, militia (in Russian-militciya), rebels (in Russian-povstanci), or separatists, which do not the same connotation as opolchencii" (a person part of the opolcheniye). Yet they were not used, because these words do not trigger an emotional response in the public. Therefore,

in Russia's both public and official discourse, the pro-Russian separatists in Donbass were illustrated as "heroic defenders of the motherland".

In May 2014 Donetsk and Lugansk Peoples Republics united and created the Novorossiya confederation (Biyatov, 2014).<sup>25</sup> The name of this quasi-state formation is the second term that is going to be investigated. Novorossiya has two distinct meanings. The first one is evident. When translated into English it means New Russia. Marlene Laruelle argues that "it announces the birth of a New Russia both geographically, in Eastern Ukraine, and metaphorically, in Russia itself" (Laruelle, 2015, p. 12). The second meaning is connected to the history of the Russian empire. After the Russo-Turkish wars of 1768–1774, Catherine the Great won vast territories (a significant part of present-day Ukraine including the Donetsk and Lugansk regions) from the Ottoman Empire (Laruelle, 2015, p. 2). The lands North of the Black Sea were given the name Novorossiya (Laruelle, 2015, p. 2). New Russia was part of the of the Russian empire until the October Revolution of 1917 (Alekseyevich, 2017). According to Andrew Wilson, when elites create an ideology they often use historical myths that have strong resonance within the targeted population (Wilson, 1995, p. 266). He argues that such myths need to be linked with the popular memory of the population and locate them in time and space (Wilson, 1995, p. 266). Wilson states that "an association with a specific homeland" is and effective tool, for the creation of a successful ideology. He continues, "the ideologues of a given ethnic group will stake claim to their historic territory by the claim to be 'indigenous' - to have been historically the first group, or significant group, to occupy that particular land" (Wilson, 1995, p. 266). Therefore, the name Novorossiya was intentionally chosen because it symbolizes the idea that ethnic Russians are the native population of Donbass and they have the right to take these lands from Ukraine. Even though the fact that since the end of the 19<sup>th</sup>, throughout the 20<sup>th</sup> and the beginning of the 21<sup>st</sup> centuries the majority ethnic group in Donbass were Ukrainians (Wilson, 1995, p. 275), (Nacionalnii sostav naseleniq, grajdanstvo, 2001).

One of the most celebrated national holidays in Russia is Victory Day (9<sup>th</sup> of May), which commemorates the Soviet Union's victory in the Second World War. The success over fascism is one the deepest symbols of national pride in Russia. This can be seen in the term that Russians use

<sup>&</sup>lt;sup>25</sup> In May 2015, the leaders of the Donetsk and Lugansk Peoples Republics announced that they the Novorossiya project has been put on hold (until further notice) because it violated the Minsk agreements (Kirilov, 2015)

for WW2 - the Great Patriotic War. The second dominant line in Novorossiya's ideology is connected to the exploitation of the historical symbolism of WW2. The leaders of Donetsk and Lugansk Peoples Republics, with the help of the Russian media and Russian officials, created the feeling amongst the separatists that they are fighting a "holy war" against fascism (Whewell, 2014). Two terms, which dominated the Russian public discourse during the first year of the crisis, were used to facilitate this process of antagonization. The first one is linked to the Ukrainian political activists and leader of the Organization of Ukrainian Nationalists (OUN) - Stepan Bandera (Vladimir Sazonov, 2016, p. 31). During WW2, the OUN split into two and Bandera and his followers formed the Ukrainian Insurgent Army (Vladimir Sazonov, 2016, p. 31). This paramilitary organization for a brief period of time collaborated with the Nazi forces in order to fight the Soviets in Ukraine (Vedeneev, 2002), (Vladimir Sazonov, 2016, p. 31). However, in 1941 Bandera was put into a German concentration camp because the same year his organization tried to establish an independent Ukrainian state (Vedeneev, 2002). In 1959, Bandera was killed by a Soviet secret agent in Munich, Germany (Mitrokhin, 1999, p. 362). For some Ukrainians, Bandera is a symbol for Ukraine's fight for independence, but in Soviet and Russian rhetoric he is a fascist and Nazi collaborator (Hyde, 2015). During the Ukrainian crisis, especially in 2014, the term "Banderivtsi" (in Russian followers of Bandera) was often used in the Russian public and official discourse as a synonym of the Ukrainian government and the Ukrainian Army (Riabchuk, 2016, pp. 81-82). For example, in 2014, during a speech, President Putin declared: "we can all clearly see the intentions of these ideological heirs of Bandera, Hitler's accomplice during World War II", referring to the Ukrainian authorities (Post T. W., 2014). The second popular epithet, which was often used in Russia's discourse to describe Ukrainian officials were "fascist junta" or simply "junta" (Vladimir Sazonov, 2016, p. 69), (BBC, 2014). In this respect, the message is clear. It can be argued that the specific Nazi discourse was used in order to reinforce the feeling of righteousness of the separatist's cause. The effectiveness of this operation is described by Maria Snegovaya "Unlike in the Soviet war in Afghanistan, many Russians volunteered to fight in Eastern Ukraine because they watched state television not because the military directly mobilized them" (Snegovaya, 2015, p. 15).

In conclusion, it can be seen that by using a specific terminology in the public and official discourse, the Russian military and strategic planners created an ideology which facilitated the hybrid operation in Ukraine. Historical myths and analogies were used for the development of an internal narrative of the separatists in Donbass. The Ukrainian officials were depicted as Nazis and

fascists, while the separatists as "heroic defenders of the motherland" hat are fighting a "holy war" against fascism. This had a mobilization effect not only amongst Russian-speakers in Donbass, but also amongst Russian citizens. As a result, many people volunteered to fight on the side of the separatists.

# 5.3. Existential threat to the hybrid actor

According to McCulloh's third principle, a hybrid actor will abandon "conventional military wisdom" as a result of a perceived "existential threat by a potential adversary" and their goal to achieve long-term survival.

Ukraine is one of the biggest countries in Europe. Throughout its history, it has always been under Russian influence (see section 4.2.1.). In February 2014, when the pro-European Maidan protests escalated and the pro-Russian Ukrainian president Viktor Yanukovych was dismissed from his position, Russia had a lot to lose (Kyivpost, 2014). This section will examine the key concerns of Russia's military and strategic planners (in relation to the Ukrainian crisis) and explain why they used an irregular and covert military approach, instead of openly engaging in a conflict.

## 5.3.1. Russia's geopolitical concerns in Ukraine

Between 1676 – 1878 Russia fought a number of wars with the Ottoman Empire (Chauhan, 2014). Russia's key goal during these wars was to take control over the Black Sea and the Dardanelles and Bosporus straits, thus potentially expanding its influence over the Mediterranean Sea (Chauhan, 2014). While they did not fully fulfill their objective (control over the straits), Russia managed to win the Northern coastline of the Black Sea and build warm-water ports (Chauhan, 2014). After the dissolution of the Soviet Union, and the establishment of an independent Ukrainian state, a significant part of the Northern coastline of the Black Sea was no longer Russian. However, in 1997 and in 2010, Russia and Ukraine signed a series of agreements which allowed Russia to station up to 25.000 troops, 132 armored combat vehicles in the Crimean Peninsula (Kimball, 2014). It can be argued that from a military perspective, Crimea is an "unsinkable aircraft carrier" and though its military bases situated there Russia can exert control over the whole Black Sea region. Consequently, for Russia, it was vital to keep its control over this territory. Also, Ukraine

is one of the most important Russian neighbors in Europe because amongst Russian officials there is a "conviction that in order to protect its borders Russia must exert control far beyond them" (Geers, 2015, p. 19). As Zbigniew Brzezinski has described it: "Without Ukraine, Russia ceases to be an empire, but with Ukraine suborned and then subordinated, Russia automatically becomes an empire" (Economist, 2013). Therefore, it can be seen that Ukraine is of high geostrategic importance to Russia.

Russia's biggest concern is Ukraine becoming US ally and a member state of NATO because this undoubtedly would change the balance of power not only in Europe, but also globally. In 2007, during the 43<sup>rd</sup> Munich Conference on Security Policy, President Vladimir Putin openly criticized NATO's enlargement in the East, and he also rejected the unipolar system of international relations (Post T. W., 2007). He stated: "I consider that the unipolar model is not only unacceptable but also impossible in today's world" (Post T. W., 2007). The following year in Bucharest, NATO's heads of states declared that they are considering to admit Georgia and Ukraine to the alliance (NATO, 2008). Sergei Markov, (a political scientist and in that time a member of the Russian Parliament) commented: "Georgia's accession into NATO will be seen here as an attempt to trigger a war in the Caucasus, and NATO membership for Ukraine will be interpreted as an effort to foment a conflict with Russia" (Isachenkov, 2008). Four mounts later Russia invaded Georgia (BBC, 2008). A clear message was sent to Ukraine, the US and NATO. In March 2014, after the Westernorientated provisional government took the power in Kev, the acting Foreign Affairs Minister of Ukraine Andriy Deschytsia, stated that Ukraine has no intention to join NATO (Interfax-Ukraine, 2014). However, after eight mounts Russia's fear was justified when the Ukrainian government announced that its major goal is membership in NATO (Europe, 2014).

## 5.3.1. Russia's economic concerns in Ukraine

As presented in section 4.2.4., both countries have strong economic connections. Russia's economy is highly dependent on fossil fuels export (Barnato, 2016). In 2014, oil and gas comprised 66 percent, of the country's total exports (Bank W., 2016, p. 55). According to the World Bank: "Russia's dependence on oil and gas exports is exacerbated by its focus on a single export market, the European Union" (Bank W., 2016, p. 55). A substantial number of Russia's oil and gas pipelines pass through Ukrainian territory. They include: the Druzhba pipeline (the biggest oil pipeline in the World), the Trans-Siberian Pipeline, and Yamal - Europe pipeline (Chausovsky,

2013). Therefore, Russia's primary economic sector is to a significant degree is dependent on Ukraine.

Another important factor in the economic relations between Russia and Ukraine is the Eurasian Economic Union – a Moscow-led project for economic integration between the countries from the former Soviet Union (Soyuz, 2017). Some experts argue that the union is part of a grander strategy for "re-Sovietization" and restoration of the Russian Empire (Hassan, 2017, p. 226). Taking into account the words of Zbigniew Brzezinski, quoted in the previous section, the Union would not be a powerful factor in international relations (both economically and politically), if Ukraine does not become a part of it. Russia has continuously opposed EU's projects for integration with Ukraine (Balmforth, 2013). In the winter of 2013, the Ukrainian government was in a position where a choice between the Eurasian Economic Union and the EU had to be made (Walker S. , 2013). The scholar Vsevolod Samokhvalov describes it as the "zero-sum logic [for] the future of Ukraine" (Samokhvalov, 2015, p. 1373). In the beginning of November 2013, Ukraine officials were in favor of closer relations with the EU, yet on the 21<sup>st</sup> the same month, after discussions between the Russian and Ukrainian prime ministers, President Viktor Yanukovych, declared that he is suspending talks with Brussels (Grytsenko I. T., 2013). It can be argued that this decision, marked a turning point and because of it the Ukrainian crisis started.

## 5.3.3. Analysis of the existential threat of the hybrid actor

Overall, Russia's main concerns in relation to Ukraine can be summarized as: Ukraine becoming a strong US ally, a member of NATO and the EU. If the Russian elite wants to restore the "the largest geopolitical catastrophe of the [20th] century" (the collapse of the Soviet Union), Ukraine has to be under Russian control. When the Ukrainian Parliament dismissed the pro-Russian President Viktor Yanukovych in February 2014, Russian officials knew that the new Ukrainian government would not serve their interests (Kyivpost, 2014). Therefore, they had to intervene. A potential option was to officially engage in an open conflict. The Russian Army is much bigger and better equipped than the Ukrainian (Recknagel, 2014). As President Vladimir Putin had declared in September 2014, Russia could "take Kiev in two weeks" (Squires, 2014). In 2008, Russia used a similar approach during the Russo-Georgian War and fulfilled their objectives in five days (BBC, 2008). Yet during this conflict, Russia suffered high equipment losses from a country with a population of around 4.382 million (in 2008), and a defense budget of 1.09 billion dollars

(McDermott, 2009, pp. 70-73), (Georgia, 2008) (Bank W., 2017). A conventional military operation in Ukraine, which is one of the biggest European countries with a population of approximately 45.553 million, and a military expenditure of 4.01 billion dollars (in 2014), would have been extremely expensive for Russia (Statistics, 2013), (Bank W., 2017). Also, the Russian oil and gas pipelines which pass through Ukraine probably would have been the first target of the Ukrainian Army. If they have been destroyed this would have had an extremely negative effect on Russia's economy which as, already mentioned, is highly dependent on the export of fossil fuels to the EU (Bank W., 2016, p. 55). Moreover, a full-scale Russian invasion of Ukraine would have had unforeseen international consequences, because when the crisis escalated, both the EU and the US repeatedly had demonstrated their support for Ukraine (Yuhas, 2014).

Therefore, it is easy to understand why Russia did not openly intervene in Ukraine. As President Putin had declared: "Our responses should be based on intellectual superiority. They will be asymmetric [and] less costly" (Buckley, 2006). Russia's military and strategic planners exploited the advantageous conflict setting (described in section 4.2.) and launched an irregular and covert military campaign in Ukraine. First, by annexing the Crimean Peninsula they guaranteed Russia's geostrategic domination in the Black Sea region. Second, by stimulating anti-Ukrainian sentiment among the Russian minority in the South-Eastern regions of the country, they triggered a civil war, which is continuously whittling away Ukraine's economic resources. <sup>26</sup> It can be argued that in the long term the conflict in Donbass may increase Ukraine's economic dependence on Russia. This military approach was much more efficient than a conventional intervention.

# **5.4.** Attritional tactics

McCulloh's seventh principle holds that a hybrid actor will use attritional tactics in the cognitive domain in order to undermine the morale of the conventional opponent and to diminish their capabilities.

During the operations in Crimea and Donbass, Russia took advantage of the unique context of the conflict. Particularly the socio-cultural and temporal settings were exploited, with the prime

<sup>&</sup>lt;sup>26</sup> In 2013 the Ukrainian GDP was 183 billion dollars (Bank T. W., 2017). Today it is almost twice as less - 93 billion dollars (Bank T. W., 2017).

goal to weaken Ukraine's security forces will to fight. Different approaches were utilized during the distinct stages of the crisis. However, in many cases they overlapped.

When the crisis begun, in both theaters of action, the initial tactic included the mobilization of civilians (the creation of a hive mind, see section 5.1), which were directed to occupy government buildings. It can be argued that in this stage of the crisis, the Ukrainian government could have restored its control over the country. However, the local policeman did not intervene and in many occasions, they even cooperated with the protesters (Harding, 2014). Russia's massive information attack, which was successful in the creation of an environment of fear and discontent amongst the Russian minority in Ukraine had also affected Ukraine's security forces (see section 5.1). In an interview, a local Ukrainian officer from Donetsk explains why he remains passive while pro-Russian protesters take control over the city: "This situation is all Kiev's fault. They say we in the east are slaves, half-humans. They revere people like Stepan Bandera [the second world war Ukrainian nationalist leader] who shot our brothers. We are normal citizens like everyone else". Russia's information campaign was so successful that in June 2014, the Interior Minister of Ukraine Arsen Avakov, reported that approximately 20 000 policemen have joined the separatists in Donbass (Nedeli, 2014).

During the period when Russia was preparing for the annexation of Crimea, the messages that the Ukrainian government is an illegitimate "fascist junta", were reinforced by another cognitive operation which also had a ground component. The main target was the Ukrainian Navy. The first step in Russia's campaign was to besiege the Ukrainian solders station in the Peninsula. Next, Russia exploited the high levels of corruption in the country (see section 4.2.4.). On the 2<sup>nd</sup> of March, the commander-in-chief of the Ukrainian Navy Admiral Denis Berezovsky, who was appointed the previous day by the Interim President Oleksandr Turchynov, announced that he is defecting to the Russian-supported Crimean authorities (Walker S. , 2014). In his announcement, he declared his loyalty to the dismissed President Viktor Yanukovych and stated that: "The seizure of power in Kiev was orchestrated from abroad" (Walker S. , 2014). Three weeks afterward, Berezovsky became the Russian Black Sea Fleet deputy commander (Post K. , 2014). This act of treachery in the highest level of the Ukrainian Army, undoubtedly had demoralized the Ukrainian servicemen. In March 2014, they were in a position where their fate was put under question. Not only the soldiers were besieged by the Russian special forces but also, they were incapable to do

anything about it because they did not receive any orders from Kiev (Sengupta, 2014). The same month there were reports that Russian officers had offered the Ukrainian soldiers in Crimea, higher salaries, free flats and an "equal position in the armed forces of a future Crimean state" if they defect (Shuster, 2014), (Sengupta, 2014). A 26-year-old Ukrainian sergeant explains why he had joined the Russia side:

I cannot see how the situation can change, Crimea will become a part of the Russian Federation. We as soldiers were not allowed to do anything; we were without direction. What is this government in Kiev doing? Nothing for us. We got no help from them. At the end, what choice do we have but to join the new army? We were let down, weren't we?

This Russia operation was quite successful. The result was that more than 50% of the Ukrainian Navy serviceman defected to Russia, 25% deserted and only 25% stayed in the Ukrainian Army (Bartkowski, 2015, p. 9).

In the end of March 2014, the Ukrainian government announced partial mobilization (Press, 2014). It included three subsequent waves (in the end of March, the beginning of May, and in the end of July), and by the end of 2014, approximately 104 thousand Ukrainians were mobilized (Sevodnya, 2016). When the fight in Donbass intensified, the Russian media reacted swiftly and launched an information campaign that targeted the men that had to be mobilized, their relatives and the Ukrainian society at large (Vladimir Sazonov, 2016, p. 110). The prime goal was to spread fear and discourage the people that had to go fight the separatists. One of the main tools that was used were news reports on Russian and linked to Russia Ukrainian media. They included news pieces on the poor conditions in the army, pictures and videos of bloodied and mutilated corpses of Ukrainian soldiers, also captured Ukrainian soldiers, that are being humiliated (Vladimir Sazonov, 2016, pp. 86-110). This Russian campaign was successful and in the summer of 2014, there was a wave of ant-mobilization protests in Ukraine (Ukraina R. N., 2014). By August 2014, more than 16 000 cases of desertion were reported (Reserch, 2015). The problem was so big that in the beginning of 2015, the Ukrainian government passed a law that allowed commanders to shoot deserters (RT, 2015) (Rada, 2015).

In conclusion, during the different stages of the Ukrainian crisis, Russia successfully used attritional tactics in the cognitive domain. A significant part of Ukraine's security forces were demoralized and as a result a great number of them either defected or deserted.

# 5.5. Conclusions

In this chapter, Russia's actions in the non-physical domain of operations were analyzed through the theoretical framework of hybrid warfare, developed by Timothy McCulloch. Four of his seven principles were applied to the Ukrainian case. It can be concluded that Russia's military approach exposed during the crisis fits McCulloch's understandings of hybrid warfare.

# 6. General conclusions

"It's extremely dangerous to try to resolve political problems outside the framework of the law"

Vladimir Putin, 2004 (McLaughlin, 2004)

In this final chapter, the results of the research that explored Russia's military approach during the Ukrainian crisis (2013-2015) will be presented. First, the answers to the sub research questions will be summed up. Then, an answer to the central research question will be suggested, followed by a discussion regarding the limitations of Timothy McCulloh's hybrid warfare theoretical framework and this particular study as well. In the final section, recommendations for further research will be offered.

## **6.1.** Answers to sub research questions

### 6.1.1. Sub research question number one

The first sub research question, "what are hybrid wars?", was discussed in the second chapter which presented McCulloh's theoretical framework. According to him, hybrid wars are: "an optimized form of warfare that allows a combatant to attempt to utilize all available resources—both conventional and unconventional—in a unique cultural context to produce specific effects against a conventional opponent" (Johnson, 2013, p. 16). After a review of the theoretical developments which examine this form of conflict, a more detailed explanation was suggested. According to it:

hybrid wars are not only a form of conflict, but also a strategy and a tactic. Such conflicts can be initiated by both state and non-state actors. Their effectiveness and efficiency derive from the exploitation of the unique context of the conflict and the optimization of the hybrid entity's force structure by the utilization of both conventional and unconventional resources, that are simultaneously applied in both the physical and cognitive domains, thus producing a synergistic effect. This allows the hybrid entity to overcome a more powerful (in terms of resources and equipment) conventional actor. (see section 2.3.).

#### **6.1.2.** Sub research question number two

In the third chapter, the research design of the thesis was presented. In doing so, the second sub research question "how can the Ukrainian crisis (2013-2015) be studied through the lens of the hybrid war theory?" was answered. The investigation was organized as an exploratory case study with two units of analysis. The focus of the first one was on expressions of hybrid warfare in the physical domains of operations, and the second, on expressions of hybrid warfare in the nonphysical domain of operations. In order to facilitate the research, a time boundary was set. It covered the period from November 2013 to February 2015, because this was the active phase of the Ukrainian conflict. Within the case study, two research methods were used. Theory testing process tracing was selected as the main method of analysis, through which both units of the case study were examined. In chapter three, two of the three steps, in theory testing PT were performed. Thus, Timothy McCulloh's hybrid war theory was reformulated in order to express causality and his seven principles of hybrid warfare were operationalized and divided into two groups in consonance with the units of analysis. The third step in theory testing PT - the collection of evidence was carried throughout the fourth and fifth chapters. The second method that was used was discourse analysis. It enabled the exploration of the nuanced characteristics of Russia's military approach in the non-physical domain of operations. The employed research design allowed for a high validity of this work.

#### **6.1.3.** Sub research question number three

The third sub research question, "which expressions of hybrid warfare in the physical domains of operations can be identified in the Ukrainian crisis?" was examined in the fourth chapter. In doing so, the first unit of the case study was explored. Five of McCulloh's principles of hybrid warfare were applied to the Ukrainian case. It has to be noted that Russia's campaigns in Crimea and Donbass, were not exactly the same (probably because the objectives in both theaters of actions were different) however, important similarities were found. Within them, Russia's military approach in the physical domains of operations was exposed. The findings will be presented in detail, because the campaign was quite complex and involved many important components.

In the core of Russia's overall strategy was the exploitation of the unique conflict context which according to McCulloh's first principle, includes the historical, socio-cultural, geographic

and temporal settings. The historical overview showed that for more than a millennium Russia and Ukraine have shared their past, which has strong implications for the socio-cultural and geographic settings. First, in the South-Eastern regions of Ukraine, a significant part of the population is either ethnic Russians or Russian-speakers. These people were mobilized and organized into serving Russia's interest. They were used as a fifth column, which allowed Russia to present the crisis as a legitimate bottom-up initiative. Initially, in both theaters of action an anti-Ukrainian protest movement was formed, then these people were directed to occupy governmental buildings (in Crimea with the support of the Russian special forces). In Donbass, when violence escalated, they formed voluntary combat groups, that fought the Ukrainian Army. Second, Russia's military and strategic planners also took advantage of their common border with Ukraine and their military bases situated on the Crimean Peninsula.<sup>27</sup> In Donbass, Russia directly sent military personnel (with no insignia) and heavy military equipment to support the separatists, and in Crimea, the Russian military bases were used for the deployment of the little green men. Russia also exploited the temporal setting. They swiftly launched their operations in Crimea and Donbass (while they still had the advantage) at a time when Ukraine was in chaos and the new pro-Western political circle had not consolidated their positions and had not received substantial help from abroad.

According to McCulloh's theoretical framework, in order to classify a conflict as a hybrid one, there should be a capability overmatch between the more powerful (in terms of military posture) conventional opponent and the hybrid actor. The analysis proved that in Crimea the little green men had only light weapons, against the better equipped Ukrainian Army.<sup>28</sup> Thus, there was a capability overmatch. Whereas in Donbass, the separatists had in their possession major conventional arms and in terms of types of weapons used by both sides there we no differences. This was the first discrepancy that was found between McCulloh's hybrid warfare principles and the Russian military operation in Ukraine.

Both in Crimea and Donbass, Russia used a combination of conventional and unconventional elements, which is in consonance with McCulloh's fifth principle. During the operation for the annexation of the Peninsular, the unconventional forces included the pro-Russian

\_

<sup>&</sup>lt;sup>27</sup> The geographic proximity between both countries and the Russian military bases in Crimea were also used as a deterrent measure.

<sup>&</sup>lt;sup>28</sup> The focus of the analysis was on the actual forces that were used. As a result, the official Russian military personnel situated in Crimea was not included in the investigation because they did not openly participate in the annexation of the Peninsula, although they were used as a deterrent measure.

protesters, among which there were "titushky" and the little green men<sup>29</sup> (see section 4.2.5.). The Russian militaries stationed on the Peninsula were used as a deterrent measure which was the conventional component in this operation. In Donbass, Russia used a similar approach. During the initial phase of the crisis (before the Ukrainian Army was sent to fight the separatists) the irregular forces were comprised of pro-Russian protesters and "titushky". When violence escalated, a number of "voluntary groups" or in other words pro-Russian insurgents, appeared in Eastern Ukraine. These groups included both local protesters and tens of thousands guerrilla fighters which had entered into Ukraine from Russia. The operation in Donbass, as in Crimea had also a conventional element. Between April and September 2014, Russia carried a number of military exercises, two of which along its' border with Ukraine. These demonstrations of military superiority were used as a deterrent measure.

McCulloh argues, that an actor that is waging a hybrid war will develop a defensive strategy and rely mostly on defensive operations. In Crimea, although there were no fights, Russia's actions were examined as offensive because they involved an armed besiegement of official military forces, whereas in the Donetsk and Lugansk regions (during the initial phase), the protesters did not receive support from the Russian special forces – the little green men. This is the second discrepancy between McCulloh's hybrid warfare principles and the Russian military operation in Ukraine which was found. During the active phase of the conflict in Donbass, it was concluded that the overall strategy used by the separatist was defensive. Their goal was to establish and hold their control over the capitals of the two regions Donetsk and Lugansk and gain legitimacy over as much territory they as they can (within the original borders of both regions).

Russia's military approach also fit the description of McCulloh's last principle of hybrid warfare. Three attritional tactics were identified. The first one was used during the initial phase of the crisis (before April 2014), both in Crimea and Donbass. The occupation of governmental buildings by civilians (in Crimea supported by the little green men) was an effective and efficient approach that Russia used in order to take control over the South-Eastern regions of Ukraine. Within it, three interconnected components were identified. First, the participation of civilians allowed Russia to present its actions as a struggle for self-determination. Second, this approach did not involve an escalation of violence which enabled the pro-Russian forces to consolidate their

\_

<sup>&</sup>lt;sup>29</sup> The little green men were examined as an irregular force because when they were executing their tasks they did not have any insignia on them. Only after Russia had fulfilled its objective in Crimea, President Putin admitted that they were actually Russian militaries.

positions. Third, the Ukrainian government was placed in a position where in they had used force against the protesters, they would have lost the support of the US and the EU. As a result, in Crimea the besieged Ukrainian soldiers felt that they were let down by their new government. The second attritional tactic was examined in Donbass after Russia had annexed the Crimea and the fights had broke out. During the period when Ukraine launched their massive attacks in the summer of 2014, the separatists successfully averted a major battle which could have decided the outcome of the conflict. They gradually retreated to their strongholds in the cities of Donetsk and Lugansk and waited for support from Russia. The third attritional tactic, became apparent when the Donbass separatist initiated their two major offenses in August 2014 and in January 2015. In both occasions, the separatists received military support from Russia, which allowed them to gain control over the battlefield. During both attacks, the separatist managed to encircle thousands of Ukrainian soldiers. Their swift successes made it possible for them to enter into peace negotiations. Thus, two documents were signed – the first and second Minsk Protocols. The result was that the separatists strengthened their positions and today the conflict in Ukraine is at a stalemate situation.

Overall these were the expressions of hybrid warfare identified in the physical domains of operations. It can be concluded that during the initial phase of the conflict the Russian military approach both in Crimea and Donbass to a significant degree overlapped, yet in Eastern Ukraine (Donetsk and Lugansk regions) once the violence had escalated, Russia used a more complex strategy.

#### 6.1.4. Sub research question number four

In the fifth chapter, the second unit of analysis of the case study and the forth sub research question "which expressions of hybrid warfare in the non-physical domain of operations can be identified in the Ukrainian crisis?" were explored. Four of McCulloh's principles of hybrid warfare were applied to the Ukrainian case. As a result, key characteristics of Russia's military approach in the non-physical domain of operations were presented.

Russia's overall strategy in Ukraine was centered around exploitation of the unique conflict context (McCulloh's first principle). In chapter five it was shown how Russia managed to mobilize and utilized the Russian-speaking minority in Ukraine. This was achieved through a massive information attack, launched on the strongly centralized Russian media. The main tool that was used was fear, with the purpose to consolidate the Russian-speaking minority, trigger an emotional response and direct these people to carry actions against the Ukrainian government. These goals

were achieved by framing. The pro-European Maidan protesters were represented as violent radicals, and the new Ukrainian government, as an illegitimate fascist junta, which is threatening the rights of the Russian minority. It has to be noted that the way Russia conducted its information operation in Ukraine, "to a significant degree, followed the guideline for indirect regime change operations created by the Russian scholar Andrew Korybko" and the 2014 Russian military doctrine (see section 5.1.).

McCulloh's second principle was also explored. According to it, an actor that is waging a hybrid war will develop a specific ideology which is linked to the unique context of the conflict and supports and justify one's actions. Through the use of discourse analysis, which was focused on four specific terms (opolcheniye, Novorossiya, Banderivtsi and fascist junta) used in the Russian public and official rhetoric, it was proven that the leaders of separatists in Donbass, along with Russian officials and the Russian media, successfully exploited the unique historical setting. By the used historical myths and analogies, they managed to develop an internal narrative of the hybrid organization.<sup>30</sup> The result was that, the separatists were represented as heroic defenders of the motherland, that are fighting a "holy" war against fascism. This ideology had a mobilization effect. As consequence, many people volunteered to fight on the side of the separatists.

In relation to McCulloh's third principle, it was argued that Ukraine is of great importance to Russia's economy and geopolitics. The Federation's main concerns include: Ukraine becoming a strong US ally, a member of NATO and the EU. Therefore, when in February 2014, Western-orientated politicians took the power in Kiev, the Russian elite perceived this political shift as an existential threat (Kyivpost, 2014). The analysis showed that if Russia's military and strategic planners had decided to initiate a conventional war in Ukraine, this would have had not only a devastating effect on Russia's economy but also, unforeseen international consequences. The decision to use an irregular and covert military approach served Russia's interests because they managed to guarantee their geostrategic domination in the Black Sea region and whittle away Ukraine's economic resources (which will probably increase Ukraine's economic dependence on Russia in the future) at very low costs.

By exploiting the advantageous conflict setting, Russia successfully used attritional tactics in the cognitive domain of operations. This corresponds to McCulloh's last principle of hybrid

<sup>&</sup>lt;sup>30</sup> The focus of the analysis was on the separatists in Donbass, because in Crimea, there was not enough, time for the formation of an ideology.

warfare. Three tactics were identified. They were used during the different stages of the crisis, yet they also overlapped and reinforces each other. First, the massive information campaign described in the second paragraph of this section which targeted the Russian minority had also affected Ukraine's security forces. As a result, when the anti-Maidan protests escalated, a significant number of the local policemen did not fulfill their duty and in many cases, they started cooperating with the protesters because they felt fear and discontent for the new Ukrainian government. The second attritional cognitive tactic, which also had a ground component, was more complex and it targeted the Ukrainian Navy servicemen in Crimea. Within it three aspects were identified. Firstly, while the Russian media was continuously representing the Western-orientated Ukrainian government as an illegitimate fascist junta, the Ukrainian soldiers in Crimea were besieged by armed men. In that time, they could have broken the siege, yet they did not receive any orders from Kiev. Consequently, the Ukrainian Navy servicemen started feeling that they were left behind by their new government. Secondly, Russia exploited the high levels of corruption in the country, and commander-in-chief of the Ukrainian Navy Admiral Denis Berezovsky defected to the Russiansupported Crimean authorities, which had a demoralizing effect on the besieged soldiers. Thirdly, Russia again took advantage of the high levels of corruption, and the besieged and demoralized soldiers were offered higher salaries, free flats and an equal position in the Russian Navy, if they defect. This tactic was quite effective and in the end, Ukraine lost approximately 75% of its Navy servicemen (Bartkowski, 2015, p. 9). The third attritional tactic was launched after the armed conflict in Donbass had begun. The main target were the Ukrainian men that were mobilized, their families and the Ukrainian society at large. Russia again utilized its strongly centralized media, which repeatedly showed horrific pictures and videos of mutilated corpses of Ukrainian soldiers and captured Ukrainian servicemen that are being humiliated. Once more Russia was successful in the creation of an environment of fear. The results were that ant-mobilization protests spread across Ukraine and thousands of men deserted.

In sum, these were the expressions of hybrid warfare in the non-physical domain of operations which were identified in the Ukrainian crisis.

## **6.1.** Answers to central research question

Before the central research question is answered, an important component of McCulloh's framework, which was not examined in chapters four and five has to be presented. According to him the efficiency and effectiveness of hybrid wars, derive from the synergistic effects which their integral parts produce (Johnson, 2013, p. 17). In this regard, it can be stated that Russia's hybrid operations in physical and non-physical domains were not separate initiatives. Instead, they were part of a whole and were carried out simultaneously, thus a grander strategy was exposed. At the core of Russia's intervention in Ukraine was the exploitation if the unique context of the conflict. In relation to it, the first step that Russia's military and strategic planners took, was to launch a massive information attack (in the non-physical domain), the main goal of which was to mobilize the Russian-speaking minority in Ukraine. Once the "cognitive environment" was right, the second step was to direct these people to carry out physical actions – first, to protest then, to occupy governmental buildings and, in Donbass to fight. If the non-physical operation had failed, the physical operation would not have started in the first place. The third step was to demoralize Ukraine's security forces. During the different stages of the crisis, distinct steps were taken. The attritional tactics, described in sections 4.6. and 5.4., reinforced each other. In the beginning of the crisis the occupation tactic was supplemented by the first and second cognitive attritional tactics (see previous section, paragraph five). During the fights in Donbass, the second and third physical attritional tactics (see section 6.1.3, paragraph six) were augmented by the third cognitive attritional tactic tactics (see previous section, paragraph five). Therefore, Russia's overall strategy in Ukraine, was based on the synergistic combination of operations both in physical and non-physical domains.

The main goal of this research was to answer the question, "To what extent can the Russian actions in the Ukrainian crisis (2013-2015) be conceptualized as an exemplification of a hybrid war according to the theory of McCulloh?". In chapter three McCulloh's definition of the theory of hybrid warfare was reformulated in order to express causality as:

If during a conflict one of the combatants -X bases its optimized force structure on the combination of all available resources - both conventional and unconventional - in a unique cultural context to produce specific, synergistic effects against a conventionally-based opponent, then this entity is waging a hybrid war -Y (see section 3.3.1.).

When applied to this particular study, the conflict is the Ukrainian crisis (2013-2015), X - the Russian Federation and, Y - hybrid warfare. If X is to produce Y, then McCulloh's characteristics of hybrid warfare need to be met. The analysis proved that the Russian actions in the Ukrainian crisis (2013-2015), with two exceptions (the lack of a capability overmatch in Donbass and offensive strategy used in Crimea), meet McCulloh's theorization. Therefore, it can be concluded that the military approach used by the Russian Federation, in Ukraine can be conceptualized as an exemplification of a hybrid war according to the theory of McCulloh.

#### 6.3. Limitations of McCulloh's theoretical framework

During the analysis, specific limitations McCulloh's theory of hybrid warfare were found. The Russian military approach exposed during the Ukrainian crisis, was slightly more complex than the one described by McCulloh. His theory failed to account for three important aspects of the way Russia conducted its operations in Ukraine. The first one is related to the use of new technologies. During the crisis, a key aspect of the Russian overall strategy was the conduction of operations in the cyber domain (Geers, 2015). These operations facilitated Russia's actions both in the physical and non-physical domains (Geers, 2015, pp. 123-134). Diplomacy is the second aspect of the Russian grand strategy, that McCulloh's framework did not allow to be examined. During the crisis, Russian officials took different steps on the diplomatic level that increased the efficiency and effectiveness of their actions in Ukraine (Snegovaya, 2015, p. 18). The third important feature of Russia's strategy which was not examined is the use of economic coercion. Russia's officials exploited the advantages deriving from their country's strong economic connections with Ukraine and used it as an active means against them (Seely, 2015, pp. 15-17).

In conclusion, one could assert, that even though the Russian actions in the Ukrainian crisis (2013-2015) can be conceptualized as an instance of a hybrid war, according to McCulloh's work, it is not certain, if his theory is the most appropriate analytical tool for this particular case.

# 6.4. Limitations of the thesis and avenues for further research

This study offered a detailed picture of the Russian actions in the Ukrainian crisis. Yet the analysis was guided only by McCulloh's framework. As a result, key aspects of the Russian overall strategy used in Ukraine was not fully explored (see previous section). In relation to this limitation, further research is needed. One could apply different theoretical frameworks for example, Janis Berzins's new generation warfare, and/or Oscar Jonsson's and Robert Seely's full-spectrum conflict, to the Ukrainian case and then compare the results (Bērziņš, 2015), (Seely, 2015). Such a work will show what is the most suitable analytical tool and the best way to conceptualize the new Russian military approach.

The second limitation is also connected to the narrow focus of the thesis. The ongoing Ukrainian crisis is an example of the new 21<sup>st</sup> century Russian military approach. However, one cannot draw complete conclusions regarding it only from this work because the unique conflict setting (as presented in the fourth and fifth chapters) was quite favorable. If the modern Russian warfare is to be fully studied, other cases need to be explored. A potential avenue for further research can be an investigation of Russia's involvement in the continuing Syrian Civil War (BBC, 2017). Once data is obtained one could investigate parallels between the Russian actions in Ukraine and Syria. The external validity of such a work would be high and the new Russian military approach, to a significant degree, would be exposed.

The presented analysis was also limited by the time boundary which was set in chapter three (from November 2013 to February 2015). Although this was the active phase of the crisis, it can be argued that the period after February 2015 is also important because the crisis has not been settled yet. A future research may focus on a wider time frame and explore what happened after the second Minsk agreement. Such a research may suggest an explanation of the actual Russian objectives in Donbass.

# **Bibliography**

Alekseyevich, A. M. (2017). *Ukraine*. Retrieved 05 29, 2017, from Encyclopedia Britannica: https://www.britannica.com/place/Ukraine/History#ref404367

Armor, L. (2016). *Poteri bronetehniki v voine Ukraine*. Retrieved 06 05, 2017, from Lost Armor: http://lostarmour.info/analytics/

- Arwa Damon, M. P. (2014). *Ukraine: Photos show undercover Russian troops*. Retrieved 06 08, 2017, from CNN: http://edition.cnn.com/2014/04/21/world/europe/ukraine-crisis/index.html
- Bachmann, S. (2015). Hybrid Wars: The 21st-Century's New Threats to Global Peace and Security. *South African Journal of Military Studies*, 43(1), 77 98.
- Baczynska, A. V. (2014). *Acknowledging defeat, Ukraine pulls troops from Crimea*. Retrieved 06 04, 2017, from Reuters: http://www.reuters.com/article/us-ukraine-crisis-crimea-base-idUSBREA2N09J20140324
- Balmforth, R. (2013). *Ukraine leader ignores Putin warning on EU path*. Retrieved 07 08, 2017, from Reuters: http://www.reuters.com/article/us-ukraine-russia-yanukovich-idUSBRE97N05P20130824
- Bank, T. W. (2015). *Russian Federation*. Retrieved 04 30, 2017, from The World Bank: http://data.worldbank.org/country/russian-federation
- Bank, T. W. (2017). *Ukraine*. Retrieved 07 26, 2017, from The World Bank: http://data.worldbank.org/country/ukraine
- Bank, W. (2016). Russian Economic Report, No. 35, April 2016: The Long Journey to Recovery. Washington: World Bank Group. Retrieved 07 08, 2017, from https://openknowledge.worldbank.org/handle/10986/24028
- Bank, W. (2017). *Military expenditure % of central government expenditure*. Retrieved 07 14, 2017, from World Bank: http://data.worldbank.org/indicator/MS.MIL.XPND.GD.ZS?locations=GE-UA
- Barnato, G. C. (2016). Russia too dependent on oil—and it's hitting the ruble, says ex-Russian fin min. Retrieved 07 08, 2017, from CNBC: http://www.cnbc.com/2016/06/17/russia-too-dependent-on-oil-and-its-hitting-the-ruble-says-ex-russian-fin-min.html
- Bartkowski, M. (2015). *Nonviolent Civilian Defense to Counter Russian Hybrid Warfare*. Washington: The Johns Hopkins University Center for Advanced Governmental Studies.
- Bastion. (2017). Sevastopol Glavnaq Voenno-Morska baza Chernomorskovo Flota. Retrieved 05 29, 2017, from Voenno Tehnicheskii Sbornik Bastion: http://bastion-karpenko.narod.ru/CHF\_2012\_05\_foto\_01.html
- BBC. (2008). *Q&A: Conflict in Georgia*. Retrieved 05 03, 2017, from The British Broadcasting Corporation: http://news.bbc.co.uk/2/hi/europe/7549736.stm
- BBC. (2013). *Ukraine protests after Yanukovych EU deal rejection*. Retrieved 04 24, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-25162563
- BBC. (2014). *Russian troops 'overrun Crimea's Feodosia naval base'*. Retrieved 06 04, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-26710884
- BBC. (2014). *Ukraine conflict: Turning up the TV heat*. Retrieved 06 30, 2017, from The British Broadcasting Company: http://www.bbc.com/news/world-europe-28706461
- BBC. (2014). *Ukraine crisis: Donetsk and Luhansk 'facing siege'*. Retrieved 06 14, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-28630391

- BBC. (2014, April). *Ukraine crisis: President vows to fight pro-Russia forces*. Retrieved 03 20, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-27011605
- BBC. (2014). *Ukraine crisis: Shells hit Donetsk amid Russia convoy row*. Retrieved 06 14, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-28788945
- BBC. (2014). *Ukraine crisis: Timeline*. Retrieved 04 24, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-middle-east-26248275
- BBC. (2014). *Ukraine: Speaker Oleksandr Turchynov named interim president*. Retrieved 04 24, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-26312008
- BBC. (2015). *Ukraine conflict: US accuses rebels of 'land grab'*. Retrieved 06 21, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-30928107
- BBC. (2015). *Ukraine crisis in maps*. Retrieved 24 04, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-27308526
- BBC. (2015). *Ukraine crisis: Army retreats at Donetsk airport*. Retrieved 04 27, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-30929344
- BBC. (2017). *Syria profile Timeline*. Retrieved 07 31, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-middle-east-14703995
- Bennett, A. G. (2012). Case studies and theory development in the social sciences. In A. B. Checkel, *Process tracing: from philosophical roots to best practices*. Simons Papers in Security and Development.
- Bērzinš, J. e. (2015). The War in Ukraine: Lessons for Europe. Riga: University of Latvia Press.
- Biyatov, E. (2014). The self-proclaimed Donetsk and Lugansk People's Republics have merged in one state. Retrieved 07 04, 2017, from The Voice of Russia: https://sputniknews.com/voiceofrussia/news/2014\_05\_24/Donetsk-Lugansk-Peoples-Republics-unite-in-Novorossiya-1012/
- Blum, R. (2015). *The future of NATO in the face of hybrid threats*. Retrieved 03 20, 2017, from Academica: https://www.academia.edu/11044703/THE\_FUTURE\_OF\_NATO\_IN\_THE\_FACE\_OF\_HYBRI D THREATS
- Boffey, D. (2016, December). *Senior UK MPs to be briefed on dangers posed by Putin's Russia*. Retrieved 03 20, 2017, from The Guardian: https://www.theguardian.com/world/2016/dec/17/senior-mps-to-be-briefed-on-dangers-posed-by-putins-russia
- Bowen, G. (2009). Document Analysis as a Qualitative Research Method. *Qualitative Research Journal*, 27-40.
- Britannica, E. (2017). *Crimea*. Retrieved 04 24, 2017, from Encyclopedia Britannica: https://www.britannica.com/place/Crimea
- Britannica, E. (2017). *Crimea*. Retrieved 05 30, 2017, from Encyclopedia Britannica: https://www.britannica.com/place/Crimea

- Brun, D. B. (2011). What is Process-Tracing Actually Tracing? The Three Variants of Process Tracing Methods and Their Uses and Limitations. Washington: APSA 2011 Annual Meeting Paper.
- Buckley, N. (2006). *Putin warns against US double standards*. Retrieved 06 07, 2017, from Financial Times: https://www.ft.com/content/3b67088e-e00f-11da-9e82-0000779e2340
- Bulgaria, M. n. (2016). *Nacionalna Otbranitelna Strategiq*. Retrieved 03 22, 2017, from Ministerstvona Otbranata: www.strategy.bg/FileHandler.ashx?fileId=6848
- Bussell, J. (2013). *Cyberspace*. Retrieved 05 20, 2017, from Encyclopedia Britannica: https://www.britannica.com/topic/cyberspace
- Charap, S. (2015). The ghost of hybrid war. *Survival*, *57*(6), 51-58.
- Chauhan, Y. (2014). *Russo-Turkish wars*. Retrieved 07 07, 2017, from Encyclopedia Britannica: https://www.britannica.com/topic/Russo-Turkish-wars
- Chausovsky, R. K. (2013). *Pipelines Of Empire*. Retrieved 07 08, 2017, from Forbes: https://www.forbes.com/sites/stratfor/2013/11/13/pipelines-of-empire/#3af6392c5bc3
- Chen, X. (2010). An English Dictionary with AB Index and Frequency. Great Britain: by Xuhula Chen.
- CIA. (2017). *The World Factbook*. Retrieved 04 24, 2017, from Central Intelligence Agency: https://www.cia.gov/library/publications/the-world-factbook/geos/up.html
- Clausewitz, C. v. (1873). On war (Vol. 1). (J. J. Graham, Trans.) London: N. Trübner & Company.
- Coalson, R. (2014). *Pro-Russian Separatism Rises In Crimea As Ukraine's Crisis Unfolds*. Retrieved 06 30, 2017, from Radio Free Europe: https://www.rferl.org/a/ukraine-crimea-rising-separatism/25268303.html
- Commission, E. (2016). *Joint Framework on countering hybrid threats a European Union response*. Retrieved 04 30, 2017, from EUR-Lex: http://eur-lex.europa.eu/legal-content/en/TXT/?uri=CELEX%3A52016JC0018
- Committee, U. H. (2009). *Concluding observations of the Human Rights Committee Russian Federation*. Retrieved 05 22, 2017, from http://www2.ohchr.org/english/bodies/hrc/docs/co/CCPR.C.RUS.CO.6.pdf
- Cranston, M. (2014). *Ideology*. Retrieved 07 03, 2017, from Encyclopedia Britannica: https://www.britannica.com/topic/ideology-society
- Cullen, E. R.-K. (2016). *What is Hybrid Warfare?* Retrieved 03 20, 2017, from Norsk Utenrikspolitisk Institutt: https://brage.bibsys.no/xmlui/handle/11250/2380867
- Darczewska, J. (2014). *The Information War on Ukraine New Challenges*. Warsaw: Cicero Foundation. Retrieved 06 28, 2017, from http://www.cicerofoundation.org/lectures/Jolanta\_Darczewska\_Info\_War\_Ukraine.pdf
- Derrida, J. (2006). Writing and Difference. In L. Hansen, *Security as practice: discourse analysis and the Bosnian war*. New York: Routledge.
- Dictionaries, O. (2017). *Definition of cognition in English*. Retrieved 03 22, 2017, from Oxford Dictionaries: https://en.oxforddictionaries.com/definition/cognition

- Dictionary. (2017). *citizen*. Retrieved 03 22, 2017, from Dictionary.com: http://www.dictionary.com/browse/citizen
- Donbasa, N. (2014). *V Donetcke Ubili Aktivista Dimitriq Tchernqvskava*. Retrieved 04 24, 2017, from Novosti Donbasa: http://novosti.dn.ua/news/203161-v-donecke-ubyly-aktyvysta-dmytryya-chernyavskogo
- Doyle, A. C. (1892). The Adventures of Sherlock Holmes. George Newnes.
- Dubovaq, T. (2014). *Napuganie jitelei Donbasa ne propustili Ukrainkuy voennuy tehniky*. Retrieved 04 24, 2017, from Vesti: http://vesti-ukr.com/donbass/42935-v-donbasse-napugannyj-vojnoj-narod-ne-propuskaet-voennuju-tehniku
- Duma, T. S. (2015). *Stenograma zasedaniq 20 yanvarya 2015g*. Retrieved 07 03, 2017, from The State Duma official webpage: http://transcript.duma.gov.ru/node/4205/
- Economist, T. (2013). *Playing East against West*. Retrieved 07 07, 2017, from The Economist: https://www.economist.com/news/europe/21590585-success-eastern-partnership-depends-ukraine-playing-east-against-west
- Economist, T. (2014). *Protests in Ukraine On the march in Kiev*. Retrieved 04 24, 2017, from The Economist: http://www.economist.com/news/europe/21595051-protests-turn-nasty-and-violent-president-not-giving-ground-march-kiev
- Enciklopediq, B. R. (2017). *Drevno Ruskoe Gosudarstva*. Retrieved 05 29, 2017, from Bolshaqya Ruskaya Enciklopediq: http://bigenc.ru/domestic\_history/text/2631504#
- Europe, R. F. (2014). *EU Backs New Ukrainian Leadership*. Retrieved 04 24, 2017, from Radio Free Europe: http://www.rferl.org/a/ukraine-european-union-/25400911.html
- Europe, R. F. (2014). *New Ukraine Coalition Agreed, Sets NATO As Priority*. Retrieved 07 07, 2017, from Radio Free Europe: https://www.rferl.org/a/ukraine-parliament-coalition-agreement/26703123.html
- Europe, R. F. (2017). *Ukrainians Who Identify As Ethnic Russians Or Say Russian Is Their First Language*. Retrieved 05 31, 2017, from Radio Free Europe: https://www.rferl.org/a/map-ukraine-percentage-who-identify-as-ethnic-russians-or-say-russian-is-their-first-language-/25323841.html
- Farquhar, N. (2014). *Putin Lays Out Proposal to End Ukraine Conflict*. Retrieved 06 14, 2017, from The New York Times: https://www.nytimes.com/2014/09/04/world/europe/ukraine-russia.html
- Fenno, R. (1986). Observation, context, and sequence in the study of politics. *American Political Science Review*, 80(1), 3-15.
- Galeotti, M. (2016). Hybrid, ambiguous, and non-linear? How new is Russia's 'new way of war'? *Small Wars & Insurgencies*, 2(27), 282-301.
- Gallup. (2014). *Contemporary Media Use in Ukraine*. Retrieved 06 29, 2017, from Broadcasting Board of Governors: https://www.bbg.gov/wp-content/media/2014/06/Ukraine-research-brief.pdf
- Gatehouse, G. (2015). *The British Broadcasting Corporation*. Retrieved 04 24, 2017, from The untold story of the Maidan massacre: http://www.bbc.com/news/magazine-31359021
- Gazmanov, e. a. (2011). Voennaq Reforma 2008-2020 Cifri I Facti. Indeks Bezopasnosti, 33-56.

- Geers, K. (Ed.). (2015). *Cyber War in Perspective: Russian Aggression against Ukraine*. Tallinn: NATO CCD COE Publications.
- Georgia, N. S. (2008). *Population*. Retrieved 07 11, 2017, from National Statistics Office of Georgia: http://www.geostat.ge/index.php?action=page&p\_id=152&lang=eng
- Gerasimov, V. (2013). *Cennost Nauki v Previdenii*. Retrieved 05 03, 2017, from Voenno-promishlenii Kurier: http://www.vpk-news.ru/articles/14632
- Giuliani, R. (2008). *The Resilient Society A blueprint for homeland security*. Retrieved 03 20, 2017, from City Journal: https://www.city-journal.org/html/resilient-society-13060.html
- Gorchinskaya, K. (2015). *Ukraine's former ruling party hit by spate of apparent suicides*. Retrieved 06 03, 2017, from The Guardian: https://www.theguardian.com/world/2015/mar/23/ukraine-party-of-regions-members-apparent-suicides-viktor-yanukovych
- Gortney, W. (2010). *Department of Defense dictionary of military and associated terms*. Retrieved 03 22, 2017, from The Defense Technical Information Center: http://www.dtic.mil/doctrine/new\_pubs/dictionary.pdf
- Government, H. M. (2015). *National Security Strategy and Strategic Defence and Security Review 2015*. Retrieved 04 30, 2017, from UK Government official webpage: https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/478933/52309\_Cm\_9161\_NSS\_SD\_Review\_web\_only.pdf
- Graaf, E. B. (2014). Towards a theory of fear management in the counterterrorism domain: A stocktaking approach. *ICCT Research Papers*.
- Grove, G. B. (2014). *Pro-Russia separatists take armoured vehicles, humiliating Kiev forces*. Retrieved 06 05, 2017, from Reuters: http://in.reuters.com/article/ukraine-crisis-idINKBN0D20QJ20140416
- Grytsenko, A. L. (2015). *Ukrainian soldiers share horrors of Debaltseve battle after stinging defeat*. Retrieved 06 21, 2017, from The Guardian: https://www.theguardian.com/world/2015/feb/18/ukrainian-soldiers-share-horrors-of-debaltseve-battle-after-stinging-defeat
- Grytsenko, I. T. (2013). *Ukraine suspends talks on EU trade pact as Putin wins tug of war*. Retrieved 07 08, 2017, from The Guardian: https://www.theguardian.com/world/2013/nov/21/ukraine-suspends-preparations-eu-trade-pact
- Guardian, T. (2014). Agreement on the Settlement of Crisis in Ukraine full text. Retrieved 04 24, 2017, from The Guardian: https://www.theguardian.com/world/2014/feb/21/agreement-on-the-settlement-of-crisis-in-ukraine-full-text
- Guardian, T. (2014). *Ukraine crisis fuels secession calls in pro-Russian south*. Retrieved 04 24, 2017, from The Guardian: https://www.theguardian.com/world/2014/feb/23/ukraine-crisis-secession-russian-crimea
- Gunneriusson, H. (2012). Nothing is Taken Serious Until it Gets Serious: Countering Hybrid Threats. *Defence Against Terrorism Review*, *4*(1), 47-70.

- Haines, J. (2014). *How, Why And When Russia Will Deploy Little Green Men, And Why US Cannot Analysis*. Retrieved 06 07, 2014, from Eurasia Review: http://www.eurasiareview.com/12032016-how-why-and-when-russia-will-deploy-little-green-men-and-why-us-cannot-analysis/
- Hammes, T. (2004). The Sling and the Stone: On War, in the 21st Century. In T. M. Johnson, *Hybrid warfare* (pp. 1-137). St. Paul: MBI Publishing.
- Harding, L. (2014). *Ukraine forces 'helpless' to stop pro-Russia takeover in east*. Retrieved 06 03, 2017, from The Guardian: https://www.theguardian.com/world/2014/apr/30/ukraine-forces-helpless-stop-pro-russia-takeover-east
- Hardy, N. P. (2002). Discourse analysis: Investigating processes of social construction. *Sage University Papers Series on Qualitative Research Methods*, 50, 1-97.
- Hassan, K. M. (2017). The Eurasian union: actor in the making? *Journal of International Relations and Development*, 215-237.
- HCNM. (2014). *Human Rights Assessment Mission to Ukraine*. The Hague/Warsaw: OSCE. Retrieved 06 02, 2017, from http://www.osce.org/odihr/118476?download=true
- Heinsch, R. (2015). onflict Classification in Ukraine:. *International Law Studies*, 1(91), 324-360.
- Herszenhorn, S. T. (2014). *Ukrainian Military and Rebel Fighters Clash in Donetsk*. Retrieved 06 14, 2017, from The New York Times: https://www.nytimes.com/2014/07/22/world/europe/ukrainian-military-and-rebel-fighters-clash-in-donetsk.html
- Hewitt, G. (2014). *Ukraine 'hits Russia armoured column' amid aid impasse*. Retrieved 06 14, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-28810622
- Hoffman, F. (2007). *Conflict in the 21st century: The rise of hybrid wars*. Virginia: VA: Potomac Institute for Policy Studies.
- Hoffman, F. (2013). Conflict in the 21st century. In T. M. Johnson, *Hybrid warfare*. Florida.
- Huber, T. (2002). Compound Warfare: That Fatal Knot. In T. M. Johnson, *Hybrid warfare*. Florida: The JSOU Press, MacDill Air Force Base.
- Huntington, S. (1996). *The Clash of Civilizations and the Remaking of World Order*. New York: Simon and Schuster.
- Hyde, L. (2015). *Ukraine to rewrite Soviet history with controversial 'decommunisation' laws*. Retrieved 07 05, 2017, from The Guardian: https://www.theguardian.com/world/2015/apr/20/ukraine-decommunisation-law-soviet
- Igor Gujava, D. K.-M. (2014). *Kak Ukraina poterqla Krim. Rekosntrukciq sobitiq*. Retrieved 06 04, 2017, from Vesti: http://reporter.vesti-ukr.com/art/y2015/n7/13328-krymnash-krymnenash.html#.VYfr7JD4EZA
- Independent, T. (2015, November). Russia plans a 'hybrid warfare' campaign aimed at destabilising Europe, says Bulgarian President. Retrieved 03 20, 2017, from The Independent: http://www.independent.co.uk/news/world/europe/russia-plans-a-hybrid-warfare-campaign-aimed-at-destabalising-europe-says-bulgarian-president-a6734981.html

- Interfax-Ukraine. (2014). *Deschytsia states new government of Ukraine has no intention to join NATO*. Retrieved 07 07, 2017, from Interfax-Ukraine: http://en.interfax.com.ua/news/general/198372.html
- International, T. (2013). *The Corruption Perceptions Index 2013*. Retrieved 06 01, 2017, from Transparency International: https://www.transparency.org/cpi2013/results#myAnchor1
- Isachenkov, V. (2008). *Departing Putin seeks to stop NATO gains*. Retrieved 07 07, 2017, from USA Today: https://usatoday30.usatoday.com/news/world/2008-03-31-2624352020\_x.htm
- Ishchenko, V. (2016). Far right participation in the Ukrainian Maidan protests: an attempt of systematic estimation. *European Politics and Society*, 453-472.
- Jazeera, A. (2014). *Timeline: Ukraine's political crisis*. Retrieved 04 24, 2017, from Al Jazeera: http://www.aljazeera.com/news/europe/2014/03/timeline-ukraine-political-crisis-201431143722854652.html
- Johannesson, J. (2017). Russia-Ukraine balance of military power. *Journal of International Studies*, 10(1), 63-73.
- Johnson, T. M. (2013). *Hybrid warfare*. Florida: The JSOU Press, MacDill Air Force Base.
- Johnston, C. (2014). *Mikhail Gorbachev: world on brink of new cold war over Ukraine*. Retrieved 04 30, 2017, from The Guardian: https://www.theguardian.com/world/2014/nov/08/gorbachev-new-cold-war-ukraine-soviet-union-us-russia
- Jones, J. F. (2014). Raising Read Flags: An examination of Arms and Munitions in the Ongoing Conflict in Ukraine. Armament Research Services. Retrieved 06 05, 2017, from http://armamentresearch.com/Uploads/Research%20Report%20No.%203%20-%20Raising%20Red%20Flags.pdf
- Kasapoglu, C. (2015). Russia's Renewed Military Thinking: Non-Linear Warfare and Reflexive Control.

  Retrieved 05 08, 2017, from NATO Multimedia Library:

  http://www.natolibguides.info/hybridwarfare
- Kelley, M. (2014). Putin: What Happened In Kiev Was An Unconstitutional Overthrow And Yanukovych Is Still President. Retrieved 06 30, 2017, from Business Insider: http://www.businessinsider.com/putin-what-happened-in-kiev-was-an-unconstitutional-overthrow-and-yanukovych-is-still-president-2014-3
- Kendall, B. (2014, November). *Hybrid warfare: The new conflict between East and West*. Retrieved 03 20, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-29903395
- Kimball, S. (2014). *Bound by treaty: Russia, Ukraine and Crimea*. Retrieved 04 24, 2017, from Deutsche Welle: http://www.dw.com/en/bound-by-treaty-russia-ukraine-and-crimea/a-17487632
- Kirilov, V. D. (2015). *Proekt Novorussiya Zakrit*. Retrieved 07 04, 2017, from Gazeta RU: https://www.gazeta.ru/politics/2015/05/19\_a\_6694441.shtml
- Kissinger, H. (1994). Diplomacy. New York: Simon & Schuster.

- Korhonen, O. (2015). Deconstructing the Conflict in Ukraine: The Relevance of International Law to Hybrid States and Wars. *German LJ*, *16*, 452.
- Korybko, A. (2015). *Hybrid Wars: The Indirect Adaptive Approach to Regime Change*. Moscow: Institute For Strategic Studies and Predictions.
- Krasnozhon, L. (2014). *The Ethnicities Of Ukraine Are United*. Retrieved 04 27, 2017, from Forbes: https://www.forbes.com/sites/realspin/2014/03/13/the-ethnicities-of-ukraine-are-united/#384a609a110e
- Krechetnikov, A. (2014). "Opolchenci" i "Terroristui": voina slov produljaet. Retrieved 07 04, 2017, from The British Broadcasting Corporation Russia: http://www.bbc.com/russian/russia/2014/07/140714 russia ukraine rhetoric
- Kushch, L. (2014). *Pro-Russia protesters occupy regional government in Ukraine's Donetsk*. Retrieved 06 21, 2017, from Reuters: http://www.reuters.com/article/us-ukraine-crisis-donetsk-protests-idUSBREA220XI20140303
- Kyivpost. (2014). Rada speaker announces dissolution of parliamentary coalition. Retrieved 04 24, 2017, from Interfax-Ukraine News Agency: https://www.kyivpost.com/article/content/ukraine-politics/rada-speaker-announces-dissolution-of-parliamentary-coalition-357747.html
- KyivPost. (2014). *Ukrainian military bases in Crimea besieged by Russian forces*. Retrieved 06 04, 2017, from Kyiv Post: https://www.kyivpost.com/article/content/ukraine-politics/ukrainian-military-bases-in-crimea-besieged-by-russian-forces-340244.html
- Lanoszka, A. (2016). Russian hybrid warfare and extended deterrence in eastern Europe. *International Affairs*, 92(1), 175-195.
- Laruelle, M. (2015). The three colors of Novorossiya, or the Russian nationalist mythmaking of the Ukrainian crisis. *Post-Soviet Affairs*, 1-20.
- Lasconjarias, A. J. (2015). *NATO's Hybrid Flanks: Handling Unconventional Warfare in the South and the East*. Retrieved 05 07, 2017, from NATO Military Library: http://www.natolibguides.info/hybridwarfare
- Lenta. (2014). *V Luganke obqvlena narodnaq respublika*. Retrieved 04 25, 2017, from Lenta RU: https://lenta.ru/news/2014/04/28/lugansk/
- Loiko, S. (2014). *Ukraine makes plans for possible pullout of its troops from Crimea*. Retrieved 06 04, 2017, from Chicaho Tribune: http://www.chicagotribune.com/la-fg-wn-ukraine-troops-crimea-20140319-story.html
- Luhn, A. (2014). East Ukraine protesters joined by miners on the barricades. Retrieved 06 08, 2017, from The Guardian: https://www.theguardian.com/world/2014/apr/12/east-ukraine-protesters-miners-donetsk-russia
- Luhn, A. (2014). *Russia holds huge military exercises near Ukraine border*. Retrieved 06 08, 2017, from The Guardian: https://www.theguardian.com/world/2014/aug/04/russia-military-exercises-ukraine-border

- MacAskill, E. (2014). *Putin's military exercises are more than a game*. Retrieved 06 08, 2017, from The Guardian: https://www.theguardian.com/world/2014/apr/23/putin-military-exercises-ukraine-border
- Matuszak, S. (2012). *The oligarchic democracy: The influence of business groups on Ukrainian politics.*Warsaw: Centre for Eastern Studies.
- McDermott, R. (2009). Russia's conventional armed forces and the Georgian War. *Parameters*, 65-80.
- McLaughlin, D. (2004). *Putin accuses West of backing revolutions*. Retrieved 08 03, 2017, from The Irish Times: https://www.irishtimes.com/news/putin-accuses-west-of-backing-revolutions-1.1171171
- Meduza. (2014). *Vejivie Ludi Kto eto?* Retrieved 06 04, 2017, from Meduza: https://meduza.io/special/polite/howeaponed
- Melkonyan, A. G. (2017). *Inside Russia's New Foreign Policy Master Plan*. Retrieved 04 30, 2017, from The National Interest: http://nationalinterest.org/feature/inside-russias-new-foreign-policy-master-plan-18943?page=show
- Michael Aaronson, S. D. (2011). NATO countering the hybrid threat. PRISM, 2(4), 111-124.
- Miller, C. (2014). SBU detains Russian provocateur believed to have planned raid on parliament, cabinet buildings (UPDATE). Retrieved 06 02, 2017, from Kyiv Post: https://www.kyivpost.com/article/content/ukraine-politics/sbu-detains-russian-provocateur-believed-to-have-planned-raid-on-ukrainian-parliament-cabinet-341555.html
- Minasyan, S. (2015). "Hybrid" vs. "Compound" War Lessons From The Ukranian Conflict. Yerevan: Caucasus Institute .
- Mitrokhin, C. A. (1999). *The Sword and the Shield: The Mitrokhin Archive and the Secret History of the KGB*. New York: Basic Books.
- Mogashoa, T. (2014). Understanding Critical Discourse Analysis in Qualitative. *International Journal of Humanities Social Sciences and Education*, 104-113.
- Monaghan, A. (2015). *A 'New Cold War'? Abusing History, Misunderstanding Russia*. The Royal Institute of International Affairs. Retrieved 04 30, 2017, from http://www.pmln.us/wp-content/uploads/2015/06/20150522ColdWarRussiaMonaghan1.pdf
- Moreland, S. J. (2016). ISIS: An Adaptive Hybrid Threat in Transition. *Small Wars Journal*. Retrieved 05 07, 2017, from http://smallwarsjournal.com/jrnl/art/isis-an-adaptive-hybrid-threat-in-transition
- MVnR. (2014). Statement of the Spokesman of the MFA on the cancellation of the law on state language policy by the Verkhovna Rada of Ukraine on 23 February 2014. Retrieved 04 24, 2017, from Bulgarian Ministry of Foreign Affairs official webpage: http://www.mfa.bg/en/events/6/1/1311/index.html
- MZS. (2014). *Protocol on the results of consultations of the Trilateral Contact Group (Minsk, 05/09/2014*). Retrieved 06 23, 2017, from Ministry of Foreign Affairs of Ukraine: http://mfa.gov.ua/en/news-feeds/foreign-offices-news/27596-protocolon-the-results-of-consultations-of-the-trilateral-contact-group-minsk-05092014

- Nabrejnov, G. (2014). *Minoboronii raskazal ob istochnikah polucheniq orujia opalchencami*. Retrieved 06 05, 2017, from RBC Information Systems: http://www.rbc.ru/politics/30/08/2014/570421729a794760d3d4119c
- Nate Haken, J. J. (2013). *The Failed States Index 2013*. Washington: The Fund for Peace. Retrieved 07 19, 2017, from http://library.fundforpeace.org/library/cfsir1306-failedstatesindex2013-06l.pdf
- NATO. (2008). *Bucharest Summit Declaration*. Retrieved 07 07, 2017, from NATO official webpage: http://www.nato.int/cps/in/natohq/official\_texts\_8443.htm
- NATO. (2014). *Wales Summit Declaration*. Retrieved 03 20, 2017, from North Atlantic Treaty Organization official webpage: http://www.nato.int/cps/en/natohq/official\_texts\_112964.htm?selectedLocale=en
- NATO. (2015). Press statements by the NATO Secretary General Jens Stoltenberg and the EU High Representative for Foreign Affairs and Security Policy, Federica Mogherini. Retrieved 05 07, 2017, from The North Atlantic Treaty Organization official webpage: http://www.nato.int/cps/en/natohq/opinions\_125361.htm
- NATO. (2016). *Collective defence Article 5*. Retrieved 03 20, 2017, from NATO official webpage: http://www.nato.int/cps/en/natohq/topics\_110496.htm
- Nedeli, Z. (2014). *Milicii Donbasa bujno 20 000 chelovekov dlq zameni predatelev i dezertiorov, Avakov*. Retrieved 06 03, 2017, from Zerkalo Nedeli: https://zn.ua/UKRAINE/milicii-donbassa-nuzhno-20-tys-chelovek-dlya-zameny-predateley-i-dezertirov-149826\_.html
- Nemeth, W. (2002). Future war and Chechnya: A case for hybrid warfare. Retrieved 03 21, 2017, from Calhoun: The NPS Institutional Archive: http://calhoun.nps.edu/bitstream/handle/10945/5865/02Jun\_Nemeth.pdf
- Novosti, L. (2014). *Putin: Ukrainskih voenih v Krimu blokirovali Russkie voiska*. Retrieved 06 03, 2017, from Liga Novosti: http://news.liga.net/video/politics/4086201-putin\_ukrainskikh\_voennykh\_v\_krymu\_blokirovali\_nashi\_voennye.htm
- Novosti, R. (2014). *Opolcheniye DNR kontraatakuet*. Retrieved 07 03, 2017, from RIA Novosti: https://ria.ru/world/20140806/1018943217.html
- OHCHR. (2014). Report of the United Nations High Commissioner for Human Rights on the situation of human rights in Ukraine. New York: Office of the United Nations High Commissioner for Human Rights. Retrieved 04 27, 2017, from http://www.ohchr.org/EN/HRBodies/HRC/RegularSessions/Session27/Documents/A-HRC-27-75\_en.pdf
- OHCHR. (2014). Report on the human rights situation in Ukraine 15 December 2014. Retrieved 06 14, 2017, from Office of the United Nations High Commissioner for Human Rights: http://www.ohchr.org/Documents/Countries/UA/OHCHR\_eighth\_report\_on\_Ukraine.pdf
- OHCHR. (2017). Report on the human rights situation in Ukraine 16 November 2016 to 15 February 2017. Office of the United Nations High Commissioner for Human Rights. Retrieved 04 27, 2017, from http://www.ohchr.org/Documents/Countries/UA/UAReport17th\_EN.pdf

- Oliphant, R. (2014). *Ukraine crisis: the inevitable fall of Belbek*. Retrieved 06 04, 2017, from The Telegraph: http://www.telegraph.co.uk/news/worldnews/europe/ukraine/10717048/Ukraine-crisis-the-inevitable-fall-of-Belbek.html
- Oliphant, R. (2015). *Pro-Russia separatists vow further advances into Ukraine after taking Donetsk airport*. Retrieved 06 14, 2017, from The Telegraph: http://www.telegraph.co.uk/news/worldnews/europe/ukraine/11363929/Pro-Russia-separatists-vow-further-advances-into-Ukraine-after-taking-Donetsk-airport.html
- OSCE. (2014). Memorandum of 19 September 2014 outlining the parameters for the implementation of commitments of the Minsk Protocol of 5 September 2014. Organization for Security and Cooperation in Europe. Retrieved 04 27, 2017, from http://www.osce.org/home/123806
- OSCE. (2015). *Kompleks mer po vuipolneniu Minskih soglashennii*. Retrieved 06 21, 2017, from Organization for Security and Co-operation in Europe: http://www.osce.org/ru/cio/140221?download=true
- OSCE. (2015). Latest from OSCE Special Monitoring Mission to Ukraine (SMM) based on information received as of 18:00 (Kyiv time), 4 January 2015. Retrieved 06 14, 2017, from Organization for Security and Co-operation in Europe: http://www.osce.org/ukraine-smm/133421
- OSCE. (2015). *Package of Measures for the Implementation of the Minsk Agreements*. Retrieved 04 27, 2017, from http://www.osce.org/cio/140156
- OSCE. (2016). Response to Chief Observer of the Observer Mission at the Russian Border Checkpoints Gukovo and Donetsk | Statement to the PC. Retrieved 06 02, 2017, from US Mission to the OSCE: https://osce.usmission.gov/response-chief-observer-observer-mission-russian-border-checkpoints-gukovo-donetsk-statement-pc/
- OSCE. (2017). *Daily and spot reports from the Special Monitoring Mission to Ukraine*. Retrieved 06 21, 2017, from Organization for Security and Co-operation in Europe: http://www.osce.org/ukraine-smm/reports
- Oxford, D. (2017). *Definition of hybrid in English*. Retrieved 05 09, 2017, from Oxford Dictionaries: https://en.oxforddictionaries.com/definition/hybrid
- Pan, P. (2014). *Ukraine to extend Russia naval base lease, pay less for natural gas*. Retrieved 07 19, 2017, from Washington Post: http://www.washingtonpost.com/wp-dyn/content/article/2010/04/21/AR2010042103836.html
- Pawlak, P. (2017). *Countering hybrid threats: EU-NATO cooperation*. Retrieved 04 30, 2017, from European Parliament official webpage: http://www.europarl.europa.eu/thinktank/en/document.html?reference=EPRS\_BRI(2017)599315
- Peleschuk, D. (2014). *Ukraine is still at war, and its bloodiest battle isn't over*. Retrieved 06 23, 2017, from Public Radio International: https://www.pri.org/stories/2015-08-29/ukraine-still-war-and-its-bloodiest-battle-isnt-over
- pesen, a. (n.d.). 1.
- Popescu, N. (2015). Hybrid tactics: neither new nor only Russian. EU Institute for Security Studies.

- Post, K. (2014). *Kremlin-backed rebels form Novorossiya army*. Retrieved 06 08, 2017, from Kyiv Post: https://www.kyivpost.com/article/content/war-against-ukraine/kremlin-backed-rebels-form-novorossiya-army-364887.html
- Post, K. (2014). *Party of Regions, Communist Party banned in Ivano-Frankivsk and Ternopil regions*. Retrieved 06 03, 2017, from Kyiv Post: https://www.kyivpost.com/article/content/ukraine-politics/party-of-regions-communist-party-banned-in-ivano-frankivsk-and-ternopil-regions-335655.html
- Post, K. (2014). Shoigu appoints ex-Ukrainian Navy commander as Russian Black Sea Fleet deputy commander. Retrieved 07 12, 2017, from Kyiv Post:

  https://www.kyivpost.com/article/content/ukraine-politics/shoigu-appoints-ex-ukrainian-navy-commander-russian-black-sea-fleet-deputy-commander-340610.html
- Post, T. W. (2007). *Putin's Prepared Remarks at 43rd Munich Conference on Security Policy (transcript)*. Retrieved 07 07, 2017, from The Washington Post: http://www.washingtonpost.com/wp-dyn/content/article/2007/02/12/AR2007021200555.html
- Post, T. W. (2014). *Transcript: Putin says Russia will protect the rights of Russians abroad*. Retrieved 07 05, 2017, from The Washington Post: https://www.washingtonpost.com/world/transcript-putin-says-russia-will-protect-the-rights-of-russians-abroad/2014/03/18/432a1e60-ae99-11e3-a49e-76adc9210f19\_story.html?utm\_term=.732e7cbd6a98
- Pravda, U. (2014). *Sili ATO znishtili snaiperev u Lisichansku*. Retrieved 06 08, 2017, from Ukrainskaya Pravda: http://www.pravda.com.ua/news/2014/07/26/7033039/
- Press, E. (2014). *Ukraine's Parliament Announces Mobilization*. Retrieved 07 13, 2017, from Euromaidan Press: http://euromaidanpress.com/2014/03/18/ukraines-parliament-announces-mobilization/
- Pulkki, A. (2014). *Crimea Invaded By High Readiness Forces Of The Russian Federation*. Retrieved 06 04, 2017, from Soldier of Finlad magazine: https://web.archive.org/web/20150330124704/http://www.suomensotilas.fi/en/artikkelit/crimea-invaded-high-readiness-forces-russian-federation
- Pursiainen, T. F. (2017). The Psychological Dimension of Russian Foreign Policy: Putin and the Annexation of Crimea. *Global Society*, 220-244.
- Putin, V. (2005). *Annual Address to the Federal Assembly of the Russian Federation*. The Kremlin, Moscow. Retrieved 04 29, 2017, from Official Internet Resources of the President of Russia: http://kremlin.ru/events/president/transcripts/22931
- Putin, V. (2014). *Direct Line with Vladimir Putin*. Retrieved 06 03, 2017, from President of Russia official webpage.
- Putin, V. (2014). *Presient Rossii Vladimir Putin obratilsya k oplachencev Novorossii*. Retrieved 07 03, 2017, from President of Russia official webpage: http://kremlin.ru/events/president/news/46506
- Rácz, A. (2015). *Russia's Hybrid War in Ukraine:Breaking the Enemy's Ability to Resist*. Retrieved 03 20, 2017, from Finnish Institute of International Affairs: http://www.fiia.fi/en/publication/514/russia\_s\_hybrid\_war\_in\_ukraine/
- Rada, V. (2015). *Zakon Ukraini, dokument 158-19*. Retrieved 07 13, 2017, from Verhovna Rada of Ukraine official web portal: http://zakon3.rada.gov.ua/laws/show/158-19

- Ray, M. (2014). *Donets Basin*. Retrieved 06 08, 2017, from Encyclopedia Britannica: https://www.britannica.com/place/Donets-Basin
- Ray, M. (2016). *Ukraine crisis*. Retrieved 04 25, 2017, from Encyclopedia Britannica: https://www.britannica.com/topic/Ukraine-crisis
- Recknagel, C. (2014). *Explainer: How Do Russia's And Ukraine's Armies Compare?* Retrieved 07 11, 2017, from Radio Free Europe: https://www.rferl.org/a/russia-ukraine-armies-compare/25287910.html
- Rein, D. S. (2013). Frame Reflection: Toward the Resolution of Intractable Policy Controversies. In D. Y. Haarb, *People out of place: allochthony and autochthony in the Netherlands' identity discourse metaphors and categories in action* (pp. 227-261). Journal of International Relations and Development.
- Renz, B. (2016). Russia and 'hybrid warfare'. Contemporary Politics, 22(3), 283-300.
- Reserch, G. (2015). *Ultimate Fiasco of the Ukrainian Army*. Retrieved 06 03, 2017, from Global Reserch: http://www.globalresearch.ca/ultimate-fiasco-of-the-ukrainian-army/5487227
- Reuters. (2014). *Military airport in Ukraine's Crimea taken over by Russian soldiers-Interfax*. Retrieved 06 04, 2017, from Reuters: http://uk.reuters.com/article/ukraine-crisis-idUKL6N0LX0CG20140228
- Reuters. (2014). Russian marine kills Ukraine navy officer in Crimea, says ministry. Retrieved 05 04, 2017, from Reuters: http://www.reuters.com/article/us-ukraine-crisis-military-idUSBREA360GB20140407
- Riabchuk, M. (2016). Ukrainians as Russia's negative 'other': History comes full circle. *Communist and Post-Communist Studies*, 75-85.
- Rojansky, M. K. (2015). A closer look at Russia's "Hybrid War. Kennan Cable, 7(1), 1-8.
- Rossii, P. (2014). *President Rossii official webpage*. Retrieved 06 28, 2017, from Voennaq doktrina Rossiiskoi Federacii: http://kremlin.ru/supplement/461
- Roth, A. (2014). *From Russia, 'Tourists' Stir the Protests*. Retrieved 04 24, 2017, from The New York Times: https://www.nytimes.com/2014/03/04/world/europe/russias-hand-can-be-seen-in-the-protests.html? r=0
- RT. (2014). 'We did what we had to do': Putin opens up on Crimea reunification plan. Retrieved 06 22, 2017, from Russia Today: https://www.rt.com/news/239197-putin-crimea-referendum-decision/
- RT. (2014). Facts you need to know about Crimea and why it is in turmoil. Retrieved 04 24, 2017, from Russia Today: https://www.rt.com/news/crimea-facts-protests-politics-945/
- RT. (2014). *Putin acknowledges Russian military serviceman were in Crimea*. Retrieved 06 06, 2017, from Russia Today: https://www.rt.com/news/crimea-defense-russian-soldiers-108/
- RT. (2014). *Russia tests 100,000 troops in 'Vostok 2014', biggest-ever post-Soviet drills*. Retrieved 06 08, 2017, from Russia Today: https://www.rt.com/news/189900-kamchatka-military-drills-shoigu/
- RT. (2014). *Thousands rally against 'illegitimate govt', raise Russian flags in eastern Ukraine*. Retrieved 04 24, 2017, from Russia Today: https://www.rt.com/news/donetsk-kharkov-ukraine-protest-365/

- RT. (2014). *Ukraine's east on fire: Kharkov demands referendum, Donetsk prosecutor's HQ stormed.*Retrieved 04 24, 2017, from Russia Today: https://www.rt.com/news/ukraine-kharkov-rights-donetsk-202/
- RT. (2015). *Ukrainian parliament passes law allowing army deserters to be shot*. Retrieved 07 13, 2017, from Russia Today: https://www.rt.com/news/229739-ukraine-army-shoot-defectors/
- Ruitenberg, I. H. (2004). Citizen response to disasters: a survey of literature and some practical implications. *ournal of Contingencies and Crisis Management*, 12(3), 98-111.
- Rynning, S. (2015). The false promise of continental concert: Russia, the West and the necessary balance of power. *International Affairs*, 91(3), 539-552.
- Samokhvalov, V. (2015). Ukraine between Russia and the European Union:. *Europe-Asia Studies*, 1371-1393.
- Secretary, T. W. (2014). *Statement by the President on Ukraine*. Retrieved 04 24, 2017, from Obama Whitehouse archives: https://obamawhitehouse.archives.gov/the-press-office/2014/03/20/statement-president-ukraine
- Seely, O. J. (2015). Russian Full-Spectrum Conflict: An Appraisal After Ukraine. *The Journal of Slavic Military Studies*, 28(1), 1-22.
- Sengupta, K. (2014). *Crimea crisis: With no orders from Kiev, Ukraine's besieged forces are starting to feel let down*. Retrieved 07 13, 2017, from The Independent: http://www.independent.co.uk/news/world/europe/ukraine-crisis-with-no-orders-from-kiev-the-besieged-forces-in-crimea-are-starting-to-feel-let-down-9182606.html
- Sevodnya. (2016). *Vsio chto nujno znat o mobilizacii v Ukraine*. Retrieved 07 13, 2017, from Sevodnya: http://www.segodnya.ua/ukraine/vse-chto-nuzhno-znat-o-mobilizacii-v-ukraine-kogo-prizovut-chto-grozit-uklonistam-i-novshestva--684250.html
- Shevchenko, V. (2014). "Little green men" or "Russian invaders"? Retrieved 04 24, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-26532154
- Shuster, S. (2014). *Ukraine Troops in Crimea Face Dilemma: To Defect, Flee or Fight*. Retrieved 07 12, 2017, from Time: http://time.com/17356/ukraine-troops-in-crimea-face-dilemma-to-defect-flee-or-fight/
- Shuster, S. (2014). *Ukrainian Base in Crimea Resists Russian Raid*. Retrieved 06 04, 2017, from Time Magazine: http://time.com/16294/russia-crimea-sevastopol/
- Simao, L. (2016). The Ukrainian conflict in Russian foreign policy: Rethinking the interconnections between domestic and foreign policy strategies. *Small Wars & Insurgencies*, 491-511.
- Sindelar, D. (2014). *Was Yanukovych's Ouster Constitutional?* Retrieved 04 24, 2017, from Radio Free Europe: http://www.rferl.org/a/was-yanukovychs-ouster-constitutional/25274346.html
- Sindelar, D. (2014). *Who Are Ukraine's 'Ultras'?* Retrieved 06 07, 2017, from Radio Free Europe: https://www.rferl.org/a/ukraine-protests-sports-fans-euromaidan/25244357.html
- Snegovaya, M. (2015). *Putin's information warfare in Ukraine*. Washington: Institute for the Study of War. Retrieved 06 28, 2017, from

- https://www.researchgate.net/profile/Maria\_Snegovaya/publication/282326292\_Putin's\_Informati on\_Warfare\_in\_Ukraine/links/560bf38208aea68653d349fb.pdf
- Sokolsky, R. (2017). *The New NATO-Russia Military Balance: Implications for European Security*. Retrieved 05 25, 2017, from Carnegie Endowment for International Peace: http://carnegieendowment.org/2017/03/13/new-nato-russia-military-balance-implications-for-european-security-pub-68222
- Solovjova, J. (2016, July). *Is Russia really a threat to the Baltic states?* Retrieved 03 20, 2017, from Al Jazeera: http://www.aljazeera.com/indepth/features/2016/07/russia-threat-baltic-states-160707054916449.html
- Solutions, W. I. (2017). *Ukraine Trade at a Glance : Most Recent Values*. Retrieved 06 01, 2017, from World Integrated Trade Solutions: http://wits.worldbank.org/CountrySnapshot/en/UKR
- Soyuz, E. E. (2017). *Obshtaya informaciya*. Retrieved 07 08, 2017, from Evraziiskii Ekonomicheskii Soyuz: http://www.eaeunion.org/#about
- Squires, B. F. (2014). *I can take Kiev in two weeks, Vladimir Putin warns European leaders*. Retrieved 05 27, 2017, from The Telegraph: http://www.telegraph.co.uk/news/worldnews/europe/ukraine/11069070/I-can-take-Kiev-in-two-weeks-Vladimir-Putin-warns-European-leaders.html
- Statistics, S. (2013). *State Statistics Service of Ukraine documents publishing*. Retrieved 07 11, 2017, from Population: https://ukrstat.org/en/operativ/operativ/2007/ds/nas\_rik/nas\_e/nas\_rik\_e.html
- Stern, D. (2014). *Ukraine crisis: Donetsk rebels in mass withdrawal*. Retrieved 06 14, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-28177020
- Stern, D. (2014). *Ukraine crisis: Rebels abandon Sloviansk stronghold*. Retrieved 06 14, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/world-europe-28174104
- Stoyanov, N. (2016). Hybrid wars and non-violent revolutions. Sofia: Iztok-Zapad.
- Strovsky, D. (2015). The Russian Media Coverage of the 'Ukrainian Issue': the Priorities of Informing. *Studies in Media and Communication*, 98-108.
- Sukhov, O. (2015). *Ukraine's navy barely recovering from its near-death experience*. Retrieved 06 01, 2017, from Kyiv Post: https://www.kyivpost.com/article/content/kyiv-post-plus/ukraines-navy-barely-recovering-from-its-near-death-experience-396689.html
- Szostek, J. (2014). Russia and the News Media in Ukraine . *East European Politics and Societies* , 463-486.
- TASS. (2014). *Putin signs laws on reunification of Republic of Crimea and Sevastopol with Russia*. Retrieved 04 24, 2017, from Russian News Agency: http://tass.com/russia/724785
- TASS. (2015). *Putin: RF vsegda budet zashishat sootechestvenikov zarubejom, v tom chisle Livii i Sirii*. Retrieved 05 31, 2017, from Russian News Agency TASS: http://tass.ru/politika/2409317
- Tavernise, A. R. (2014). *Russians Revealed Among Ukraine Fighters*. Retrieved 06 14, 2017, from The New York Times: https://www.nytimes.com/2014/05/28/world/europe/ukraine.html?\_r=1

- Tchuzakov, S. (2014). *Na Ukraine otmenili zakon o regionalnom statuse ruskovo jazika*. Retrieved 04 24, 2017, from Lenta.RU: https://lenta.ru/news/2014/02/23/language/
- Telegraph, T. (2014). Fears of full-scale Russian invasion as eastern Ukraine cities toppled. Retrieved 06 02, 2017, from The Telegraph:

  http://www.telegraph.co.uk/news/worldnews/europe/ukraine/10763008/Fears-of-full-scale-Russian-invasion-as-eastern-Ukraine-cities-toppled.html
- Telegraph, T. (2015). *Ukraine crisis: timeline of major events*. Retrieved 06 24, 2017, from The Telegraph: http://www.telegraph.co.uk/news/worldnews/europe/ukraine/11449122/Ukraine-crisis-timeline-of-major-events.html
- Thibault, D. (2005). Commented APP-6A Military symbols for land based systems NATO's current military symbology standard. DRDC Valcartier. Retrieved 05 19, 2017, from http://www.mapsymbs.com/APP-6ADRDCValcartierEdition121(Mod).pdf
- Thomas, T. (2004). Russia's reflexive control theory and the military. *Journal of Slavic Military Studies*, 237-256.
- Thomas, T. (2015). Russia's Military Strategy and Ukraine: Indirect, Asymmetric—and Putin-Led. *The Journal of Slavic Military Studies*, *3*(28), 445-461.
- Traynor, I. (2007, May). *Russia accused of unleashing cyberwar to disable Estonia*. Retrieved 03 20, 2017, from The Guardian: https://www.theguardian.com/world/2007/may/17/topstories3.russia
- Tsvetkova, A. V. (2014). *This Elusive Muscovite With 3 Names Has Taken Control Of Ukraine Rebels*. Retrieved 06 08, 2017, from Business Insider: http://www.businessinsider.com/r-elusive-muscovite-with-three-names-takes-control-of-ukraine-rebels-2014-15?IR=T
- Tudoroiu, T. (2016). Unfreezing Failed Frozen Conflicts: A Post-Soviet Case Study. *Journal of Contemporary European Studies*, 375-396.
- Tzu, S. (1910). "The art of war (L. Giles, Trans.).". London: Luzac & Co.
- Ukraina, G. K. (2001). *Nacionalnii sostav naseleniq, grajdanstvo*. Retrieved 05 31, 2017, from Gosudarstvenii Komitet Statistiki Ukraina: http://2001.ukrcensus.gov.ua/rus/results/nationality\_population/
- Ukraina, G. P. (2014). Samoprovuzglashonom ministr oboroni DNR strelkovy inkriminiruetsya stvoreniq teroristichnoi organizacii ta vchineniq terraktii v Ukraini. Retrieved 06 08, 2017, from Generalnaq Prokuratura Ukraina press slujba: http://www.gp.gov.ua/ua/news.html?\_m=publications&\_c=view&\_t=rec&id=138810
- Ukraina, R. N. (2014). *Protestii Protiv mobilizacii*. Retrieved 07 13, 2017, from RIA Novosti Ukraina: http://rian.com.ua/infografika/20140731/355517286.html
- *Ukraine Reform Monitor: April 2017.* (2017). Retrieved 06 23, 2017, from The Carnegie Endowment For International Peace: http://carnegieendowment.org/2017/04/19/ukraine-reform-monitor-april-2017-pub-68700
- Ukraini, D. P. (2007). *Zagalna harakteristika derjavnovo kordonu Ukranini*. Retrieved 06 01, 2017, from Derjavna Prirodna Slujba Ukraini:

- $https://web.archive.org/web/20071206030446/http://www.pvu.gov.ua:80/control/uk/publish/article/art_id=47129\&cat_id=46429$
- *Ukrainskaq Pravda*. (2014). Retrieved 06 02, 2017, from Separatistam platit 500 dolarim za shturm djerelom: http://www.pravda.com.ua/news/2014/04/13/7022256/
- UN. (2008). Small Arms and Light Weapons: Selected United Nations Documents. New York: United Nations Publications.
- UN. (2017). *Categories of major conventional arms*. Retrieved 05 19, 2017, from United Nations Register of Conventional Arms: https://www.unroca.org/categories
- UNIAN. (2016). *Yanukovych status in Russia clarified*. Retrieved 06 03, 2017, from Ukrainian Independent Information Agency of News: https://www.unian.info/politics/1670577-yanukovych-status-in-russia-clarified.html
- Union, G. C. (2017). *Press Release No 6/17*. Luxembourg,. Retrieved 06 02, 2017, from https://curia.europa.eu/jcms/upload/docs/application/pdf/2017-01/cp170006en.pdf
- V mqtejnom Kieve policiq ubegaet ot naroda. Buntovshiki s leko uvladeli stolice. (2013). Retrieved 06 30, 2017, from Konsomolskaya Pravda: http://www.balkans.kp.ru/daily/26166.5/3053321/
- Valeiskaq, J. (2014). Shokirushtoe video. Retrieved 04 24, 2017, from Glavnii Sait V Gorode: http://dn.vgorode.ua/news/sobytyia/215093-shokyruuischee-vydeo-na-koleny-tvary-yly-kak-prokhodyla-travlia-ukraynskykh-aktyvystov
- Vedeneev, D. (2002). Viiskovo poliova jandarmeriya specialnii organi Ukrainskoipovstanskoi armii. *Voenna Istoriya*. Retrieved from http://warhistory.ukrlife.org/5\_6\_02\_4.htm
- Verhovod, V. (2014). *Opolcheniye*. Retrieved 07 03, 2017, from Bolshaya Ryskaya Enciklopediya: http://bigenc.ru/military\_science/text/2690959#
- Veritas. (2005). *The physical domain*. Retrieved 03 22, 2017, from Veritas Society: http://vsociety.net/wiki/Physical\_domain
- Vladimir Sazonov, H. M.-V. (2016). Russian Information Warfare against the Ukrainian State and Defence Forces: April-December 2014. Riga: NATO Strategic Communications Centre of Excellence. Retrieved 06 28, 2017, from http://stratcomcoe.org/russian-information-campaignagainst-ukrainian-state-and-defence-forces
- Vlasiuk, V. (2015). Hybrid war, international law and eastern Ukraine. *Evropsky politicky a pravni diskurz*, 2, 14-28.
- Vlasova, O. G. (2015). *Thousands of Ukrainian soldiers trapped as Debaltseve pocket closes*. Retrieved 06 23, 2017, from Kyiv Post: https://www.kyivpost.com/article/content/war-against-ukraine/thousands-of-soldiers-endangered-in-debaltseve-pocket-380978.html
- Walker, P. (2014). *Pro-Russian insurgents retreat in Ukraine's east*. Retrieved 06 14, 2017, from USA Today: https://www.usatoday.com/story/news/world/2014/05/16/ukraine-russia/9162677/
- Walker, S. (2013). *Ukraine's EU trade deal will be catastrophic, says Russia*. Retrieved 07 08, 2017, from The Guardian: https://www.theguardian.com/world/2013/sep/22/ukraine-european-union-traderussia

- Walker, S. (2014). *Ukraine ceasefire 'agreed for east of country' at Minsk peace talks*. Retrieved 06 14, 2017, from The Guardian: https://www.theguardian.com/world/2014/sep/05/ukraine-ceasefire-east-minsk-peace-talks
- Walker, S. (2014). *Ukraine navy officers reject plea to defect to Russian-backed Crimea*. Retrieved 07 12, 2017, from The Guardian: https://www.theguardian.com/world/2014/mar/03/ukraine-navy-officers-defect-russian-crimea-berezovsky
- Walker, S. (2015). *Putin admits Russian military presence in Ukraine for first time*. Retrieved 06 08, 2017, from The Guardian: https://www.theguardian.com/world/2015/dec/17/vladimir-putin-admits-russian-military-presence-ukraine
- Waslander, S. (2014). Why is Ukraine so Important for Russia? Retrieved 05 29, 2017, from Foreign Investor: http://foresightinvestor.com/articles/18361-why-is-ukraine-so-important-for-russia
- Welch, G. L. (2011). *Cyberspace-the Fifth Operational Domain*. Institute for Defense Analyses Research Notes. Retrieved 05 19, 2017, from https://www.ida.org/~/media/Corporate/Files/Publications/ResearchNotes/RN2011/2011%20Cybe rspace%20-%20The%20Fifth%20Operational%20Domain.pdf
- Whewell, T. (2014). *The Russians fighting a 'holy war' in Ukraine*. Retrieved 06 14, 2017, from The British Broadcasting Corporation: http://www.bbc.com/news/magazine-30518054
- Williams, A. (2014). *Russia holds military exercises in Baltic in response to NATO*. Retrieved 06 08, 2017, from Reuters: http://uk.reuters.com/article/uk-russia-military-exercises-idUKKBN0EN10020140612
- Wilson, A. (1995). The Donbas between Ukraine and Russia: The Use of History in Political Disputes . *Journal of Contemporary History*, 265-289.
- Yin, R. (2003). Case study research design and methods. London: Sage Publications, Inc.
- Yuhas, A. (2014). *Ukraine crisis: an essential guide to everything that's happened so far.* Retrieved 07 11, 2017, from The Guardian: https://www.theguardian.com/world/2014/apr/11/ukraine-russia-crimea-sanctions-us-eu-guide-explainer
- Zainal, Z. (2007). Case study as a research method. *Jurnal Kemanusiaan*, 9(1), 1-6.
- Zverev, A. (2014). Exclusive: Ukraine rebel commander acknowledges fighters had BUK missile. Retrieved 06 08, 2017, from Reuters: http://www.reuters.com/article/us-ukraine-crisis-commander-exclusive-idUSKBN0FS1V920140723